

https://theses.gla.ac.uk/

Theses Digitisation:

https://www.gla.ac.uk/myglasgow/research/enlighten/theses/digitisation/

This is a digitised version of the original print thesis.

Copyright and moral rights for this work are retained by the author

A copy can be downloaded for personal non-commercial research or study, without prior permission or charge

This work cannot be reproduced or quoted extensively from without first obtaining permission in writing from the author

The content must not be changed in any way or sold commercially in any format or medium without the formal permission of the author

When referring to this work, full bibliographic details including the author, title, awarding institution and date of the thesis must be given

Enlighten: Theses <u>https://theses.gla.ac.uk/</u> research-enlighten@glasgow.ac.uk

THE ROLE OF PERIPHERAL QUANTITATIVE COMPUTED TOMOGRAPHY AT THE ULTRA-DISTAL RADIUS IN THE ASSESSMENT AND MANAGEMENT OF OSTEOPOROSIS

· · · · ·

and the second second

.

a second second second

by

James Clason Martin

Submitted for the degree of

Doctor of Medicine (MD)

to

Glasgow University

in

March 1997

Work for this thesis was performed at:

The Osteoporosis Research Unit Aberdeen Royal NHS Trust

. ... ·.

ProQuest Number: 10391422

All rights reserved

INFORMATION TO ALL USERS

The quality of this reproduction is dependent upon the quality of the copy submitted.

In the unlikely event that the author did not send a complete manuscript and there are missing pages, these will be noted. Also, if material had to be removed, a note will indicate the deletion.



ProQuest 10391422

Published by ProQuest LLC (2017). Copyright of the Dissertation is held by the Author.

All rights reserved.

This work is protected against unauthorized copying under Title 17, United States Code Microform Edition © ProQuest LLC.

ProQuest LLC. 789 East Eisenhower Parkway P.O. Box 1346 Ann Arbor, MI 48106 – 1346



and a second start of the start of the start of the second start of the start of the start of the start of the



Declaration of the Work of Others in this Thesis

and the second second second second second second

All studies were designed and subjects recruited personally, except those described in chapter 10, and the hip fracture study described in chapter 8.

Most of the pQCT scans were performed, and all were analysed personally. I performed all pQCT scans on those individuals described in chapter 10, and most of those described in the hip fracture study in chapter 8. Scans not done personally, were done by radiographers trained by myself in the use of pQCT.

Many of the DXA and ultrasound scans were performed, and all analysed personally. The precision measurements for calcaneal ultrasound were performed by Dr A Stewart, of the Osteoporosis Research Unit, Aberdeen Royal NHS Trust. She also designed the study, and recruited the patients participating in the hip fracture study discussed in chapter 8. However, she was not involved in any aspect of the pQCT measurements.

Dr R Munro assisted in the assessment of patients with rheumatoid arthritis studied in chapter 11.

All data presented in this thesis were collected, recorded and analysed personally. Statistical guidance was sought from M Campbell, Senior Statistician, Health Services Research Unit, Department of Public Health, Polwarth Building, Aberdeen.

I wrote this thesis in its entirety.

For my wife

Daniela,

and daughter

Rebecca.

ACKNOWLEDGEMENTS

Many people have helped me in many ways.

かいかい かんそうかん かんじかい かんしょうかい

Special thanks to my supervisor, Dr David Reid. But for his counsel, this thesis would not have been possible. A special friend always.

I would like to thank all of the administrative, research and radiography staff of the Osteoporosis Research Unit. They welcomed me warmly into their research team, and were always available for support, supervision and guidance.

I would like to thank Marion Campbell for her statistical assistance.

I am indebted to the people of Grampian Region, Scotland. For research into osteoporosis, they gave willingly of their time. Without such commitment, it would be difficult to promote knowledge and awareness of an increasing common condition.

I give great thanks to my parents for their support throughout these recent years, and many before.

Finally, to my wife and daughter I owe the greatest debt. Enormous understanding and encouragement were always forthcoming, when so often required.

iv

SUMMARY

Osteo-densitometry has some well defined roles in the management of osteoporosis. This thesis examined the role of a new technique, peripheral quantitative computed tomography (pQCT) at the ultra-distal radius, which quantifies trabecular and cortical bone mineral density (BMD) separately through a volumetric measurement. Comparison was made with dual energy X-ray absorptiometry (DXA), and calcaneal ultrasound attenuation (BUA) and velocity (VOS) in some studies.

2

and the Party

National Contraction

5

- Sector

The precision [expressed as the co-efficient of variation (CV)] of pQCT in young females was 1.24%, 1.33%, 1.58% and 1.88% for total (Qtot), trabecular (Qtrab), subcortical (Qscort) and cortical (Qcort) BMD respectively. The precision was improved to these figures by minimising the difference in voxel numbers between repeat scans, a strategy which was thereafter utilised to aid repositioning in follow-up scans. The corresponding CV's were higher in women with vertebral fractures, or rheumatoid arthritis. There were small, and insignificant differences in BMD between dominant and non-dominant forearms. To minimise the variance in measurements, the non-dominant forearm was always scanned unless contra-indicated.

Normative pQCT data were derived from 332 normal females aged 18-90 years. Differences compared to the manufacturers reference data were noted, stressing the importance of creating normal range from the local population, a Age (all measurements), postmenopausal status (Qtot, Qscort) and height (Qtot, Qtrab) were negatively, and weight (Qtot, Qtrab, Qcort) was positively related to radial BMD. A cubic regression model best fitted all pQCT BMD's as a function of age for the whole population. pQCT BMD was effectively stable in premenopausal women. In postmenopausal women, estimated rates of change in

v

BMD (derived from best fit, non-linear regression models as a function of years postmenopause) were greatest for Qtot, Qscort and Qcort at 20 years postmenopause (-1.36%/yr - -1.39%/yr), which were greater than those for Qtrab throughout the postmenopausal years (-0.53%/yr - -0.76%/yr). Significant annualized rates of change in Oscort (-1.35%/yr,p=0.024) and Qtot (-1.04%/yr,p=0.027), but not Qtrab (-0.19%/yr) or Qcort (-0.95%/yr) were observed in 23 late premenopausal women studied longitudinally. Bearing in mind precision values, these results suggest a 3 year interval between scans would be necessary to detect clinically significant individual changes in BMD. Perimenopausal differences (between 79 late premenopausal and 40 early postmenopausal women) were smaller for pQCT [BMD (Tscore difference, p-value): Qscort(-0.58,0.018), Otot(-0.54,0.013), Qtrab(-0.35,0.098), Qcort(-0.29,0.174)] than for DXA lumbar spine (LS:-0.61,0.003), and femoral neck (FN:-0.76,<0.001), trochanter (FT:-0.81,0.001) and Ward's (FW:-0.77, <0.001). These observed and estimated rates of change in pQCT BMD, and the differences between late premenopausal and early postmenopausal women, are contrary to the accepted theory of accelerated trabecular bone loss immediately following the menopause, at least at the radius. Rather they suggest remodelling, with trabecularisation of endosteal cortical bone.

그렇게 잘못한 그가 안못 가지 않는다.

19. 19.

ar . 32a.

1.1

4

1

and some

1.4

and the second s

200 A

. .

a with a street and able of

". N. W. B.

100 million (100 million)

a Artesta

In 216 perimenopausal women, correlations of Qtot, Qtrab and Qscort with DXA LS and hip measurements were moderate (r:0.35-0.53), and poorer with EUA (r:0.24-0.31) and VOS (r:0.1-0.17). Corresponding figures for Qcort were even poorer. Consequently, pQCT would be of no value in pre-selecting perimenopausal women for axial DXA assessment. To detect all potential LS or FN osteopenia based upon a pQCT measurement, 98.6%-100% of the perimenopausal population would require a pQCT scan.

BMD in female vertebral and hip fracture populations, and the

vi

power of pQCT compared to DXA and ultrasound in fracture discrimination was studied. There were significant differences in Qtrab [(Z-score difference, p-value): -1.35,<0.001], Qtot (-0.94,0.001), LS (-1.37,<0.001), FN (-1.08,0.003), FT (-1.03,0.006) and FW (-0.64,0.023), but not Qscort (-0.71,0.063) or Qcort (-0.24, 0.2) between vertebral fracture (n=39) and nonfracture (n=30) groups. Qtrab was superior (numerically, but not statistically) to other pQCT and axial DXA measurements in discriminating vertebral fractures [Area under the Receiver Operator Curve (AUC): 0.853]. There were statistically significant differences in all pQCT (Z-score differences: -0.48 to -0.73), DXA hip (-0.64 to -1.04) and ultrasound (-0.64 to -(0.91) measurements between hip fracture (n=165) and nonfracture (n=45) groups. \mathbf{FN} (AUC:0.796) was superior (numerically and statistically) to all pQCT measurements and BUA, but not VOS (AUC:0.783) in discriminating hip fractures. Corresponding AUC's for pQCT measurements (except Qcort) were greater in the vertebral than hip fracture study, suggesting that pQCT may have a greater role in the prediction of vertebral fractures.

.1

1.1

ł

a the second second

4

and the second se

.

1

Osteo-densitometry has a recognised role in monitoring response to drug therapy. Changes in pQCT EMD were compared to DXA hip and spine BMD during treatment for 1 year with HRT (n=11) and cyclical etidronate (ETD:n=10) in postmenopausal women. In both groups, the greatest increase after 1 year was found at LS (HRT:+6.6%,p<0.001; ETD:+5.2%,p=0.013), with significant change detected after only 4 months (HRT:+3.3%,p=0.02; ETD:+3.8%, p=0.01). In both groups, pQCT BMD remained virtually unchanged, suggesting that routinely, response to HRT and ETD in best detected by monitoring LS EMD. An adverse effect of warfarin on BMD was observed in a cross-sectional study. 40 men requiring longterm warfarin for cardiovascular disease, were matched for underlying disease with 40 controls. All pQCT and axial DXA BMD

vii

measurements were lower in the warfarin group, reaching statistical significance for LS (-10.4% difference,p=0.003) and Qtrab (-9% difference,p=0.024).

j.

3

3

1.44

.

and the second of the second second

AL.

1000

i i i trai

A CONTRACT TO A CONTRACT

BMD is reduced in patients with rheumatoid arthritis (RA). The pathogenesis is complex, and the contribution of low dose corticosteroid therapy controversial. pQCT was compared to axial DXA and ultrasound in 75 postmenopausal women, of which 29 were controls, 21 non-steroid treated and 25 steroid treated RA patients. There was no significant steroid effect on any BMD measurement. Qtrab (p=0.001), Qscort (p=0.009) and Qtot (p=0.016), were significantly lower in the RA groups, as were both ultrasound measurements (p=0.001) and all DXA hip measurements (p=0.012-0.002). Differences in LS and Qcort were insignificant. The principle determinant of Qtot, Qtrab and ultrasound measurements was the degree of radiological damage (Larsen score). pQCT and ultrasound measurement sites are periarticular, and may be of value in assessing RA disease activity and response to therapy.

		Page
INDEX		ix
LIST OF TA	ABLES	xiv
LIST OF FI	IGURES	xvii
CHAPTER 1	INTRODUCTION	1
1.1	Bone	2
1.1.1	Normal Cellular Function	2
1.1.2	Bone Types, Skeletal Distributions and Biomechanical Properties	4
1.1.3	Normal Age Related Bone Mass Changes	8
1.1.4	Describing Bone Mineral Density (BMD) Measurements: T and Z Scores	10
1 2	Osteoporosis	13
1 2 1	Wistory and Definition of	13
	Osteoporosis	
1.2.2	Osteoporotic Fractures	15
1.2.2.1	Hip Fracture	16
1.2.2.2	Vertebral Fracture	17
1.2.2.3	Distal Fore-arm Fracture	18
1.2.3.1	Risk Factors and Causes of Secondary Osteoporosis	19
1.2.3.2	Corticosteroid Induced Osteoporosis	19
1.3	Techniques Used to Assess Bone Mass	23
1.3.1	Radiographs	23
1.3.2	Single Photon Absorptiometry (SPA)	25
1.3.3	Dual Photon Absorptiometry (DPA)	25
1.3.4	Dual Energy X-ray Absorptiometry (DXA)	26
1.3.5	Ultrasound	28
1.3.6	Quantitative Computed Tomography (QCT)	29
1.3.7	Peripheral Quantitative Computed Tomography (pQCT)	31
1.4	Clinical Indications for BMD Measurement	33
1.5	Screening for Osteoporosis	35
1.6	Rational behind Thesis Work	40

はたいで、ここのたい

11日本の「後に開設したいない」ではないに認られていた。 シアング 歴史で、大学校で

-

うまうにいたいろうぼう 読ん カイト・アイド ある

ix

and the second second second

「人口」である。それ、日本の職業である。

「「「あっ」のないので

ロウトロ

1

ji L

i. T

- 時間の時間のの一部に、 時間、 間で、 アイト

う あいかい い

まとうという、ちがた、いないにあり、ない

ġ

CHAPTER 2	METHODOLOGY	48
2.1	Introduction	48
2.2	Peripheral Quantitative Computed Tomography (pQCT)	48
2.2.1	The Osteo-densitometer	48
2.2.2	Scanning Method	49
2.2.3	Bone Mineral Density (BMD) measurements acquired by pQCT	50
2.3	Dual Energy X-ray Absorptiometry (DXA)	51
2.4	Ultrasound (CUBA)	52
2.5	Statistical Analyses	52
2.6	Ethics and Consent	53
CHAPTER 3	THE PRECISION OF POCT AND HOW IT IS IMPROVED USING SCAN VOXEL NUMBERS	61
3.1.	Introduction	61
3.1.1	Precision	61
3.1.2	Voxel Numbers and "Trending".	61
3.2	Methods	62
3.3	Statistical Analyses	62
34	Study Populations	62
35	Results	63
3.5	Discussion	63
3.0	D ¹ B C RB21 011	03
CHAPTER 4	THE EFFECT OF DOMINANCE ON RADIAL PQCT BMD MEASUREMENTS	69
4.1	Introduction	69
4.2	Patient Recruitment and Methods	60
4 3	Statistical Analyses	0.3 70
4.J 4 A	Begulte	70
45	Discussion	70
I. J	DISCUSSION	70
CHAPTER 5	CREATION OF A NORMAL RANGE AND DETERMINATION OF IMPORTANT ANTHROPOMETRIC FACTORS ON RADIAL PQCT BMD MEASUREMENTS IN FEMALES AGED 18-90.	74
5.1	Introduction	74
5.2	Study Population	74
5.3	Statistical Analyses	75
	-	_

х

したでは父母をいた

Construction of Con-

なりてい、うちないというたいであったいない

Page

CHAPTER 6 COMPARISON OF RADIAL pQCT BMD 88 MEASUREMENTS WITH MEASUREMENTS AT OTHER SKELETAL SITES: IMPLICATIONS FOR THE USE OF pQCT IN SCREENING FOR LOW AXIAL BMD. 6.1 Introduction 88 89 6.2 Study Populations and Bone Mass Measurements б.З Statistical Analyses 90 6.4 Results 91 6.4.1 Demographic and Bone Mass Data 91 6.4.2 Intra-Technique BMD Correlations for 91 **PQCT** 6.4.3 Correlations 92 of POCT BMD with Measurements Bone Mass Measurements at Other Skeletal Sites 6.4.4 Comparison of the Proportion of 92 Females in the Lowest Ouartiles of the Different Measurements for the Perimenopausal Population (Group 1) 6.4.5 Proportional Cut-Off Figures for 93 pOCT Measurements in Detecting Low Axial BMD in the Perimenopausal Population :Comparison with Ultrasound Measurements (Group 1). 6.5 Discussion 94 CHAPTER 7 AGE RELATED CHANGES IN RADIAL DOCT 123 BMD MEASUREMENTS: COMPARISON WITH OTHER SKELETAL SITES. 7.1 Introduction 123 7.2 Study Populations and Bone Mass 124 Measurements 7.3 Statistical Analyses 125 7.3.1 Cross-sectional Designs 126

一日日、日本町町 日本 二日

ALC: NO AND

4

いたまで、これによりなない、いたいで、これでいいでは、「ないない」

ģ

CHAPTER	8	THE ABILITY OF PQCT TO DISCRIMINATE FRACTURE AND NON-FRACTURE POPULATIONS: COMPARISON WITH OTHER BONE MASS MEASUREMENTS	157
8.1		Introduction	157
8.2		Study Populations and Bone Mass	159
8.2.1		Vertebral fracture study	159
8.2.2		Hip fracture study	160
8.3		Statistical Analyses	161
8.4		Results	161
8.4.1		Vertebral fracture study	16 1
8.4.2		Hip fracture study	163
8.5		Discussion	164
CHAPTER	9	CHANGES IN PQCT BMD MEASUREMENTS IN RESPONSE TO DRUG THERAPY: COMPARISON WITH MEASUREMENTS AT OTHER SKELETAL SITES.	181
Q 1		Introduction	101
9.2		Study Populations and Bone Mass	182
<i></i>		Measurements	
9.3		Statistical Analyses	184
9.4		Results	1.84
9.5		Discussion	185
CHAPTER	1	0 THE EFFECT OF LONGTERM WARFARIN THERAPY ON RADIAL AND AXIAL BMD.	196
10.1		Introduction	196
10.2		Study Population and Bone Mass Measurements	197
10.3		Statistical Analyses	198
10.4		Results	198
10.5		Discussion	199

- 1 X

- C. .

· • • ·

xii

CHAPTER	11	THE CORTI BONE	EFFECT COSTERO	OF IDS OJ	DISEASE N APPEND: COSTMENO	AND ICULAR	207
		WOME ARTHI	N W XITIS: MEASUE	ITH COME	RHEUM	ATOID WITH	
11.1	Intr	oducti	lon				207

.

and the second second

11.2 S	tudy	Popul	ation,	C1	inical	208
А	Assessments		anđ	Bone	Mass	
М	leasuremen	ts				
11.2.1 S	tudy Popu	lation	n			208
11.2.2 C	Clinical Data					208
11.2.3 B	Bone Density Measurements					209
11.3 S	tatistica	l Ana	lyses			210
11.4 R	esults					210
11.5 D	iscussion					213

CHAPTER	12	OVERALL	CONCLUSIONS	AND	229
		DISCUSSION			

REFERENCES

and the second strategy of the second strategy of the

238

APPENDIX 1 Glossary of Abbreviations 276 APPENDIX 2 Papers Published From the 279 Thesis Work APPENDIX 3 Abstracts Published From the 280 Thesis Work APPENDIX 4 Presentations to Learned 282 Societies From the Thesis Work

xiii

and the state of the second second second second

LIST OF TABLES

ģ.

A State of the second stat

古野学会議員内保護・

Table	1.1	Risk factors for low bone mass.	43
Table	1.2	Clinical indications for assessment	44
		of bone mineral density.	
Table	3.1	Subject demographic and pQCT BMD	67
		data examining the effects of voxel	
		numbers upon the precision of pQCT	
		in three different study populations	C 0
Table	3.2	co-efficients of variation (CV) for	68
		different study pepulations	
Table	4 1	DOCT BMD measurements of the	73
tante	4 .1	dominant and non-dominant forearms	75
		in a study population and the male	
		and female subgroups.	
Table	5.1	Total and trabecular pOCT BMD values	82
		for our normal population compared	
		with the manufacturers reference	
		data.	
Table	5.2	Subcortical and cortical BMD values	83
		for our normal population.	
Table	6.1	Details of subjects in groups 1 and	98
		2 studied in chapter 6.	
Table	6.2	pQCT, DXA hip and spine BMD, and	99
		calcaneal ultrasound measurements of	
		the perimenopausal population (Group	
m- 1- 1 -	<i>c</i> 2		100
Tapre	0.0	pour BMD and DAA whole body (WB-BMD)	100
Tabla	6 A	and arm (WB-ARM) BMD of Group 2. $(ut-off walue and proportion of the second properties of the $	101
100010	0.1	nopulation below this value for each	TOT
		of the DXA, pOCT and ultrasound	
		measurements to detect all	
		perimenopausal women with lumbar	
		spine and femoral neck osteopenia.	
Table	7.1	Rates of change of pQCT BMD	138
		measurements in women aged 18-90	
		years.	
Table	7.2	Demographic, pQCT and DXA BMD data	139
		of premenopausal and postmenopausal	
		subgroups of perimenopausal women	
		(group 2).	
Table	7.3	Demographic and baseline pQCT and	140
		DXA BMD data of premenopausal and	
		postmenopausal women studied	
		rongreudinally.	

and the second first of the second second

A State of the Sta

- Table 7.4 Annualized rates changes in pQCT and141DXABMDmeasurementsofpremenopausalandpostmenopausalwomen studied longitudinally.
- Table 7.5 Relationshipsbetweenannualized142rates of change in pQCT and DXA BMDinpremenopausalwomenstudiedlongitudinally.
- Table 7.6 Relationshipsbetweenannualized143rates of change in pQCT and DXA BMDinpostmenopausalwomenstudiedlongitudinally.
- Table 8.1 Demographic data of the hip and171vertebral fracture groups with the
corresponding control groups.
- Table 8.2 Area under the receiver operator172curves indicating the power of pQCTand DXA to discriminate vertebralfracturefromnon-fracturepopulations.
- Table 8.3 Area under the receiver operator173curves indicating the power of pQCT,DXA and calcaneal ultrasound todiscriminate hip fracture from non-fracture populations.
- Table 9.1 Baseline clinical, pQCT and DXA data190for the groups treated with hormonereplacement therapy and etidronate.
- Table 9.2 Height and weight at months 0, 4, 8191and 12 in hormone replacementtherapy (HRT), and etidronate (ETD)groups.
- Table 9.3 Percentage changes in pQCT BMD for192groupstreatedwithreplacementtherapyandetidronate.
- Table 9.4 Percentage changes in DXA BMD for193groupstreatedwithhormonereplacementtherapyandetidronate.
- Table 10.1 Anthropometric, lifestyle and 202 warfarin related data for the WARFARIN and CONTROL groups.
- Table 11.1Characteristics of the control, no-219-20steroidandsteroidtreatedrheumatoidarthritisgroups.
- Table 11.2Important determinants of pQCT and221DXABMD, and calcaneal ultrasoundmeasurements in all subjects.

وريدهي وحالية وحداد الروار كرامة ففارت المادة

Table 11.3 Relationships between rheumatoid 222 arthritis disease parameters and pQCT BMD and calcaneal ultrasound measurements in the combined steroid and non-steroid treated rheumatoid arthritis groups.

a that a start a share a

and the second second

·.· . .

LIST OF FIGUR	ES	Page
Figure 1.1	Schematic representation of the bone multicellular unit (BMU).	45
Figure 1.2	Gender and age related changes in BMD, with T and Z-scores.	46
Figure 1.3	Age and gender related incidence of osteoporotic fractures.	47
Figure 2.1	The Stratec XCT-960 pQCT osteo- densitometer.	54
Figure 2.2	A graphic representation of the pQCT scan images and BMD values.	55
Figure 2.3	A woman undergoing DXA scanning of the lumbar spine and proximal femur using the Norland XR26-Mark II.	56
Figure 2.4	A Norland XR26-Mark II DXA lumbar spine scan showing the spinal image and BMD values.	57
Figure 2.5	A Norland XR26-Mark II DXA proximal femur scan showing the image and BMD values.	58
Figure 2.6	A Norland XR26-Mark II DXA whole body scan showing the image and BMD values. The left upper limb has also been selected as an additional region of interest (ROI).	59
Figure 2.7	A volunteer undergoing a calcaneal ultrasound measurement using the McCue CUBA Clinical scanner.	60
Figure 5.1	Age related distribution of radial pQCT total BMD for women.	84
Figure 5.2	Age related distribution of radial pQCT trabecular BMD for women.	85
Figure 5.3	Age related distribution of radial pQCT subcortical BMD for women.	86
Figure 5.4	Age related distribution of radial pQCT cortical BMD for women.	87
Figure 6.1	Relationship of pQCT total with subcortical and cortical BMD.	1.02

AMA 1, 111

Sec. 1

第1日本では、1日本1日本1日本1日

and the second

ļ

いたりというために、たちがというもの

and the second se

11日 の日本の日本の人

A Reference of the second

100

Protocontexts Townships

「おい、これではない」 シャング かいがた はんしんたい

100 C

100

Figure	6.2	Relationship of pQCT total and trabecular BMD, and cortical with subcortical BMD	103
Figure	6.3	Relationship of pQCT trabecular with subcortical and cortical BMD.	104
Figure	6.4	Relationship of pQCT total BMD with DXA lumbar spine and femoral neck BMD.	105
Figure	6.5	Relationship of pQCT total BMD with DXA femoral trochanter and Ward's area BMD.	106
Figure	6.6	Relationship of pQCT total BMD with calcaneal ultrasound attenuation and velocity.	107
Figure	б.7	Relationship of pQCT total BMD with DXA whole body BMD and ipsilateral whole arm BMD.	1.08
Figure	6.8	Relationship of pQCT trabecular BMD with DXA lumbar spine and femoral neck BMD.	109
Figure	6.9	Relationship of pQCT trabecular BMD with DXA femoral trochanter and Ward's area BMD	1.1.0
Figure	6.10	Relationship of pQCT trabecular BMD with calcaneal ultrasound attenuation and velocity	111
Figure	6.11	Relationship of pQCT trabecular BMD with DXA whole body BMD and insilateral whole arm BMD	112
Figure	6.12	Relationship of pQCT subcortical BMD with DXA lumbar spine and femoral neck BMD	113
Figure	6.13	Relationship of pQCT subcortical BMD with DXA femoral trochanter and Ward's area BMD.	114
Figure	6.14	Relationship of pQCT subcortical BMD with calcaneal ultrasound attenuation and velocity.	115
Figure	6.15	Relationship of pQCT subcortical BMD with DXA whole body BMD and ipsilateral whole arm BMD.	116
Figure	6.16	Relationship of pQCT cortical BMD with DXA lumbar spine and femoral neck BMD.	117

1 C 18

18.

21

12.11

「「「「「「「「「「「」」」」

The Parks

and the second second

「京都市」の言語

100 C

Sec. 4

Constraint and the second

É

South and the second second

and the second

Figure	6.17	Relationship of pQCT cortical BMD with DXA femoral trochanter	118
		and Ward's area BMD.	
Figure	6.18	Relationship of pQCT cortical BMD with calcaneal ultrasound attenuation and velocity.	119
Figure	6.19	Relationship of pQCT cortical BMD with DXA whole body BMD and ipsilateral whole arm BMD.	120
Ricerco	6 20	Propertions of women in the	121
rigure	0.20	Proportions of women in the	ТУТ
		measurements also in the lowest quartiles of other mutually exclusive DXA measurements and pQCT BMD measurements.	
Figure	6.21	Proportions of women in the	122
		lowest quartile of pQCT BMD	
		measurements. also in the	
		lowest quartile of calcaneal	
		ultrasound attenuation and	
		velocity measurements	
Fimme	7 1	Age related changes in pOCT	144
riguie	,. ±	total BMD	*17
Rimmo	7 9	Age related changes in pOCT	145
** g ure	,.4	trabecular BMD	
Figure	73	Age related changes in pOCT	146
radre		subcortical BMD	1.4U
Figure	7.4	Age related changes in pOCT	147
riguie		cortical BMD	
Wiguro	75	Changes in pOCT total and	149
	,	trabecular BMD in	110
		postmenonaugal women ag a	
		function of duration diago the	
		TUNCTION OF QUIATION SINCE THE	
m 1	7 6	Menopause.	
Figure	/_0	Changes in poer subcortical and	149
		cortical BMD in postmenopausal	
		women as a function of duration	
		since the menopause.	
Figure	7.7	Differences in pQCT BMD values	150
		between premenopausal and	
		postmenopausal subgroups of	
		perimenopausal women.	
Figure	7.8	Differences in axial DXA BMD	151
		values between premenopausal	
		and postmenopausal subgroups of	
		perimenopausal women.	

Figure	7.9	Annualized rates of change in pQCT BMD values for premenopausal and postmenopausal women studied	152
Figure	7.10	longitudinally. Annualized rates of change in axial DXA BMD values for premenopausal and	1.53
Figure	7.11	postmenopausal women studied longitudinally. Relationships between the annualized rates of changes for radial pQCT total and DXA lumbar spine BMD in	154
Réaussa	7 10	premenopausal and postmenopausal women studied longitudinally.	160
Figure	7.12	Relationships between the annualized rates of changes for radial pQCT total and DXA femoral neck BMD in premenopausal and postmenopausal women studied longitudinally.	155
Figure	7.13	Relationships between the annualized rates of changes for DXA lumbar spine and femoral neck BMD in premenopausal and postmenopausal women studied longitudinally	156
Figure	8.1	Differences in pQCT BMD values between vertebral fracture and non-fracture groups.	174
Figure	8.2	Differences in axial DXA BMD values between vertebral fracture and non-fracture groups.	175
Figure	8.3	Receiver operator curves for axial DXA and pQCT BMD measurements discriminating vertebral fracture from non- fracture groups	176
Figure	8.4	Differences in pQCT BMD values between hip fracture and non- fracture groups.	177
Figure	8.5	Differences in DXA hip BMD values between hip fracture and non-fracture groups.	178

e en la servició de l

7

The second s

.,

いましてた いちの おおにあいやい 山

.

ter in the second structure of the

Figure	8.6	Differences in calcaneal ultrasound values between hip fracture and non-fracture	179
Figure	8.7	Receiver operator curves for hip DXA, pQCT BMD and calcaneal ultrasound measurements discriminating hip fracture	180
Figure	9.1	Longitudinal percentage changes in pQCT and axial DXA BMD measurements in response to hormone replacement therapy.	194
Figure	9.2	Longitudinal percentage changes in pQCT and axial DXA BMD measurements in response to cyclical etidronate.	1 9 5
Figure	10.1	Differences in pQCT BMD values between warfarin and control groups.	203
Figure	10.2	Differences in axial DXA BMD values between warfarin and control groups.	204
Figure	10.3	Relationship between pQCT trabecular BMD and warfarin duration, dose and INR.	205
Figure	10.4	Relationship between DXA lumbar spine BMD and warfarin duration, dose and INR.	206
Figure	11.1	Axial DXA BMD measurements of controls, non-steroid treated and steroid treated women with rheumatoid arthritis.	223
Figure	11.2	pQCT BMD measurements of controls, non-steroid treated and steroid treated women with rheumatoid arthritis.	224
Figure	11.3	Calcaneal ultrasound measurements of controls, non- steroid treated and steroid treated women with rheumatoid arthritis	225
Figure	11.4	Relationship between pQCT trabecular BMD and selected disease variables of rheumatoid arthritis.	226

	₽	a	g	e
--	---	---	---	---

でいた。 「「「「「」」」では、「」」では、「」」では、「」」では、「」」では、「」」では、「」」では、「」」では、「」」では、「」」では、「」」では、「」」では、「」」では、「」」では、「」」では、「」」では、「」」では、「」」 「「」」

Figure	11.5	Relati	onship	between	calcaneal	227
		BUA	and	selected	disease	
		variat	oles	of	rheumatoid	
		arthri	tis.			
Figure	11.6	Relati	.onship	between	calcaneal	228
		VOS	and	selected	disease	
		variah	les tic	of	rheumatoid	
		للبيدة فكما ملدة ت	، ت السالي ال			

CHAPTER 1

INTRODUCTION

an an ghairte an g

and the second second second

. . . .

シャンスクライ オーキークロップ

clinical This thesis examines the role \mathbf{of} peripheral quantitative computed tomography (pQCT) in the expanding field of osteoporosis. pQCT is a relatively new method of measuring bone mineral density (BMD) at appendicular sites, and when the work for this these was undertaken, there was very little published work on its use. It is a technique which offers potential advantages over existing methods by assessing trabecular and cortical bone compartments independently at an easily accessible peripheral site. This facility is important as trabecular and cortical bone behave differently depending upon the prevailing circumstances. In addition. pQCT measurements are truly volumetric, and therefore reflect true BMD, unlike many other techniques which "correct" bone mineral content for the projected area. While several peripheral sites are accessible to pQCT, work for this thesis was limited purely to the ultra-distal radial site. Osteoporosis is recognised as a huge clinical problem and a considerable drain upon NHS resources. Delivery of effective health care for those suffering from osteoporosis includes provision of facilities to measure BMD (1.1). The purpose of this thesis was to evaluate pQCT in relevant areas of clinical care for those suffering from osteoporosis, in an attempt to define a niche for its use.

Understanding the fundamentals of bone pathophysiology and biomechanics; osteoporosis - its definition and aetiologies; and osteo-densitometry - particularly present day techniques and defined roles, was essential in planning the work

presented hereafter. The remainder of the introduction sets out present day knowledge of these matters.

2 11 2 CA.

13

1.12.4

, Sj

ŝ

Ş

単にないた

Sec. 1

-

Ì.

1.1. Bone

1.1.1 Normal Cellular Function

Bone is the living tissue that forms the skeleton, providing a mechanical framework of support. Skeletal growth occurs during childhood and adolescence, and involves bone modelling by either endochondral or intramembranous ossification. During adult life, bone undergoes the continuous process o£ the changing needs remodelling in response to of an . individual. Remodelling comprises of a series of complex interactions between cells and the extracellular matrix. Two processes are responsible for remodelling - bone resorption where osteoclasts remove bone, and bone formation where osteoblasts lay down bone. The opposing functions о£ osteoclasts and osteoblasts are closely linked by a phenomenon known as "coupling", with resorption then formation occurring in discrete areas known as "basic multicellular units" -(BMU). The process of coupling within a BMU is depicted in figure 1.1. Arrival of osteoclasts on the bone surface signals the initiation of bone remodelling. Osteoclasts are derived from haematopoietic granulocyte-macrophage fibroblast colonyforming units (1.2,1.3) which also produce monocytes and macrophages. The point at which they become committed to the osteoclast lineage is not clear, but osteoclast progenitors migrate from bone marrow to bone through the circulation or by direct migration (1.3). Having arrived on the bone surface, osteoclasts remove a quantum of bone leaving a "resorption pit". This is achieved through the expression of many features of the macrophage (1.3). Bone resorpting agents are secreted

into the extracellular space and bone is broken down. Internalization of the products occurs through phagocytosis before being further processed (1.2). The process of coupling attracts osteoblasts into the resorption cavity. Osteoblasts are derived from the pluripotent mesenchymal stem cells of the (1.2, 1.3),which can differentiate into bone marrow fibroblasts, chondrocytes, adipocytes, and muscle cells. The progenitors which produce these cell lineages are known as fibroblast colony-forming units (1.3). The main function of osteoblasts is secretory. Deposition of the extracellular within the resorption cavity that matrix occurs is subsequently mineralised. The synthesis, deposition and mineralization of the extracellular matrix require the orderly expression of several genes, their correct expression being prerequisite for the next (1.4). Type I collagen is the major matrix component secreted by the osteoblast and the main structural protein of bone. Its expression occurs early, during the proliferative phase of osteoblast function. Synthesis of many other non-collagenous proteins occurs. То name but a better known few, alkaline phosphatase and osteopontin are expressed later during the osteoblast matrix maturation phase. Osteocalcin is expressed even later during its mineralization phase (1.4). Filling of the resorption cavity is achieved with completion of osteoblast function. The surface is then covered with flattened cells that line the surface of all bone that is quiescent (1.2). Remodelling occurs continuously throughout the skeleton during adult life. Inspection of the surface of a section of bone reveals several resorption hollows at different stages surrounded by areas of quiescent bone. The control of osteoclast and osteoblast function and its coupling is extremely complex, involving

an su character a charac

ふうしょうぶん かやくない

ġ.

Ű.

13 A A A

cellular interplay, the interaction of many cytokines and hormones, and local mechanical factors (1.2,1.3).

ふうだいせい とうかい きん

1.1.2. Bone Types, Skeletal Distributions and Biomechanical Properties

Division of bone is possible into two distinct types - cortical and trabecular.

Cortical bone makes up approximately 80% of the skeleton. It consists of a series of irregularly spaced, overlapping cylindrical structures known as Haversian systems, each comprising a central canal surrounded by concentric layers of bony tissue. Remodelling of cortical bone occurs at periosteal and endosteal surfaces, and within the Haversian canals. Increased Haversian system resorption will cause increased porosity of cortical bone, while increased periosteal and endosteal resorption results in increased numbers of surface BMU's (1.5). The proportion of cortical and trabecular bone varies within individual bones and according to skeletal site. Cortical bone is found primarily in the diaphyseal region of long bones in the appendicular skeleton, with metaphyseal areas containing a smaller proportion. With increasing age, exceeds periosteal resorption resulting endosteal in progressive cortical thinning, an increased intramedullary space and a small increase in the diameter of the bone (1.5,1.6). There is considerably less cortical bone in the axial skeleton, where trabecular bone predominates. However, within the clinically important vertebrae, the vertebral body has a thin envelope of cortical bone, and the posterior are predominantly cortical bone elements (1.5, 1.7),contributing significantly to antero-posterior lumbar spine bone mass assessments - 53% in women and 44% in men (1.7).

1. 12 T

Trabecular bone comprises approximately 20% of the skeleton, and consists of a meshwork of interconnecting trabeculae. It is found predominantly in the axial skeleton within the vertebral bodies and flat bones, and within the metaphyseal region of lone bones in the appendicular skeleton (1.2,1.5). Due to the greater surface area to volume ratio of trabecular bone, it is metabolically more active and responds more quickly to stimuli than cortical bone (1.2,1.3,1.5,1.8). Consequently, about 25% of trabecular, compared to 3% of cortical bone is remodelled each year during adulthood (1.3,1.9). Of great importance is the effect of the menopause upon bone quantity and quality. Ovarian failure and reduced levels of circulating oestrogen in women results in increased activation of BMU's, with increased bone resorption and a deficit in bone formation within each BMU (1.10,1.11). The net effect is bone loss, which preferentially effects trabecular bone. Not only is the BMU activation frequency increased, but the depth of resorption cavity is also increased (1.11,1.12). This results in increased perforation of trabecular plates, decreased connectivity and disruption of the trabecular lattice structure, and subsequent reduction in load carrying capacity (1.11,1.13).

ģ

Sec. 1

Also of importance is the presence of bone marrow, which is in close proximity to trabecular bone within the medullary cavities of bone. During childhood and adolescence, red or haemopoietic marrow is present in all bones. By the age of 20, marrow in the long bones has become inactive, being replaced by fat, which is then known as yellow marrow (1.14). This

occurs in all long bone with the exception of the upper femur and humerus. In addition to the upper humerus and femur, haemopoietic marrow persists in the axial skeleton. However, marrow fat increases by about 7% per decade in the vertebral body (1.15). Consequently, the proportion of active marrow in elderly females with vertebral fractures is as low as 25% (1.5). This age related change in the distribution and type of bone marrow may have important implications for the measurement of BMD, and the response of bone to different stimuli.

したが無人 かかえい たぜした かがくしん

東京の学会を

のないではない

三部市 古いたのとあた

1000

and the state of the second second

A 4 14 14

In this thesis, the radius, lumbar spine, proximal femur and os calcis are the main skeletal sites studied, and the proportion of cortical and trabecular bone varies with skeletal site. At the radius there is variation in the proportion of cortical and trabecular bone along the radial length. Four common measurement sites are encountered in the literature, these being denoted by the percentage of the total radial length, proximal to the most proximal point of the radio-carpal joint that the measurement occurs. The most distal measurement site is the 4% "ultra-distal" site where typically 50-55% of bone is trabecular (1.16,1.17). pQCT measurements occur at this site. At the 8-10% "distal" site, characteristically 25% of the bone mass is trabecular, and less than 10% at the 33% "1/3", and 50% measurement sites (1.16). The radius has been the subject of extensive study previously, mainly using single photon absorptiometry (SPA). This technique and others are discussed in greater detail later in this chapter (see section 1.3). It is worth stating at this point however, that SPA measurements occur at the distal, 1/3 and 50% measurement sites where the proportion of

trabecular bone is much lower than that found at the ultradistal "4%" site. It is also worth noting that there are dramatic changes in the proportion of cortical and trabecular bone within the distal radius, which occur within small distances (1.16, 1.17, 1.18, 1.19). The proportion of trabecular bone in the lumbar vertebral body is about 65% (1.2, 1.5, 1.20). In the proximal femur, the proportion is about 25-30% at the femoral neck (1.2, 1.5, 1.20), increasing to approximately 50% within the trochanteric region (1.20). The proportion at the os calcis is 90-95% (1.21).

ę

and the second from the state of the

「「「「「「「「「」」」」

Bone mineral density contributes about 75-85% of bone strength (1.22), with both cortical and trabecular components being important (1.22). A decline in bone mass results ìn a disproportionately greater decrease in bone strength (1.9,1.22). Bone mass of the spine (1.9,1.13,1.23,1.24,1.25), proximal femur (1.26,1.27,1.28) and distal radius (1.29,1.30) is strongly related to bone strength. For trabecular bone, both bone density (1.13,1.31,1.32,1.33) and structure (1.9,1.22) are important in determining its strength. Halving the cross-sectional area of a vertical trabecular column, or doubling the effective vertical length of the column through loss of horizontal support reduces its carrying capacity by 75% (1.22). Although trabecular bone is of great importance, cortical bone the spine (1.31,1.34), at femoral neck (1.26, 1.35) and radius (1.36) are also important in determining the overall strength of bone. More pertinent in the context of this thesis, both cortical shell geometry and trabecular BMD at the ultra-distal radial site (1.36), and cortical BMD and thickness at the distal 1/3 site (1.37) as assessed by pQCT, is related to fracture force.

1.1.3 Normal Age Related Bone Mass Changes

A simple graphic representation of gender and age related changes in BMD is shown in figure 1.2. Skeletal growth and bone mass increases during childhood (1.38,1.39,1.40,1.41). During puberty, there is rapid skeletal growth matched by an increase in bone density (1.41,1.42,1.43,1.44,1.45). Marked accumulation occurs at the lumbar spine during puberty (1.43,1.44), while a more linear increase has been found at the radius (1.40), albeit assessed by SPA. It is possible that accretion of cortical and trabecular BMD during childhood and adolescence may differ. This issue cannot be adequately addressed using measurement techniques which measure areal BMD, such as SPA and DXA. Maximum accretion of bone mass occurs during early adulthood, and the maximum potential in an individual is known as peak bone mass (PBM).

그 너희 물리 동안에는 가동 지하는 것 같이 다 있는 것 같이.

the pathogenesis PBM is an important concept ìn of osteoporosis, as fracture risk in later life is considered (although not yet proven) to be dependent upon PBM, and the rate of subsequent bone loss in later life (1.46,1.47,1.48). PBM in men is generally accepted as being greater than that While this in women. is accepted at the radius (1.40,1.49,1.50,1.51,1.52), it is more controversial at the having found greater bone in males spine, some mass (1.53,1.54), others having found no gender difference (1.52,1.55,1.56). It is thought that areal BMD measurements acquired using dual energy x-ray absorptiometry (DXA) and dual photon absorptiometry (DPA), rather than volumetric BMD measurements using QCT may account for in part these discrepancies, as the greater bone size in males results in an overestimation of areal BMD gender differences (1.57).

However, even data acquired using volumetric measurements have been conflicting (1.53,1.54,1.55,1.56). PBM is predominantly influenced by genetic factors, which account for about 80% of the population variance (1.58,1.59). Other factors such as parity and lactation, oral contraceptive use, diet, physical activity and smoking have been shown to have an inconsistent and variable effect upon PEM (1.49,1.60). There is controversy regarding the timing of PBM (1.46). Based upon older, crosssectional data, the consensus was that PEM was attained by the end of the third decade (1.46,1.51). Recent cross-sectional (1.44,1.60,1.61,1.62,1.63,1.64) and longitudinal (1.45,1.65) studies with greater numbers of subjects suggest that PEM is achieved by the end of the second decade.

1 A 44 A 7 A

1997 - 1997 - Alf

にあたり

Ċ,

いたながら、ためになったいというないないというというというというない ちょうちょう

「たちにない」の言語を見ていた。「「「「「「「「」」」」、「」」

÷

in the Aligneet Aligneet States of the

After acquisition of PBM, bone mass is generally considered to be maintained until the latter part of the fifth decade, after which continuous trabecular and cortical bone loss occur at all skeletal sites, in all individuals of both sexes (1.47, 1.49, 1.51, 1.64, 1.66, 1.67, 1.68, 1.69, 1.70, 1.71, 1.72, 1.73). Following the menopause, women experience a phase of accelerated bone loss which preferentially effects trabecular bone (1.49, 1.64, 1.66, 1.67, 1.74, 1.75, 1.76, 1.77). This lasts approximately ten years and is caused by ovarian failure and oestrogen deficiency. There is controversy regarding the timing of bone loss at different skeletal sites. Most would agree that radial BMD is stable until the end of the fifth decade, or until the onset of the menopause (1.50,1.51,1.63,1.68,1.69,1.70,1.78,1.79), although others have shown bone loss in premenopausal women in a longitudinal study (1.80). The issue is less clear at axial sites, where some cross-sectional studies have documented reductions in BMD

at the hip (1.68,1.72,1.81) and spine (1.51,1.74) earlier than the fifth decade, as have longitudinal studies at the spine (1.70,1.82,1.83). If premenopausal bone loss does occur, it is considerably less than postmenopausal loss, and much less important (1.84).

Controversy also exists about bone loss in the elderly. Some work has suggested that bone loss may cease or actually increase at the spine, hip and radius (1.73,1.85). This assumption however, has been called into question. Several factors such as selection of a fracture free population, secular trends in BMD, mortality of fracture patients, and accuracy errors of BMD measurements due to concomitant osteoarthritis and marrow fatty changes, confound the study of the elderly. All these confounding factors (except marrow changes) will lead to an over-estimatation of BMD, and an under-estimatation of the rates of bone loss in the elderly (1.86).

1.1.4 Describing BMD Measurements: T-Score and Z-Score

It is important to describe BMD values in such a way that is easy to understand, and that gives as much information as possible without being misleading (1.87,1.88). There are several approaches to this.

- The absolute BMD value. This alone has limitations as there are age and sex related differences in absolute BMD.
- The percentage of the expected age and sex matched value.
 Although easy to understand, it has limitations and can be misleading. With ageing the mean value decreases, and
hence a decreasing denominator results in a wider range of percentage values representing the normal range. This especially important if serial measurements are is expressed as a percentage relative to the age related normal mean (not to be confused with expressing the percentage change relative to the baseline measurement). З. The percentile of the expected age and sex matched value. This is also easily understood, does not depend upon a normal distribution of data, and is not misleading in comparison to aqe matched controls or serial measurements. Unfortunately, expression of BMD as a percentile make age related comparisons difficult, and at present, they are not widely used by the manufacturers of osteo-densitometer.

4. A unit of standard deviation (SD) from the normal mean. Data has to be normally distributed to allow BMD expression in this manner. Data not distributed normally require appropriate transformation. Population studies have demonstrated a Gaussian distribution of BMD values at all ages, in both sexes, at all skeletal sites when large enough numbers of subjects are studied. Comparison with other age groups is convenient and not misleading, but many clinicians are not familiar with data expression in SD units. Two different scores are commonly used. These are described below and graphically represented in figure 1.2.

Z-score this value is the number of SD's above or below the sex and age related normal mean.

許公式

12 NY 153

an a shirth basis shirth

an an she car a she an an an agusa.

T-score this value is the number of SD's above or below the mean value of young normals.

These values are now frequently used, and as discussed later in this chapter (chapter 1.2.1), the T-score is central to the definition of osteoporosis.

1.2 Osteoporosis

1.2.1 History and Definition of Osteoporosis

Only in recent years has osteoporosis received attention from the medical profession, and achieved public awareness that it warrants. The term "osteoporosis" was introduced to the medical literature in the 18th century (1.89). Its use implied a histological diagnosis of "porous bone" which was later refined to signify a reduced quantity of bone which was normally mineralised. This formed the basis for the first modern attempt to redefine osteoporosis using population based bone mass measurements performed by newer osteo-densitometers. Osteoporosis was thus defined as a BMD less than two standard deviations (SD) below the mean of young normal women i.e. a Tscore less than -2 (1.90). This definition is solely dependent upon reduced bone mass, yet is it clear the structural deterioration in bone tissue is also of great important in determining bone strength. Moreover, few would argue that the most important end point of reduced bone mass, is fracture. This also is neglected in the above definition. Subsequently, the definition of osteoporosis was changed in 1991 to the following:

の語言を描述する

Osteoporosis is a disease that is characterized by abnormalities in the amount and architectural arrangement of bone tissue, which lead to impaired skeletal strength and an increased susceptibility to fracture (1.91).

Although this definition encompasses all the important concepts of osteoporosis, its clinical and research application is limited by being purely qualitative. To overcome this problem, the World Health Organisation recently

devised an operational definition which is based upon BMD and fracture occurrence (1.92).

化化学学 化结构 计可能存在 化分子 经公司管理 化合物 化分子

OSTEOPENIA

BMD more than 1 SD below the young normal mean, but less than 2.5 SD below.

i.e. -1.0 > T-score > -2.5

OSTEOPOROSIS

BMD more than 2.5 SD below the young normal mean, <u>without</u> fracture.

27年にの一日の一部の一部のの一部である。

i.e. **T-score** < -2.5 (no fracture)

SEVERE OSTEOPOROSIS

BMD more than 2.5 SD below the young normal mean, with fracture.

i.e. **T-score < -2.5** (plus fracture)

However, limitations persist even for these definitions. Structural considerations have been omitted, and the diagnoses of osteopenia and osteoporosis (without fracture) are dependent upon BMD measurement. Osteo-densitometry is therefore essential for the diagnosis of osteoporosis without fracture, a service not readily available to all at the present time in the UK. A further problem in definition is posed by intra-individual variation in BMD. Due to this, an individual can be classified as osteoporotic according to BMD at one skeletal site, yet not based upon another.

Consequently, the prevalence of osteoporosis in Caucasian women varies from 16% to 30% depending upon the number of skeletal sites assessed (1.92). Inspite of these limitations, this is the definition of osteoporosis which prevails at the present time. 「「日本」の「日本」のないないで

日本の後に、町にから市場が長いていたのであった。 時日本の一日前日

2.00

1.1.2.1.2

「「「「「「「「」」」を見ていている。

经济股份 化氯化化氯化化化

Some idea of the magnitude now posed by osteoporosis has been shown by the applications of these definitions to a local population American in Rochester, with subsequent extrapolation nationwide. It has been estimated that 16.8 million (54%) postmenopausal white American women have osteopenia, and a further 9.4 million (30%) have osteoporosis from this calculation (1.93). Comparable figures from other populations (notably the UK) are not yet available, but these figures serve to illustrate the magnitude of the problem posed by osteoporosis.

1.2.2 Osteoporotic Fractures

Osteoporotic fractures are characterised by incidence rates greater in females than males, rates which increase steeply with age, and a tendency to occur in skeletal sites richer in trabecular bone (1.94). The sites which are typically affected are the hip, thoracolumbar spine and distal radius, although fractures of the proximal humerus and distal femur also have the above characteristics (1.95). There are considered to be two distinct types of osteoporosis, leading to different fracture types (1.67). Type I osteoporosis (and fracture) is caused by ovarian failure, reduced levels of circulating oestrogen and increased bone loss. Trabecular bone is preferentially effected and greater bone loss occurs at skeletal sites richer in this bone type - namely the spine and

distal radius. Ultimately, this predisposes to vertebral and Colles' fractures (1.67). Type II osteoporosis (and fractures) results from age related reduction in both cortical and trabecular bone. This affects all individuals of both sexes, hip fractures (1.67).and predisposes to Graphic representation of the age and gender related incidence of hip, vertebral and distal forearm fracture is shown in figure 1.3. Caucasian women and men have a lifetime fracture risk of 17.5% and 6% respectively for hip fracture, 15.6% and 5.6% for clinically diagnosed vertebral fracture, and 16% and 2.5% for distal forearm fractures (1.96). Determining the exact prevalence and incidence of vertebral fracture is complicated by asymptomatic events, and the application of differing fracture criteria (1.97). This is discussed in greater detail in section 1.2.2.2. However, from the age of 50 years onwards, the life-time risk for any of the three fractures is 39.7% for women and 3.1% for men (1.96). Consequently, in the UK around 50,000 hip fractures, 40,000 clinically diagnosed vertebral fractures and 50,000 distal forearm fractures will occur annually in women (1.12), costing the National Health Service an estimated £740 million (1.1). With an increasingly aged population, these figures are set to rise (1.98). Specific osteoporotic fractures are discussed briefly below.

二文 いとなたい とたれ とやし さめい

A SAME

1.2.2.1 Hip Fracture

Hip fracture is the most serious of the osteoporotic fractures. Its incidence rises exponentially with age, and above the age of 50 there is a females to male ratio of 2:1 (see figure 1.3). The lifetime age and sex related hip fracture rates are higher in Caucasian compared to Black or Asian populations (1.99), the female predilection also being

absent in the latter populations (1.95). There are also regional differences in hip fracture incidence between (1.95,1.100,1.101), and within countries (1.95). There is significant morbidity and mortality associated with hip fracture, with the relative survival rate at 5 years being 0.83 (1.94). The majority of deaths occur within 6 months of fracture, and are related to concomitant illnesses (1.102), although some are due to complications of the fracture or its surgical management (1.103). Aetiological factors in the pathogenesis of hip fracture are discussed further in chapter 8.

1.2.2.2 Vertebral Fracture

Inspite of being synonymous with the diagnosis οĒ osteoporosis, information on the epidemiology of vertebral fracture remains scanty. This is in part due to clinically silent fractures, which necessitates clarification of the distinction between "fracture" and "deformity". "Fracture" applies to morphometric abnormalities detected as а consequence of symptoms, usually back pain or loss of height. "Deformity" refers to abnormalities detected from morphometric studies which have been performed regardless of clinical symptoms (1.95). Gathering of epidemiological information has also been hindered by the lack of universal agreement on the definition of a vertebral fracture. There are many differing definitions based upon vertebral body morphology (1.104,1.105,1.106,1.107,1.108,1.109). There are two main methods of determining deformities/fractures at the present time. Quantitative method relies solely upon vertebral height measurements. Semi-quantitative methods require inspection of the x-ray by a radiologist as well as measurement of vertebral

South Street

heights. Semi-quantitative methods have the advantage of allowing differentiation of change due to conditions other than osteoporosis (such as Schuermanns disease) from true vertebral fracture. However, it is also highly dependent upon the experience of the radiologist, and reproducibility can be a problem, especially if criteria are not laid down. As each vertebral body has characteristic dimensions (1.104), probably widely accepted definitions the most incorporate a distribution of vertebral body dimensions for each spinal level, with cut-off values defining a fracture (1.95,1.106,1.107). However, to date there is no agreement as to which of these numerous methods of defining a vertebral fracture is best.

に加える。 ないたい 生活 生活 生活 一日 「「「「「「」」」

The second second second

alitication and an index

1

: #

のないので、「「「「「「」」」

あるかられないない あため 二方子 一方子 ないかいい

man production of the state

The incidence of clinically diagnosed vertebral fractures is shown in figure 1.3. In men the incidence rate increases exponentially, whilst in women there is a more linear increase. The most common spinal level for fracture is the lower thoracic and upper lumbar spine (1.95), most occurring as a result of minimal trauma. It has long been appreciated that morbidity is associated with vertebral fracture. Now there is recognised to be significant associated mortality, with a relative survival rate of 0.82 (1.110).

1.2.2.3 Distal Fore-arm Fracture

The incidence pattern of this type of fracture differs from that of hip and vertebral fractures (see figure 1.3). In women, the incidence rate increases linearly between the ages of 40 and 65 years before stabilising. In men, there is a linear increase from the age of 20 to 80 years. The majority of fractures occur in females (females:male ratio of 4:1).

Although morbidity arises from this fracture, there is no associated mortality (1.95).

A statute of the statute

しゃか わってい かいかい われた ひょうひかい しかく

のかけのはなからであるとうのないない

いいのであるというないのであるが

Risk Factors and Causes of Secondary Osteoporosis 1.2.3.1 The importance of ageing in type 2 osteoporosis, and the menopause, ovarian failure and hypoestrogenaemia in type 1 osteoporosis has already been eluded to in the preceding sections of the introduction. Table 1.1 lists the large numbers of lifestyle factors, concomitant diseases and drugs which adversely affect bone and predispose to osteoporosis (1.11,1.49,1.111). They are too numerous to discuss in detail in the introduction of this thesis, although some are discussed in greater detail in the relevant chapters of this thesis. Worthy of further discussion at this point however, is corticosteroid induced osteoporosis which is probably the commonest cause of secondary osteoporosis. Unfortunately, it was not possible to study its effect on bone per se, although its was studied in patients with rheumatoid arthritis, work which is presented in chapter 11. Corticosteroid induced osteoporosis is therefore discussed below.

1.2.3.2 Corticosteroid Induced Osteoporosis.

Direct Bone Effects.

Corticosteroids uncouple bone metabolism in favour of resorption, at least early in the course of therapy. Osteoblast function and precursor differentiation are directly inhibited, whilst the rate of mineral apposition is reduced resulting in inhibition of bone formation (1.112). Conversely, bone resorption is probably increased, although the evidence supporting this is less convincing and there is no convincing evidence that corticosteroids directly stimulate osteoclasts.

However, increased numbers of osteoclasts have been demonstrated histomorphometrically (1.113). It is thought that the increased osteoclast activity is a consequence of mild secondary hyperparathyroidism induced by depressed intestinal calcium absorption (1.114) and increased urinary calcium excretion (1.115). Thus the overall effect of corticosteroid is to reduce bone formation and probably increase bone resorption resulting in net bone loss (1.116).

The set of the set

Indirect Hormonal Effects

.

Adequate levels of oestrogen and testosterone are necessary for healthy bone. Corticosteroids have been shown to reduce circulating cestrogen levels by blunting the LHRH dependant pituitary release of LH (1.117), and the FSH dependant gonadal oestrogen (1.118,1.119) release of and testosterone (1.120,1.121). Likewise, there is evidence that adrenal androgens are important in maintaining bone density, with low dehydroepiandrosterone (DHEA) levels predicting reduced femoral BMD in post menopausal females (1.122,1.123). Corticosteroids cause ACTH suppression and adrenal atrophy, reducing androstenedione and oestrogen levels (1.124).

Vitamin D plays an important role in calcium homeostasis and bone metabolism, but as yet its role in corticosteroid induced osteoporosis in unclear. The inhibition of intestinal absorption would appear to be independent of Vitamin D status (1.125, 1.126). The importance οf mild secondary hyperparathyroidism has already been eluded to, net calcium loss through decreased intestinal absorption and increased renal excretion being stimulus the (1.114, 1.115). Normocalcaemia is usual, and although conflicting levels of

PTH (normal or high) have been demonstrated, bone biopsies showing increased osteoclast activity(1,113) and increased nephrogenic cAMP measurements(1.127) support the contention that corticosteroids induce a state of mild secondary hyperparathyroidism. Thus corticosteroids have a profound effect on a variety of hormones involved in bone metabolism with changes favouring bone loss.

10.00

A STATE OF A

j. U

「大学」のないので

us théilitea

3

Effects upon Bone Mineral Density and Fracture.

and the second second second

High dose corticosteroid therapy can cause annual bone loss of 5-15% (1.128,1.129,1.130). However, not all patients lose bone (1.129,1.130), and recovery after cessation is possible (1.131). Although both trabecular and cortical bone are adversely effected, greater loss occurs from the former (1.128,1.132,1.133,1.134). The long term consequences of this is an increased fracture rate of femoral neck, rib and particularly thoracolumbar spine (1.116,1.135). The true incidence of fracture at this latter site has not been accurately defined, because painless fractures often escape detection.

Bone loss occurs most rapidly during the first few months of corticosteroid therapy, with subsequent slower, but continued bone loss(1.128,1.136,1.137). Low dose therapy (less than 7.5mg/ day prednisolone) may have less effect on premenopausal than postmenopausal women or men(1.138). Larger doses however, affect all population groups irrespective of age, menopausal status, sex or racial origin, with evidence suggesting that children and postmenopausal women lose bone especially quickly(1.139). The latter group are most at risk of fracture presumably due to the additive effect of oestrogen deficiency.

The cumulative dose has been found by most investigators to relate to the amount of bone loss, although a threshold value over which this becomes critical has not been established. Alternate day dosing offers no protection against bone loss(1.139), and all corticosteroids affect bone adversely although, equipotent doses of Deflazacort have been shown to have a less detrimental effect than other synthetic steroids (1.140). Corticosteroid induced bone loss occurs irrespective of the underlying condition being treated. Recent interest has focused on the effect of inhaled steroids used in the management of asthma on bone. Evidence is mounting that even this small dose may adversely affect bone(1.141,1.142). The longterm importance of this is yet to be established.

1.00 M

Same and the loss of the second

というない

Constraint and the second

00

・おおいるとしてないであるというで、いいですのか、いいのでいい

where is the state of the state.

1 A A A A A

Through quantification of trabecular bone, pQCT would be ideally suited to monitor the effects of corticosteroid. Ruegsegger et al (1.139, 1.143) have studied a small numbers of asthmatics treated with corticosteroid for 1 year and found dose related reductions in trabecular BMD at the radius up to 17.5%, while cortical BMD was unaffected. Beyond these small, initial studies there is a paucity of data examining the use of pQCT in monitoring the effect of corticosteroid on bone. At the present time, axial DXA BMD assessments are indicated for the management of patients likely to be treated with corticosteroid for more than one year (1.144,1.145,1.146). Further work is necessary to establish whether pQCT has a role in the management of corticosteroid induced osteoporosis.

1.3 Techniques Used to Assess Bone Mass

Development of methods to assess bone mass has contributed immensely to the recent increased awareness of osteoporosis as a serious health issue. Clinicians and the public alike, are now informed that there are methods available to assess bone mass, albeit some more than others. These methods must satisfy several criteria before they can be utilised clinically. They (reflect bone mass), precise be accurate true must (reproducible) in the short and longterm, acceptable to the patient and safe in terms of radiation exposure. A number of methods exists to assessing bone mass, some more readily available than others. These have been reviewed extensively before (1.88,1.147-1.154). Techniques such as neutron activation analysis (1.147, 1.155), compton scattering (1.147) and proton activation (1.147) are not routinely used. A new area of interest, which is still very much a research tool is quantitative magnetic resonance imaging. This may reveal information about trabecular bone microstructure as well as density (1.147,1.152,1.153,1.154). The following section is a summary of the commonly used, present day methods of assessing bone mass.

and the second of the second second

いたちがい いっかい しょうしょう かんかい たいちょう かいたい かいたい かいかい あんかい かんしょう かい

の日本のになってもです。

1.3.1 Radiographs

A plain x-ray can detect fracture, but is not sensitive enough to accurately assess BMD (1.5,1.156). Structural analyses of radiographs have been developed in an attempt to improve the diagnostic ability of x-rays.

The pattern of trabeculae in the proximal femur can be graded, and is known as the Singh index (1.157). With age related trabecular thinning, the grade changes from VI to I. Although

the index is lower in women with hip, vertebral of Colles fractures, it has been criticised by some (1.155,1.156), but not others (1.159) for its subjectivity and high interobserver variability. It has a low sensitivity but high specificity for diagnosing low bone mass, with a large overlap in BMD between grades (1.158,1.159). It may have a adjuvant role in conjunction with BMD, geometric or morphometric assessments in fracture prediction, but this requires further study.

如此是我们的是不是不能能能。"他们就是这些你们是我们的是你们的是你们的是你不能能有什么?""你们的你?""我们就是你们的你们的?""你们就是不能能让你的是我们?""你们也是你们

じょうえい だいしゃ

化乙酰氨基乙酰氨基乙酰氨基乙酰氨基

Radiogrammetry is a technique where a high quality anteroposterior x-ray of the hand is performed under standard conditions. Less frequently the radius, humerus, clavicle, tibia and femur have been studied. In the hand, numerous measurements from the metacarpal (usually the second) are used to calculate various indices. As a technique of assessing bone mass, it is limited to peripheral site, is time consuming, requires skilled personnel, assesses only cortical bone, detects only advanced osteoporosis and is therefore not routinely used today (1.5, 1.155).

Radiographic absorptiometry, which was previously known as photodensitometry, is a technique where an x-ray of a peripheral bone (usually phalanges, less commonly radius or tibia) is obtained under standard conditions together with an aluminium reference wedge. The density of bone on x-ray is then derived by comparison with the wedge using an optical densitometer. There has been some resurgence of interest in photodensitometry (1.160), with computer enhanced analysis and improved performance. Precision has improved to 1.5-2% (1.161,1.162) which is comparable with other scanning

modalities such as DXA. It is a readily available technique of low cost and low radiation dose. It correlates well with in radial SPA (1.161)vivo BMD as measured by and DXA (1.162,1.163), and moderately with axial hip and spine BMD as measured by DPA (1.161) and DXA (1.162,1.163,1.164). An age related decrease in the metacarpal index (one of the calculated parameters) has been documented (1.163),and evidence exists that it provides an estimate of vertebral and hip fracture risk (1.160,1.165). It may prove to be of value as a population screening tool, bearing in mind that DXA in not universally available.

いるないのないのないのないで

1000

and the second shares

1.3.2 Single Photon Absorptiometry (SPA)

SPA was developed in the 1960's. A highly collimated photon beam from a single energy radiation source (usually ¹²⁵I) is used to measure photon attenuation at the measurement site. Its use is limited to scanning peripheral sites as the scan site requires to be immersed in water to ensure uniform absorption of the single energy radiation from the surrounding tissues. The cortical rich radius is most commonly scanned, although the exact site can be varied to increase the proportion of trabecular bone. The distal femur and calcaneus can also be scanned. Precision is 1-3%, accuracy 5%, scan duration 15 mins with a radiation dose of 0.5-1.0 μ Sv. A recent development has been the replacement of the ¹²⁵I source with a mini X-ray tube, giving the advantage of a stable photon source with improved longterm precision. This variation is known as single energy x-ray absorptiometry.

1.3.3 Dual Photon Absorptiometry (DPA)

DPA overcomes the problems of soft tissue composition and

water immersion encountered by SPA, by using a double energy source which emits two different photons energies. These photon energies have different bone and soft tissue attenuation properties which can be subtracted from each other allowing the BMD of axial sites such as the proximal femur and lumbar spine to be quantified. Total body scans can also be Initially two isotopes (iodine: ¹²⁵I, performed. and ²⁴¹Am) were used, before the americanium: single isotope gadolinium (153Gd), with dual emission energies of 44 and 100keV, became the standard. Typically, precision values are 2-4%, accuracy 4-10%, scan duration 20 mins (45 mins for total body scans) and the radiation dose is around 1 μ Sv. Decay of its the radiation source requires replacement after approximately 18 months which can become expensive and further adversely affect precision. This lead to the replacement of isotopes by an X-ray source giving dual energy X-ray absorptiometry.

and the second secon

500 M N

からに、またいないないで、 「

「「「ないない」を見ていていた。

1.3.4 Dual Energy X-ray Absorptiometry (DXA)

DXA is now probably the most widely used scanning technique, and was introduced commercially in 1987. The radionucleotide source used in SPA and DPA was replaced with an x-ray tube. Two distinct energy level beams are generated by one of two methods. One method (used in Hologic scanners) uses rapid switching of the x-ray generator between high and low voltage settings. The other (used by Norland and Lunar scanners) produces a beam from a constant voltage, which is then filtered into two different energy beams. Measurements using DXA are possible at almost any skeletal site. Sites most commonly scanned are the antero-posterior and lateral lumbar spine, proximal femur, whole body, radius and calcaneus. At the hip a number of different measurements are recorded. The femoral neck and trochanter areas have differing trabecular contributions as detailed in chapter 1.1.2. Ward's area is a derived region of maximum trabecular bone content in the proximal femur. For AP scanning of the lumbar spine, precision is less than 1%, and 2-3% at the hip. Accuracy is 5-10%, scan duration is less than 10 mins (20 mins for total body scans) and the radiation dose 0.1-0.4 μ Sv. Recently fan beam (rather than pencil beam) DXA scanner have been developed and are now commercially available. These scanners quantify BMD in the thoracolumbar spine, and the images produced also allow vertebral body morphometric analysis. This development is known as morphometric x-ray absorptiometry (MXA).

이 문을 가지 않는 것을 가지 않는 것이 없다.

and the second

a the second second

DXA is now considered to be the benchmark method of measuring BMD. There are however limitations worthy of discussion. DXA, like DPA and SPA provide planar, two dimensional BMD measurements, which are expressed as q/cm^2 . Such BMD measurements are influenced by skeletal size, and can be underestimated in small individuals (1.88). Although the lumbar spine is considered to be a trabecular rich site, antero-posterior scanning incorporates the posterior elements of the spinal column which is predominantly cortical bone (1.5, 1.7), and contribute 53% of BMD in women, and 44% in men (1.7). This can be minimised by using a lateral projection, but at the expense of poorer precision and accuracy, a smaller region of interest and a radiation dose 5 times greater. Both spinal and hip DXA measurements are influenced by marrow fat content (1.86,1.166,1.167). Increasing amounts of fatty marrow with age underestimates the actual BMD, and overestimates the rate of age related bone loss (1.86). Spinal measurements can

aortic calcification also be falsely elevated by (1.168,1.169,1.170) or degenerative changes secondary to lumbar spondylosis (1.171-1.177), both of which are common with increasing age. There is also evidence that spinal osteoarthritis is related to an increased BMD secondary to changes in bone metabolism, not merely a consequence of osteophytosis (1.174,1.178,1.179), which can have a generalised effect upon BMD (1,177). AP spine Scanning of the hip is less plagued by confounding factors, although smaller regions of interest and rotation during positioning does affect data acquisition (1.180), resulting in poorer precision compared to that of the lumbar spine.

をたくたちになって

のないないないのである。

2

j.

Contraction of the second

states when when

childrenter.com

1.3.5 Ultrasound

Ultrasound was first used to measure properties of bone in the 1960's, but was not widely employed until much more recently. Ultrasound use in the context of bone analysis is based upon interaction of sound waves with bone the tissue. Two ultrasonic properties are altered with transmission through bone: 1) wave velocity, and 2) wave amplitude. Alterations in amplitude are known as attenuation, which is highly dependent upon the frequency of ultrasound used. With low frequency ultrasound, the attenuation is almost linear, whereas with higher frequencies, it is non-linear. For this reason, low frequency ultrasound (0.2 - 0.6MHz) is used . Ultrasound measurements occur at peripheral sites, most commonly the calcaneus, less commonly at the tibia, patella or phalanges. In this thesis, only calcaneal measurements were done. This is a weight bearing, trabecular rich site with flat, parallel surfaces, at which overlying soft tissue is limited even in obese subjects, which has been extensively studied before.

Ultrasound measurements in this thesis were done using the McCue CUBA Clinical machine, the methodology of which is described in chapter 3. Two measurements are produced. Broadband ultrasonic attenuation (BUA) which is measured in dB/MHz, and velocity (VOS) which is measured in m/s. There are slightly different variants of several the velocity measurement. "True velocity" is the velocity of sound through bone tissue only. "Heel velocity" is the velocity of sound through bone and soft tissue. It is this measurement which is used throughout this thesis. Lastly, "time of flight" is the velocity of sound through coupling media, soft tissue and bone. The precision of BUA is 1-3.8%, and that for velocity 0.1-1.2%. Both BUA and velocity measurements are considered to reflect not only bone density, but bone structure (1.181-1.184), although the importance of this property has been questioned (1.185). The major advantages of ultrasound are that it is a low cost, easily portable, radiation free technique, with a rapid scan time of less than 5 minutes. There are however several troublesome features of calcaneal ultrasound worthy of further comment. Precision of BUA remains poorer than other scanning techniques such as DXA. This may be related to the inhomogeneous nature of calcaneal trabecular bone, which in turn can influence the measurement depending upon the region of interest studied (1.186). Recent concern has also been expressed that the present generation scanners are inherently inaccurate (1.187). BUA does not scale linearly with bone size (1.188) (although this would not be a problem unless very large or very small bones were being scanned), or with high BMD values (1.184).

ci Ci

.

and the second second second second second

1.3.6 Quantitative Computed Tomography (QCT)

Quantitative computed tomography was introduced in the mid 1970's, and measures true volumetric BMD, which is therefore expressed as g/cm³. It is also the only method which allows separate quantification of cortical and trabecular BMD. Spinal QCT has been used extensively. Accuracy of single energy spinal QCT is rather poor at 5-15% due to the effects of marrow fat (which increases with age) which falsely reduces BMD, especially in elderly osteoporotic women. Dual energy QCT is now available which improves accuracy to 5% but at the expense of a higher radiation dose and poorer precision (3-5% compared to 1-3% for single energy QCT). Scan time is about 10 mins, but radiation dose is high at 20-200 μ Sv, which limits its use in repeat scanning.

1.1.1

なるないのないとなるでした。

「行う」の書きての読みのでのです。

There are several area of recent development with QCT (1.149,1.150). The use of stacked-slice and spiral CT scans allow a greater volume to be measured with improved precision. A more detailed analysis of spinal trabecular and cortical compartments is possible with respect to the regional effects of ageing and therapy, the relative contributions to bone strength, and fracture discrimination. Such a method of scanning also permits separate analysis of proximal femur trabecular and cortical bone, as well as providing detailed information on geometry. This is not possible with planar methods, such as DXA and DPA, or conventional QCT scanning due to its complex architecture, and rapid three dimensional variation in density and composition. Also being developed is high resolution QCT which allows assessment of trabecular micro-architecture at both axial and appendicular sites. This can be considered as being equivalent to a non-invasive bone biopsy. These are all areas of future study.

Peripheral Quantitative Computed Tomography (pQCT) 1.3.7 OCT applied to peripheral measurement sites was introduced in the mid 1970's (1.189). Radionucleotide sources (usually ¹²⁵I) were used initially, but found to be restrictive. Long scan times introduced movement artifact, and only small diameter sites could be scanned (1.190). To overcome these problems, modern **pQCT** scanners use an x~rav source (1.50,1.75,1.78,1.190). There are approximately 700 pQCT scanners worldwide, mostly in Europe (1.150). Most of these commercially available systems, which are manufactured by Stratec (including the one used for this thesis), perform a single axial slice of 2.5mm thickness at 4% of the ulnar length from the distal radial cortical endplate. The scanning procedure is discussed further in chapter 3. A smaller number of scanners, produced by Densiscan, perform multi-slice, high resolution scans, with improved precision (1.75,1.149). Although the ultra-distal radius is the site most commonly scanned, measurements at the tibia are also possible (1.149). pQCT performs a true volumetric measurement of BMD, results being expressed in g/cm^3 .

「おおおいたはななない」であったなな、パート・コート・ション・ション・ストローンが、たちになっていたのです。 しゅう しゅう しゅうしゅう しゅうしょう

č.

方という言語と

2 6.7.

0.8

In-vitro accuracy of pQCT has recently been determined from the study of 7 cadaveric forearms (1.19). The correlation between pQCT total bone mineral content and ashed weight at the standard measurement site was 0.87, with an accuracy error of 15.5%. The precision of pQCT is discussed further in chapter 3. It varies according to the scanner used, with invivo precision values for trabecular BMD using the multislice, high resolution scanner being about 0.35% (1.75),compared to 1-2% for its single slice counterpart (1.19,1.50, chapter 3).

Rapid scan time of 10-15 minutes, forearm radiation dose of only 3 μ Sv and negligible whole body exposure, easy of use at readily accessible sites, and separate assessment of cortical and trabecular bone make pQCT a potentially attractive technique worthy of further study. and the second state of the second states and the second se

The second s

「たみにはないとないのでのです」という

「おおいいた」を取れていた。「おいた」

and the state of the second state of the secon

1.4 Clinical Indications for BMD Measurement

At the present time, measurement of BMD is the best way of defining fracture risk (1.1,1.88,1.101). There is almost an exponential rise in fracture risk with decreasing BMD. Consequently, small change in BMDof the а osteopenic/osteoporotic individual results in large changes in fracture risk. Data from many studies (1.101,1.191) have shown that the risk of fracture increases between 1.5 and 3.1 times for each standard deviation decrease in BMD. The gradient of fracture risk for each standard deviation decrease in BMD is largely similar between fracture types and between measurement sites. However, the gradient of risk is generally greater when site specific measurement for each fracture type 1.Sperformed - i.e. the optimum measurement site to determine hip fracture risk is the hip (1.192).

たるの言

二日の二日の二日の二日 二日 二日の一日

and the second second

There many indications for BMD are assessment (1.1,1.145,1.146), and these are listed in table 1.2. Previous fragility fracture of hip, spine of forearm should result in BMD assessment. Although hip fracture is a late consequence of osteoporosis, the same is not true of Colles fracture. Modest (5-8%) reductions in hip and spinal BMD have been found in patients with Colles fracture (1.193), which should be regarded at a risk factor for future, more serious fracture. Monitoring the effect of drugs which have an adverse effect on bone (such as corticosteroids) or those which are used to prevent or treat osteoporosis [Hormone Replacement Therapy (HRT), bisphosphonates] is advisable. Radiological evidence of osteopenia is an indication for BMD assessment, because as discussed above, a standard x-ray gives only a crude measure of BMD which is easily misinterpreted. Vertebral deformity warrants BMD assessment, as old traumatic vertebral fractures may not be associated with low BMD. Other findings, risk factors and conditions which are deserving of BMD assessment are shown in table 1.2.

이 가지 않는 것 같아요. 지수가 가지 않는 것 같아요. 상태에 전자가 가지 않는 것 같아.

-

二日 水道町 などろ やんき

いったが、いったとういうにはないがらないないないです。 ないやく ないない マーキー・マント

그 방법에 가지 않는 것 같아요. 그는 것 같아. 가지 않는 것 같아. 가지 않는 것 같아. 가지 않는 것 같아.

1.5 Screening for Osteoporosis

At the present time there is great debate as to whether population based screening for the prevention of osteoporosis should be implemented. This issue has recently been discussed at some length by a worldwide panel of experts, from which a WHO report was produced (1.101). Another important report was published in 1992 (1.194). Although funded by the Department of Health (DOH), this report was not an official doh publication. Based upon these publications and recent papers, the important issue of screening is discussed in the following section.

10.00

a ngaran ang sa sa

ŝ

A STATE AND A S

「新日本」の第一人になる。「新日本」の「「「「

÷,

If screening were found to be feasible, uptake is likely to be in the region of 70-80% (1.101,1.195,1.196). Several methods have been proposed, none of them tested, all shrouded in controversy. A single BMD measurement to assess fracture risk has been proposed. The present day scanners fulfill the necessary to be applied to screening. requirements Asdiscussed above, they are comfortable, accurate, precise machines which scan rapidly and are capable of predicting future fracture. Importantly, they are of low radiation dose (spinal QCT less so). Prospective fracture prediction data on pQCT is not yet available, and although there is good prospective data for ultrasound in hip fracture, it is limited for other fracture types. Based upon the relationship of increasing fracture risk and decreasing BMD, the best site of measurement depends upon which osteoporotic fracture is being predicted. Hip fracture is best predicted by DXA of the hip, while any osteoporotic fracture is predicted equally well by radial SPA or axial DXA. Regarding the timing of a single measurement, there are two schools of thought. One supports measurement during the perimenopausal years as there is profound bone loss following the menopause, and peak bone mass is considered important in determining future fracture risk. The other supports a measurement around the age of 65 years. favoured This option seem more at the present time (1.101,1.197) as bone loss (another important determinant of fracture risk) has had a substantial effect on bone mass, the incidence of hip fracture is still low, and it appears more cost effective to target intervention at this age compared to the age of 50 years.

ほどうえん はいしい たいれい

Sec. Sec. 1

3.

and the second second second second

2

1000

2.2

Another propose method of screening is the use of repeat BMD measurements. Fast bone losers will more quickly reach a critical bone mass at which fracture is likely to occur. A second BMD measurement 5 years following the initial perimenopausal would be considered adequate to assess the rate of bone loss for screening purposes (although an earlier measurement is advisable to assess treatment response), but only improves the estimate of ultimate bone mass by about 50%. Peak bone mass (PBM) is considered to be of greater importance in determining fracture risk up to the age of 70, but thereafter the relative importance of rate of loss and PBM equates. It should be borne in mind however that the precision of repeat measurements worsens in older age, so that a smaller proportion of BMD variance will be related to bone loss. Allied to the concept of assessing rates of bone loss is the use of markers of bone metabolism. There is now a plethora of such tests, some indicating bone formation (bone iso-enzyme alkaline phosphatase, osteocalcin, collagen propeptides markers - P1NP, P1CP), other resorption (hydroxy-proline, pyridinoline, de-oxypyridoline, collagen degradation markers -

1-CTP, 1-NTP). Analysis of a panel of such markers is useful in characterising the rate of bone loss over many years. This in conjunction with a baseline BMD assessment is the third method of screening proposed. The development of more accurate and precise measures of BMD, and more specific markers of bone turnover may increase the predictive power of these proposed methods in the future.

and the second second

(1) 10 (1) 1 (2) 20 (2)

「「「「「「「「「「「「」」」」」

and a state with the second second second

ショー とう スタング 安全の主義 クロー

化化量子 经济公司 化乙烯合物 化乙酸

Several criteria should be met before population based screening is accepted (1.101,1.194). Ideally a randomised controlled study should be performed which shows that screening reduces fracture rates - no such study yet exists. Failing this, several other criteria have to be considered. Firstly, the social burden of the disease should be great enough to warrant screening. As described above, osteoporotic fractures undoubtedly fulfil this criteria, being a major social and financial burden. Secondly, effective therapy must be available. HRT has been shown to reduce fracture incidence, but there was lack of data showing its longterm benefit in the elderly population. More recent studies shows longterm protection against vertebral and forearm fracture by HRT in the elderly (1.198,1.199). Also, conventional prescription of HRT is for a maximum of 10 years, after which bone loss resumes and fracture protection diminishes. The major benefits of HRT are to reduce cardiovascular morbidity and mortality, which weakens the case for targeting HRT treatment based upon BMD. However, BMD assessment may have a role if the result showed an individual to be at high risk, and they were willing to take HRT. BMD results have been shown to influence the prescription (1.200) and uptake (1.201,1.202) of HRT.

Conversely there is no indication to screen for low BMD if the individual is either intending to take HRT irrespective of the result, or would not consider HRT if low BMD was found. Non-HRT treatments are increasing available, and the increasing acceptance of their use should increase the justification for screening. Thirdly, the screening BMD measurement should adequately define future fracture risk. Although this has been shown to be the case, with hip fracture, the overlap in BMD between fracture and non-fracture patients is considered to be too great for screening purposes. Additionally, fracture prediction has been most extensively studied in older postmenopausal females. There is only early, preliminary data to suggest that a measurement during the perimenopausal years can predict future fracture (1.203,1.204). However, the follow up period reported is only 2 years, and the number of hip and vertebral fractures reported were small (1.203) or absent (1.204). Lastly, the result of the screening test would have to encourage longterm HRT use. Although uptake of HRT has been shown to be influenced by a BMD result (1.201,1.202), its longterm use has not.

10.00

がない

- 12

100

in the second

÷

Bearing in mind the above points, the consensus of opinion is that population based screening cannot yet be justified (1,101,1.194). However, gathering of hitherto lacking information, and advances in densitometry and available therapies will require this to be reviewed. Present day policy encourages case finding rather than population screening, whereby patient with suitable risk factors should be offered BMD measurements. There are many conditions which predispose to low bone mass, as detailed in table 1.1. However, it is only those individuals with "strong" risk factors, as detailed in table 1.2, who should routinely undergo BMD assessment, although clinical judgement must also influence this decision. It is noteworthy however, that clinical risk factors do not adequately predict BMD (1.205,1.206,1.207) and perform even worse in predicting fractures (1.207,1.208,1.209). As such they cannot be considered as a replacement for BMD measurement and fracture risk prediction, only a rough guide for further assessment.

i straje i na vradnje i spotovalja u tekno

- 11 - AV

a the first sector of the first

· •• •

riese in

1.6 Rational behind Thesis Work

The central issue regarding pQCT is independent quantification trabecular and cortical bone, the behaviour of which of differs depending upon the prevailing conditions. Ultrasound is the other newly developed technique of assessing BMD which has received alot of attention in recent years. Ultrasound measurements are therefore compared to pQCT in a number of chapters. The preceding sections of chapter 1 lay the foundation of the thesis. Chapter 2 largely describes the methodology of the different densitometry techniques used, concentrating on pQCT. The precision of an osteo-densitometer is of great importance. Chapter 3 describes the measures taken during scanning to ensure the best possible precision for pQCT. More especially, the hypothesis that the difference in scan voxel numbers between repeat scans influences the precision of pQCT is investigated. Chapter 4 examines the effect of dominance upon pQCT BMD measurements. It is important to document this effect and develop a policy regarding scanning, as even small differences could assume greater clinical importance when changes related to treatment. or differences between groups are being sought. Lack of such a scanning policy may increase the variance of measurements and reduce the chance of finding significance. Osteodensitometers usually have pre-loaded normative data which is used to create T and Z scores. In the case of the pQCT scanner used in this thesis, the normative data was based upon a German population. Differences in normal ranges are known to exist between countries, hence it is important to develop a normal range derived from the local population. Chapter 5 describes the creation of a normal female range from the local population. The relationships of important anthropometric

variables to different pQCT BMD measurements within this population are also investigated in this chapter. It is. recognised that BMD at one site may not reflect that of another. Chapter 6 examines the relationship between pQCT BMD measurements, and measurements at other sites using different scanning modalities. It also touches on the potential use of pQCT should a screening program ever be implemented. In chapter 7, cross-sectional age related changes in pQCT BMD's are determined from the population described in chapter 5. Perimenopausal change in BMD is considered to be profound, and preferentially affect trabecular bone. Differences in pQCT BMD between late premenopausal and early postmenopausal women are therefore examined, and compared to axial DXA differences in a cross-sectional study. Also in chapter 7, observed annual rates of change in pQCT BMD are compared to axial DXA changes longitudinal studies in of late premenopausal, and postmenopausal women. Of immense importance in the field of osteo-densitometry is the role of fracture prediction, and no be complete without addressing work would this issue. Prospective, fracture prediction studies require many years. which is outwith the timescale of this thesis. Discriminatory studies have demonstrated almost the same relationship between specific sites of BMD assessment and fracture risk assubsequent prospective studies, and are within a feasible timescale. The discriminatory power of pQCT in hip and vertebral fracture, was therefore compared to modalities known to discriminate and predict both these types of fracture respectively, the results of which are presented in chapter 8. There are now effective treatments for osteoporosis, although not all individuals respond adequately. Another very important role for osteo-densitometry is monitoring the response to

コンシン パート しんがんしけい

and the state of the state

41

and the second second

therapies such HRTand bisphosphonates. Both as these therapies preferentially increase trabecular BMD. and one could hypothesise that the pQCT trabecular BMD measurement may be particularly valuable in monitoring therapeutic response. The potential role of pQCT in monitoring response to HRT and etidronate is therefore examined in chapter 9, and compared to axial DXA measurements. The effect of warfarin on adult bone has received little attention. Chapter 10 describes the effect warfarinisation of longterm upon axial DXA BMD and appendicular pQCT BMD. The relative contribution of the \mathbf{of} inflammatory aspects rheumatoid arthritis (RA), its inevitable effect on mobility and the use of low dose corticosteroid appendicular and axial on BMD remains controversial. Also, peripheral BMD measurements are being cited as potentially useful tools in monitoring disease progression and treatment response in RA. In chapter 11, these issues are addressed using pQCT, with comparison with axial DXA, and ultrasound measurements. The concluding chapter draws together the important findings from the previous investigative chapters, highlights potential areas of clinical use, and proposes potential explanations for the apparent difference in behaviour of axial and appendicular bone.

42

0s life style Ph	steogenesis imperfecta, Turner's syndrome, Klinefelter's syndrome
life style ph	
ŭ	hysical inactivity, immobilization, exercise induced amenorrhoea, cigarette
	onsumption, excess alcohol and alcoholism.
Mutritional Lo	ow body weight, anorexia nervosa, calcium insufficiency.
EX	xcess salt ^(a) , caffeine ^(a) , protein ^(a) , phosphate ^(a)
Indocrine Hy	ypogonadism, early menopause, Cushing's syndrome, thyrotoxicosis, hyperparathyroidism,
hy	ypopituitarism, hyperprolactinaemia, diabetes mellitis ^(a)
Post Operation Hy	ysterectomy (whilst premenopausal), post transplantation, gastrectomy, bowel resection
kheumatological Rh	heumatoid arthritis, ankylosing spondylitis, systemic lupus erythematosis
Jrugs Co	orticosteroid, excess thyroxine replacement therapy, heparin
astrointestinal Cr	rohn's disease, ulcerative colitis, malabsorption
laematological Si	ickle cell haemoglobinaemia, homozygous eta thalassaemia, pernicious anaemia, multiple
Лш	yeloma, marrow accumulation of iron and aluminium
storage disorders Ga	auchers disease, haemochromatosis
)thers Ma	astocytosis

and the second second

11114 1144

· · · ·

- ÷

and the state of the

Table 1.2 Indications for assessment of bone mineral density

والمحي فالمتحاف الملاحين الجاريين

Radiological evidence of osteopenia, or vertebral deformity Previous fragility fracture of forearm, spine or hip Significant loss of height or thoracic kyphosis Monitoring therapy for osteoporosis Strong risk factors Early menopause (<45 years) Prolonged secondary amenorrhoea Primary hypogonadism Corticosteroid therapy Conditions associated with osteoporosis Anorexia nervosa Malabsorption Primary hyperparathyroidism Hyperthyroidism Cushings syndrome Transplantation Chronic renal failure Multiple myeloma Prolonged immobility Osteogenesis imperfecta

<u>aletike</u> – pla

Sec. 1. And Sec.

Figure 1.1. Schematic representation of the bone multicellar unit (BMU). Stage 0 - "Resting" - where the bone is quiescent and lined by flattened cells. Stage 1 - "Activation" - where osteoclast progenitors are atracted "Coupling" - where osteoclastic resorption is completed, and osteoblasts are attracted into the resorption Stage and the resorption cavity is completely filling in. The bone is then quiescent, the surface being covered 5 - "Mineralisation" - where the bone matrix undergoes mineralisation through further osteoblast processing, to the bone surface. Stage 2 - "Resorption" - where osteoclasts remove a quantum of bone. Stage 3 cavity. Stage 4 - "Formation" - where osteoblasts synthesise and secrete the extracellar bone matrix. by flattened cells - Stage 0 - "Resting".





graphic representation of T-scores and Z-scores (right graph) Figure 1.2: Gender related differences in bone mass (left graph), and

sseM enoa

46




CHAPTER 2

METHODOLOGY

2.1 Introduction

where a new property of the second second

Osteo-densitometry has many roles in the management of metabolic bone disease. In the investigation of a new bone scanner, it would be impossible to address such a multitude of important issues in one study. Therefore, for the purposes of this thesis, a number of studies were designed to examine the role of pQCT in areas of clinical importance. The study designs, the numbers of patients recruited, the methods of recruitment and the statistical analyses are discussed in the appropriate chapters. The remainder of this chapter deals largely with the different methods used to quantify BMD, with the emphasis being placed on the description of pQCT.

计自动工具的复数 计可以分离 计标识

2.2 Peripheral Quantitative Computed Tomography (pQCT)

2.2.1 The Osteo-densitometer

pQCT measurements were made of the ultra-distal radius using the Stratec XCT-960 scanner (Stratec Medizintechnik, Birkenfeld, Germany). It produces a narrow fan beam by means of a single energy X-ray tube with heavy filtering (45 kilovolts). The detector unit comprises 6 semiconductor detectors with amplifiers. Movement of the source-detector unit occurs in 3 different directions (longitudinal, transverse and rotational) anđ is controlled by microcontrollers. Patient forearm radiation dose is 0.3 μ Sv for each of the scout and measurement scans (see below). Leakage and scatter results in total body exposure of less than 0.1 μ Sv/hr to patient and operator. The operator radiation exposure can be reduced at least by a factor of 4 by simply standing 1m from the x-ray source, such that external radiation protection is unnecessary for both patient and operator. The total time for scanning and data acquisition is approximately 12 minutes. An individual being scanned using

48

the Stratec XCT-960 is shown in figure 2.1.

2.2.2 Scanning Method

the state of the second second

The patient's forearm length is measured with a millimetre ruler from the ulnar styloid process to the point of the olecranon before the arm is positioned for scanning. This measurement is used to set an elbow support in front of the scanning aperture to ensure the distal forearm and carpus is in the scanning field. The upper limb is then positioned with the elbow in its support, the fingers resting on a finger rest at the back of the scanner with the thumb dangling comfortably below.

Scanning then takes place in two stages. Firstly a coronal scan provides an image of the distal forearm and carpus, and is known as the "scout scan". This allows accurate repositioning of the scanner for the "measurement scan", by positioning a cross-sectional cursor at the distal end of the radius. The measurement scan then occurs at 4% of the previously documented length of forearm, proximal to the cross-sectional cursor (see Fig 2.2). From the measurement scan, 15 tomograms are merged to create an axial slice of 2.5mm thickness, which is depicted on a monitor screen (see figure 2.2). The reconstructed image is comprised of thousands of cubes of uniform dimension but differing density, which are known as "voxels". The density is colour coded on the screen, with high density areas depicted as white/yellow, medium density areas as red, and low density areas as blue/black. This allows visual identification of the radius and ulna within the image. A thresholding algorithm between 0 and 100 is then applied to the image. Bone with a linear attenuation co-efficient (and density) lower than the threshold is defined as trabecular bone, and that above as cortical bone. The default threshold setting is 50, which was not changed in standard scanning. Subsequently the radius is

49

<u>新生产的 人名德尔斯 化合成合金 化合成合合物的</u> 希望的时候,我知道

isolated as the region of interest (ROI), from which several BMD measurements are obtained as described below (chapter 2.2.3). The number of voxels within the ROI is also automatically calculated (typically 900-1000 for a female). As the voxel dimensions and measurement slice width remain constant, the number of voxels in any given ROI is directly proportional to its size.

化二氯化二乙酸医化二乙酯 化二氟基酚 的复数 法法公共 化乙烯酸 网络拉拉人名英格兰人姓氏贝尔 化乙基乙烯酸乙酯

and write assets of the

2.2.3 Bone Mineral Density (BMD) measurements acquired by pQCT

4 different BMD measurements are recorded and shown in Fig 2.2. These are truly volumetric measurements so the resultant BMD values are expressed in g/cm^3 .

1) **TOTAL BMD (Qtot)** is the overall measurement of radial BMD at the measurement site.

2) TRABECULAR BMD (Qtrab). From the cross-sectional image, the outer 55% of the radial area is concentrically removed by the computer software. This leaves a central core of purely trabecular bone which is then quantified giving trabecular BMD. The default setting for the peeling process is 55%, but this can be altered if, in the rare occurrence, cortical bone (white colour) persists after peeling of the outer area.

3) **SUBCORTICAL BMD (Qscort)** is the density of bone in the outer 55% of voxels which are removed. It is mainly cortical bone with a small rim of trabecular bone.

4) CORTICAL BMD (Qcort) is calculated from the outer 55% area which has been removed. The computer software defines bone with a high density, eliminating less dense bone through the further use of thresholding algorithms. This process occurs automatically for any chosen threshold used

(50 by default) in the overall analysis.

ale technic chyferi

The precision of pQCT measurements is discussed in Chapter 3. The non-dominant forearm was scanned in all individuals to excluded any potential effect of mechanical loading on BMD measurements. The effects of dominance is discussed further in Chapter 4. In the rare occurrence of a previous Colles fracture of the non-dominant arm, the dominant arm was scanned, as previous fracture distorts the anatomy at the scan site, and post-fracture radial BMD is known to be reduced (2.1).

A set of the set of

2.3 Dual Energy X-ray Absorptiometry (DXA)

The technique of scanning using DXA has been described in chapter 1.3.4, and in detail elsewhere (1.153,1.154). DXA scanning was mainly performed using a single Norland XR26-Mark II bone densitometer (Norland Corporation, Fort Atkinson, USA). Figure 2.3 shows a woman undergoing DXA scanning of the lumbar spine and proximal femur. The following measurements were made using an antero-posterior projection, with representative scans being shown in figures 2.4-2.6.

1. Lumbar spine [L2-4: (LS)]

E. L. Mary

- Left hip [femoral neck (FN), trochanter (FT) and Ward's area (FW)]
- 3. Whole body [bone mineral density (WB-BMD)].

The in-vivo short-term precision expressed as the coefficient of variation is 0.9%, 2.8%, 1.3%, 4.9% and 1.5% for lumbar spine, femoral neck, trochanter and Ward's area, and whole body BMD measurements respectively (2.2,2.3). To prevent unnecessary repetition throughout the thesis, use of the term "DXA hip" measurements will encompass femoral neck, trochanter and Ward's BMD measurements.

A small proportion of patients with vertebral fractures

51

described in chapter 8, were scanned using a Lunar DPX- α densitometer. The scanning procedure is the same as that for the Norland XR26-Mark II, with the subject lying comfortably on the scanner base which is padded. Data acquisition differs between Norland and Lunar scanners with slight disparity of BMD results. This is dealt with further in chapter 8.

(a) the second state of the second state of

2.4 Ultrasound (CUBA)

Here is a second second second space.

The technique of scanning using ultrasound has been described previously (1.154). Ultrasound was performed using the McCue CUBA Clinical (McCue Ultrasound, Winchester, UK), which is shown in figure 2.7. The non-dominant foot is placed in the foot recess of the ultrasound machine, and two silicone rubber faced transducers applied to either side of the heel. K-Y jelly is applied to eliminate air from between the contact surfaces. Two measurements are recorded: broadband ultrasound attenuation (BUA), and velocity of sound (VOS), results being expressed in dB/MHz and m/s respectively. The in-vivo shorterm precision for these measurements is 3.8% and 1.4% respectively (based upon paired measurements in 16 healthy premenopausal individuals with the foot removed from the machine between measurements). To prevent unnecessary repetition of information throughout the thesis, "calcaneal ultrasound" measurements will refer to both BUA and VOS.

Although pQCT was performed in all studies, DXA and ultrasound measurement were not. The measurements performed in specific studies are detailed in each of the appropriate chapters.

2.5 Statistical Analyses

The statistical analyses used in the various studies differ and are discussed in each of the chapters. The normality of data distribution was verified using normality plots and Shapiro-Wilks statistics. Data which violated normality criteria were compared using non-parametric tests.

52

and the state of the second state of the secon

2.6 Ethics and Consent

and the second second second second

All patients participated in studies approved by the local Ethical Committee, and gave informed, written consent.

a na sa sa kara sa kaka sa ku sa ka sa sa kaka kaka sa ka ka sa karaka ka sa ka sa ka sa ka sa ka sa sa sa sa s

Figure 2.1. The Statec XCT-960 pQCT osteo-densitometer is shown in this figure. A volunteer sits comfortably with her left forearm in the machine aperture during scanning.



Figure 2.2. The top image is a graphic representation of the "scout scan" showing the distal forearm and proximal carpus. The cross-sectional cursor (solid line) is placed at the distal end of the radius and the measuement scan takes place at the "4% measurement site", across the dotted line. The 2nd from top image is a graphic representation of the cross-sectional image of the distal forearm produced by the measurement scan. The radius is identified as the region of interest (within the hatched area) and the voxel number indicates the number of packets of bone in the cross-sectional radial image (voxel number). The total BMD (Qtot) is derived from this image. The bottom two images indicate how the remaining trabecular BMD (Qtrab), subcortical BMD (Qscort) and cortical BMD (Qcort) measurements are derived from the region of interest.



Figure 2.3. A woman undergoing DXA scanning of the lumbar spine (above) and proximal femur (below) using the Norland XR26-Mark II.



Figure 2.4. DXA scan of the lumbar spine using the Norland XR26-Mark II. The region of interest (L2-4) is selected for analysis. The BMD is shown in the box below the spinal image, with associated T and Z scores in the box on the right.



Bone image not for diagnosis

Manual Positioning

	BMD	BMC	LENGTH	AREA
	g/cm ²	g	cm	cm ²
L2	0.782	10.92	3.30	13.96
L3	0.952	15.02	3.45	15.78
L4	0.950	14.85	3.15	15.63
L2 - L4	0.899	40.79	9.90	45.37

STD CVs for L2-L4 BMD: 1.0 BMC: 1.5 See Guide for other CVs. 1.5 x 1.5 mm, 60 mm/s, 12.00 cm Rev. 2.5.2 / 1.3.1 Calib. 19/12/96 Figure 2.5. DXA scan of the left proximal femur using the Norland XR26-Mark II. The regions of interest are shown: femoral neck (FN), femoral trochanter (FT) and femoral Ward's area (FW). The BMD values are shown in the box below, with associated T and Z scores for FN in the box on the right.



Bone image not for diagnosis

	BMD	BMC	LENGTH	AREA
	g/cm ²	g	cm	cm ²
Fem Neck	0.806	3.903	1.50	4.85
Troch	0.618	8.923		14.44
Wards Tri	0.553	0.553	1.00	1.00

STD CVs for Neck BMD: 1.2 BMC: 1.7 See Guide for other CVs. 1.0 x 1.0 mm, 45 mm/s, 9.00 cm Rev. 2.5.2 / 1.3.1 Calib. 19/12/96 Figure 2.6. A whole body DXA scan using the Norland XR26-Mark II. In this instance, the left upper limb has also been selected as an additional region of interest (ROI). The medial border of the selected ROI passes through the glenohumeral joint and excludes all other bony area. The associated BMD values are shown in the boxes below, and to the right of the whole body image.

H



Bone image not for diagnosis

Total BMD (g/cm) :	0.827
Total BMC (g) :	1498
Total Lean Mass(g):	29520
Total Fat Mass (g):	13092
Total Fat % :	29.7
Siri UWE Fat % :	25.4
Brozek UWE Fat % :	24.7
Soft Tissue Fat % :	30.7
% TBMC/FFM :	4.8

DETAILED RESULTS

L

BMD	BMC	AREA	LENGTH	WIDTH	LEAN MASS	FAT MASS
g/cm ²	g	cm ²	cm	cm	g	g
1.370	346.7	253.1			2889	628.4
0.818	462.2	564.9			14365	5082
0.924	183.7	198.7			7459	2213
0.477	131.7	276.1			3125	1548
0.777	557.2	717.4			9141	5834
0.827	1498	1811			29520	13092
0.670	99.97	149.2	63.70	13.00	1162	1622
	BMD g/cm ² 1.370 0.818 0.924 0.477 0.777 0.827 0.670	BMD BMC g/cm² g 1.370 346.7 0.818 462.2 0.924 183.7 0.477 131.7 0.777 557.2 0.827 1498 0.670 99.97	BMD BMC AREA g/cm² g cm² 1.370 346.7 253.1 0.818 462.2 564.9 0.924 183.7 198.7 0.477 131.7 276.1 0.777 557.2 717.4 0.827 1498 1811 0.670 99.97 149.2	BMD BMC AREA LENGTH g/cm ² g cm ² cm 1.370 346.7 253.1 0.818 0.818 462.2 564.9 0.924 0.924 183.7 198.7 0.477 131.7 276.1 0.777 557.2 717.4 0.827 1498 1811 0.670 99.97 149.2 63.70	BMD BMC AREA LENGTH WIDTH g/cm ² g cm ² cm cm 1.370 346.7 253.1	BMD BMC AREA LENGTH WIDTH LEAN MASS g/cm ² g cm ² cm cm g 1.370 346.7 253.1 2889 2889 0.818 462.2 564.9 14365 0.924 183.7 198.7 7459 0.477 131.7 276.1 3125 0.777 557.2 717.4 9141 0.827 1498 1811 29520 0.670 99.97 149.2 63.70 13.00 1162



Figure 2.7. A volunteer undergoing a calcaneal ultrasound measurement using the McCue CUBA Clinical scanner.

and a state of the second state of the

THE PRECISION OF PQCT AND HOW IT IS IMPROVED USING SCAN VOXEL NUMBERS

3.1. Introduction

3.1.1 Precision

Precision, which is the reproducibility of measurement results, is an important concept in osteo-densitometry. The standard value quoted in the literature is the coefficient of variation (CV). The better the precision of a scanner (ie the lower the CV), then the greater its potential for detecting small but significant age, drug or disease related BMD changes. It is also important to evaluate precision in different age groups, as factors in the elderly, such as low BMD and movement artifact, may influence precision.

3.1.2 Voxel Numbers and "Trending".

With repeat measurements of the ultra-distal radius, it is vitally important to scan at the same anatomical site, as the cross-sectional area and proportion of trabecular and cortical bone changes dramatically over short distances (1.17,1.19). Ensuring standardised scanning technique will minimise inter and intra-operator variability, and improve precision.

If the cross-sectional area of a repeat scan in an individual is similar to the original, this should ensure a comparable measurement site, and improved precision. The cross-sectional area of the radius in the ROI is quantified by a "voxel number: (VN)" (Fig 2.2). This could be a useful surrogate measure of area, and therefore a marker of anatomical site in any given individual, assuming there is no real change in the crosssectional area of the radius.

The Stratec 960 has a "trending" function which allows an

61

Shared and the Street steady dealers

additional measurement scan to be performed if the site of the first follow up measurement scan is too far removed from the original. "Trending" utilises the difference in voxel numbers to make small adjustments to the scanner position for a further measurement scan. We were advised by the manufacturer to accept a difference in voxel numbers of 50 or less between repeat measurements. Typically there are approximately 900-1000 voxels per measurement slice in a female subject. Our initial experience suggested that the precision of the scanner using this cut off value was poorer than we would have expected. The effect upon the CV of reducing this value to 30 was therefore examined, then applied to a group of postmenopausal females with vertebral fractures, and another group of postmenopausal females with rheumatoid arthritis.

이 이 가슴 가슴 가지 못 한 것 가지만 영어가 있다.

3.2 Methods

化乙酸乙酯 网络帕雷德尔 化乙二乙烯乙乙烯乙烯

Each individual underwent two consecutive scans at one visit, the arm being removed from the scanner between scans. In addition to BMD data, the VN in each ROI was recorded as was the difference in voxel numbers between scans (vox-diff).

3.3 Statistical Analysis

Continuous variables are expressed as median with ranges. Mean BMD values are also shown as the CV is dependent upon the mean and the SD of measurements. The CV was calculated for each BMD measurement.

3.4 Study Populations

Group A comprised 14 men and 12 women. All women were young healthy volunteers. 13 of the men were volunteers with various cardiovascular diseases participating in a study examining the effects of warfarin on BMD (see chapter 10), whilst the remaining man was young and healthy. None were known to have suffered an osteoporotic fractures. Group A was subdivided according to vox-diff: (Group A-S1: vox-diff \leq 30; Group A-S2:

62

報 한 한 한 1997년 - 1997년 - 1997년 전 1997년 - 1 1997년 - vox-diff > 30). Group B comprised 6 postmenopausal females with at least 1 osteoporotic vertebral fracture documented on x-ray (see chapter 8). Group C comprised 8 postmenopausal females who had suffered from rheumatoid arthritis for at least 1 year (see chapter 11). The vox-diff in groups B and C was \leq 30 for all females.

of the state of th

in a straint s

and a start of the second second

3.5 Results

Demographic, BMD, VN and vox-diff data for each of the groups are shown in table 3.1. The age, gender mix and BMD data for the subgroups of group A were comparable. The VN was greatest in group A-S2, and smallest in group C. By design, the vox-diff is less in subgroup A-S1 than in A-S2. The vox-diff in groups B and C is comparable to group A-S1, although they are all females and older. All BMD values are lower in groups B and C. The CV's for each group for each BMD measurement are shown in table 3.2. By reducing the vox-diff to 30 of less, and comparing group A-S1 with A-S2, the CV is lowered for all BMD measurements. When the vox-diff is comparable (\leq 30), the CV is higher in groups B and C than in group A-S1 for each of the BMD measurements with the exception of Qscort in group B.

3.6 Discussion

the second s

The precision of an osteo-densitometer is of great importance and care should be taken to ensure high reproducibility of BMD measurements. The following technical aspects of pQCT scanning must be standardised to improve reproducibility. The ulnar length should be measured in an identical manner by all operators. This determines the scanner position for the "4% measurement site". The forearm, hand, fingers and thumb must be positioned in the scanner in a similar manner as forearm rotation influences localisation of the region of interest (radius) from the measurement scan. Localisation of the distal end of the radius on the scout scan has to be defined to ensure uniform positioning of the cross-sectional cursor. To aid

63

relocation of the same measurement site in repeat measurements, the scout scan from previous measurements (eg the baseline scan in a longitudinal study) can be recalled. This permits the baseline cross-sectional cursor and measurement site to be referred to during subsequent scanning. Standardising the above procedural elements will minimise operator variation in scanning technique.

- N 1 N 1 1

. . .

Voxel numbers can be used to further improve precision. If the number of voxels between consecutive measurement scans is greater than a predetermined threshold, fine-tuning of the scan site is possible though the "trending" function on the Stratec 960-XCT pQCT. Repositioning then occurs, the scanner moving a tiny distance towards the original scan site. An additional measurement scan (without scoutscan) then takes place at the new site within close proximity of the original scan. This is of vital importance in repeat radial measurements as there are marked changes in the proportion of cortical and trabecular bone over short distances (1.17, 1.18,1.19).

By reducing the difference in voxel number between repeat scans to 30 rather than 50, an improvement in precision was achieved as seen by comparing the CV's of subgroups 1 and 2 of group with CV's of 1.24%, 1.33%, 1.58% and 1.88% for Qtot, Qtrab, Qscort and Qcort respectively in subgroup 1 of group A. This compares well with the precision of other osteo-densitometers in our centre (see chapter 2), and other centres (1.153,1.154). They are also comparable to in-vivo precision values published for the same pQCT scanner at other centres (1.50,3.1).

The precision of pQCT is poorer in the groups with vertebral osteoporotic fractures (group B) and RA (group C), even when the vox-diff was 30 or less. These results can be explained by the fact that the median (and mean) BMD values are lower in these two groups compared to Group A. As the CV is determined

.64

and the second and the state of the second second

by dividing the standard deviation of the difference between measurements by the mean of the measurements, even if the errors between groups are similar, the lower mean value will ensure a higher CV (3.2). Another contributory factor could be movement artifact in the scan images of the older individuals in groups B and C, which falsely elevates the scan voxel number. Older individuals find it more difficult to prevent forearm movement during scanning, particularly when the "trending" function is needed and the scan time increases. Consequently, the repeat measurement site may not be so close to the original as the voxel number suggests, resulting in greater between scan BMD differences. There were however no gross movement artifacts in any of the scans analysed. Also, the median voxel numbers in groups B and C were lower than in subgroup 1 of group A. A difference of 30 is therefore relatively greater in these groups (B & C), resulting in poorer proximity of the follow-up scans. Although the precision in the RA group is poorer, the use of voxel numbers is probably more useful for repeat measurements in this population. The destructive change which occurs at the radio-carpal joint in RA can make it indistinct during the pQCT scout scan. This anatomical landmark is essential for the correct positioning of the scanner for any measurement scan. When this is less well defined, scan voxel numbers are invaluable in repositioning the follow up scan ensuring proximity to the original measurement site.

Whilst the improvements in the precision values between the subgroups of group A are obvious, there are some confounding factors. The mean BMD values of subgroup 1 were slightly greater than the corresponding values in subgroup 2, with the exception of Qscort. This could result in a bias towards better results in subgroup 1. Also the median voxel number of the scans in subgroup 1 was slightly lower than that in subgroup 2. This work could be repeated using a cadaveric radius and ulna

65

والمروان المروان والمحفظ الجاريون ومماليهم الموارك والمرواني والمراجع المراجع

mounted in resin, obtained from a normal individual, an osteoporotic individual, and an individual who suffered from rheumatoid arthritis. This would eliminate in-vivo movement artifact and fatigue factors. The effect of reducing the difference in voxel numbers upon the CV could be studied, and a threshold determined below which there is no further improvement in the CV.

2 - 1 - 2 - 2

그렇는 이 가지 가지 않는 것은 같이 많이 있는 것이 같이 가지 않는 것이 가지 않는 것이 같이 가지 않는 것이 같이 같이 나라.

It became our policy to do a further measurement scan using the "trending" function when the difference in voxel numbers exceeded 30 in repeat measurements. This ensured close anatomical proximity of follow up and baseline measurement sites which is reflected in the improved CV figures. However, there is a large difference in the cross-sectional area, and therefore voxel numbers, of the radius between individuals. It would therefore be better to specify the threshold for a further scan using the trending function, as a percentage of the original scan voxel number rather than an absolute value. A percentage difference of 3% would appear to be appropriate. This figure could be validated or adjusted using the cadaveric radius studies as suggested above.

Million Barrows

1. 124-11 - 11 12 Print - 12 Print - 14 Print

	Group A-S1	Group A-S2	Group B	Group C
Number	20	6	9	8
Age (yrs)	52 (23-73)	53 (26-68)	81 (72-85)	67 (53-73)
F/M ratio	9:11	3:3	6:0	8:0
NIA	990 (769-1471)	1066 (932-1463)	936 (496-1131)	791 (557-1205)
Vox-diff	14 (1-30)	46 (34-66)	11 (1-25)	15 (9-29)
Qtot median (range)	373 (271-578)	370 (311-478)	273 (166-321)	291 (158-381)
[mean BMD]	[382]	[379]	[264]	[289]
Qtrab median (range)	187 (110-277)	179 (141-216)	163 (49-210)	119 (54-271)
[mean BMD]	[190]	[178]	[149]	[136]
Qscort median (range)	506 (356-719)	535 (∉31-70∉)	368 (262-419)	384 (227-502)
[mean BMD]	[526]	[538]	[357]	[389]
Qcort median (range)	534 (441-689)	544 (478-667)	430 (348-514)	478 (321-539)
[mean BMD]	[551]	[550]	[425]	[468]

N9 5 4

1.16

S.

1.1.1.1.1.1.1.1

1.9 11 12 1

A 18 C 19 A.

. .

and the second second second

LAND CHARLES

ff ≤ 30), group 30), and group C		
in group A-S1 (vox-di racture: vox-diff ≤	Group C	
T BMD measurements es with vertebral f : vox-diff < 30)	Group B	
of variation (CV) for pQC B (postmenopausal femal ith rheumatoid arthritis	Group A	\$2
Table 3.2 Co-efficients o A-S2(vox-diff > 30), group (postmenopausal females w		S.T

	54	03	34	32	
	1.6	4.(1.8	2.6	
	1.73	2.96	1.57	5.18	
S2	3.43	1.78	4.05	3,31	
τs	1.24	1.33	1.58	1.88	
	tot)trab	scort	jcort	

CHAPTER 4

THE EFFECT OF DOMINANCE ON RADIAL POCT BMD MEASUREMENTS

化化学学 医外外肌炎 化乙酰

4.1 Introduction

ふうご きんとうよ

The effect of mechanical stimulation and exercise upon BMD at appendicular and axial sites is controversial (4.1). This is due to differing exercise and loading regimes being studied in varying population groups, at different skeletal sites where bone composition varies. With regard to the radius, upper limb exercise and mechanical loading has been shown to have a beneficial effect upon radial bone mass in both cross-sectional longitudinal and studies (4.2, 4.3, 4.4, 4.5)(4.6, 4.7, 4.8, 4.9, 4.10). When the issue of dominance was addressed in some of these studies, bone mass in the dominant forearm was uniformly found to be greater than its non-dominant counterpart (4.2,4.3,4.4,4.9). However these studies examined the effects of regularly playing tennis or specific upper limb exercises, neither of which typically prevail in the normal population. They do however suggest that at the radius, mechanical loading and dominance is of some importance, although the magnitude of this in the general population is poorly defined. The influence of dominance upon radial BMD in the general population was therefore investigated.

4.2 Patient Recruitment and Methods

Dominant and non-dominant forearms were scanned using pQCT on the same day in 31 individuals. 10 individuals (5 males: 5 females) were volunteers participating in a study comparing the precision of DXA scanners in 3 different Scottish centres (2.3). In addition there were 12 men participating in the study discussed in chapter 10, and 9 females participating in the study discussed in chapter 8. 2 of the 31 individuals were left handed.

4.3 Statistical Analyses

ショー・ビー・ショングリート しいしい

Continuous variables are expressed as the mean and standard deviation. Comparison between the dominant and non-dominant BMD values was performed by the paired t-test, and differences expressed as percentages.

4.4 Results

The BMD measurements of dominant and non-dominant forearms with percentage differences are shown in table 4.1. Considering all individuals as a single group, there was a trend for all BMD measurements to be greater in the dominant radius, although this failed to reach significance: percentage differences of 1.6%, 1.1%, 1.2% and 0.6% for Qtot, Qtrab, Qscort and Qcort respectively. BMD values of the dominant forearm were slightly higher than the non-dominant forearm with the exception of Qtrab in the male subgroup, and Qscort and Qcort in the female subgroup. None of the differences were statistically significant. BMD values in the male subgroup were higher than the corresponding values in the female subgroup, although they were not age matched.

4.5 Discussion

These data suggest that there is no statistically significant effect of dominance upon any radial BMD measurement in either males of females, although there was a trend for BMD to be slightly higher in the dominant arm. Although not statistically significant, the differences may be clinically important. The differences were largely in the region of 1-2%, which is similar to the precision error for the different BMD measurements. Introducing a further 1-2% variation in BMD measurements would limit the ability of pQCT to differentiate osteopenic/osteoporotic populations from normals, to detect rates of change and to monitor response to drugs.

A previous study examining the effect of dominance upon radial BMD using pQCT in normal sedentary individuals found cortical and total, but not trabecular BMD to be significantly greater in the dominant forearm, the differences being 11%, 10% and 6% respectively (4.11). The disparity with our findings is difficult to explain as the study populations seemed comparable in their level of physical activity (no elite athletes), and the same pQCT scanner was used.

Small, and statistically insignificant differences between DXA BMD of opposing proximal femora have also been documented (4.12,4.13). Infrequently however, considerable individual variations have also been noted (4.12,4.13), and thought to be due largely to local factors such as previous fracture or hip osteoarthritis (4.12). 7-8% variation in ultrasound attenuation (BUA) measurements between opposing os calces have previously been found which was unrelated to stated handedness. The conclusion of the authors was that the same site should always be measured, particularly in longitudinal studies (4.14).

In summary, dominance and local mechanical stimulation do appear to have a mild influence on radial BMD, which may be of clinical importance without achieving statistical significance. Although only small and insignificant differences were found between radii in this work, the numbers studies were relatively small. Greater differences have been recorded previously in the general population (4.11), and in those undertaking specific upper limb exercises (4.2,4.3,4.4,4.9). Bearing in mind that small percentage differences would even be clinically important, it is important to remove the influence of dominance as a confounding factor in pQCT BMD measurements. Consequently it was our policy to always scan the non-dominant forearm, unless an individual had suffered a previous Colles fracture, when the contralateral forearm was scanned. In longitudinal

studies the same forearm was always scanned.

1 - A - A - A

- - - - -

and the second of the second second second second

an an an ann an an Allan an Allan an Ann an Allanda an ann an Allanda.

Table 4.1. pQCT BMD measurements for the dominant (DOM) and non-dominant (NON-DOM) forearms in the whole study population (whole pop), and the male and female subgroups.

이 가지 않는 것이다. 이 가격에 가장이 많은 것이지 않는 것이 다 나 물건을 하는 것이 가지 않는 것이다. 이 가지 않는 것이 가지 않는 것이다. 나는 것이 나는 것이 가지 않는 것이다.

化氯化乙酸 建物化物理论 网络植物 化二乙酸盐 法公共公司 网络根花花

	DOM	NON-DOM	p	€ diff
Whole pop.	(n=31)			
Qtot	363.4 (85.1)	357.6 (77.4)	0.23	1,6
Qtrab	181.6 (53.4)	179.6 (54.0)	0.60	1.1
Qscort	500.2 (113.2)) 494.0 (96.7)	0.41	1.2
Qcort	528,6 (82.7)	525.4 (79.9)	0.67	0.6
Males (n=17	')			
Qtot	409.5 (76.9)	400.5 (67.5)	0.23	2.3
Qtrab	210.6 (43.9)	212.7 (39.1)	0.68	-1.0
Qscort	556.6 (96.7)	541.1 (82.6)	0.16	2.7
Qcort	558.7 (65.0)	545.9 (65.7)	0.19	2.3
Females (n=	:14)			
Qtot	307.4 (57.4)	305.6 (54.0)	0.77	0.6
Qtrab	146.4 (42.3)	139.4 (40.9)	0,22	4.8
Qscort	431.7 (94.1)	436.9 (82.3)	0.61	-1.2
Qcort	491.9 (89.1)	500,4 (90.5)	0.49	-1.7

Values are mean (SD). p value is for the paired t-test.

% difference is that between the mean dominant and non-dominant BMD.

73

and the second second

the second second second second second

n ann an Arailean

CREATION OF A NORMAL RANGE AND DETERMINATION OF IMPORTANT ANTHROPOMETRIC FACTORS ON RADIAL pQCT BMD MEASUREMENTS IN FEMALES AGED 18-90.

영화화동안 이 방법 이 가지 않는 것 같아요.

化马马克曼曼马曼 法书记的证书 充满 医前外裂的 法定的现在分词 法可辩论 的复数分子的 的复数

5.1 Introduction

The Stratec XCT-960 pQCT bone scanner was developed in Germany. The manufacturers did provide a normal range for trabecular and BMD values, although this was based on a German total population aged 20-79 scanned using a precursor to the Stratec XCT-960 - the Stratec SCT-900 - which used a ¹²⁵I rather than X-There had been concern that the normal range ray source. provided was inaccurate, a worry which has recently been confirmed in a German population (1.50). Therefore, it was felt prudent to investigate whether the manufacturers reference range was representative of the population of Grampian region, Scotland. The best method of determining a normal range from a reference population would be to randomly select females from a population or GP register. Financial and time restraints meant that this was not possible, but all normal females who had been scanned with pQCT were analysed. Since age, height, weight and menopausal status are known to be important factors in determining BMD (1.50,5.1,5.2,5.3), their influence on pQCT radial BMD measurements was examined in this population.

5.2 Study Population

332 normal females who had been scanned with pQCT were studied. All females completed a health questionnaire. None were known to suffer from any condition or were taking medication known to influence bone mass or metabolism and none had previously undergone hysterectomy. They were recruited in the following way:

1. 61 volunteers who worked in the Osteoporosis Research Unit

74

C. Addie C. Cale

<u>a parte de la construcción de</u>

and the hospital at which the Unit was based.

rage internet and setting of the setting

- 2. 197 females who attended an osteoporosis screening programme which has been described previously (1.196). These females were aged 45-55 years old, lived in the Grampian region of Scotland and were randomly selected from a community register. Primary screening involved DXA scanning of the lumbar spine and hip. pQCT was performed within 2 months at an additional visit .
- 3. 31 volunteers selected from a previously studied random population who took part in an epidemiological study (European Vertebral Osteoporosis Study EVOS) (5.4). They were known to fulfill criteria for normality 2 years previously, and agreed to return for further assessment. 30 again fulfilled normality criteria and were included in this study population, but one had suffered a vertebral fracture, and was excluded. They formed part of the control group for the studies discussed in chapters 8 and 11.
- 4. 45 volunteers who were recruited through advertising as normal controls for a study examining the power of different scanning modalities (DXA, CUBA and pQCT) to discriminate patients with previous hip fracture from a control population. This study is discussed further in chapter 8. One female self reported a history of rheumatoid arthritis and was therefore excluded, leaving 44 females to be included in this normal population

5.3 Statistical Analyses

BMD was stratified by age into consecutive decades, and by menopausal status (premenopausal: regular menses still occurring; postmenopausal: no menses for at least 12 months). Age specified mean and SD BMD values were compared to those provided by the manufacturer, and also a recently published German normal range. The effect of the independent variables

75

was not the second state of the second state second

a she a mana a na makaza da maya a 🔊

age, weight, height and menopausal status was examined by stepwise multiple regression analyses in the whole population. Analyses of premenopausal and postmenopausal subgroups were undertaken, with years since the menopause being substituted for menopausal status in the postmenopausal group.

"你们都不觉你,还有什么可能,这个你就是什么?"李浩说道:"你们还不是你的,你是你能是是你不知道你,你

5.4 Results

119 of the population were postmenopausal. Total and trabecular BMD data for our normal population and that provided by the manufacturer are shown in table 5.1. Cortical and subcortical BMD data for our population are shown in table 5.2 (there were no corresponding manufacturers values). Our normal range differed from that of the manufacturer and also that recently published for a German population (1,50). There are data on only 6 females in the 30-39 age band, which unfortunately cannot be considered a truly representative sample of the normal population. This could explain why Qtot, Qtrab and Qscort BMD values are lower than expected in this age band. Mean Qtot and Qtrab BMD values were generally higher in our population, compared to the manufacturers and recently reported German population (1.50). Compared to the manufacturer's figures, the difference tended to be greater for Qtot (+9.5% -+31.1%) than for Qtrab (-1.2% - +10.6%) excluding those aged 30-39).

The age related changes in Qtot, Qtrab, Qscort and Qcort are shown in figures 5.1-5.4 respectively. Multiple regression equations examining the influence of the independent factors age, height, weight and menopausal status (years postmenopause in postmenopausal females) upon BMD measurements in the whole population, and premenopausal and postmenopausal subgroups are also shown in the corresponding figures.

There was an age related decrease in all BMD values. Examining

76

and share and a sheet of the state of the state

the whole population by decades, this was not apparent until the 6th decade for Qtot, Qscort and Qcort and the 7th decade for Qtrab. Age related changes in BMD are discussed further in chapter 7. Important anthropometric determinants in the whole population were as follows: weight which was positively correlated with Qtot, Qtrab and Qcort; age which was negatively correlated to all measurements; height which was negatively correlated to Qtot and Qtrab; menopausal status which was negatively correlated to Qtot and Qscort.

1

For premenopausal females, weight was positively correlated to Qtrab, and height negatively correlated to Qtot and Qtrab. The amount of variance explained in the premenopausal group was small with only 5% of the variance in Qtot, and 8% of Qtrab being explained. The amount of variance in Qscort and Qcort explained by these independent variables (age, height and weight) was so low that no equation was formed.

For postmenopausal females, weight was positively correlated to Qtot, Qscort and Qcort, whilst age was negatively correlated to all BMD measurements. Years since the menopause was not an important determinant for any pQCT BMD measurement. The degree of variance in BMD explained in the postmenopausal group was much greater than in the premenopausal group: 35% for Qcort, 34% for Qscort, 33% for Qtot, but only 8% for Qtrab.

5.5 Discussion

These results highlight the importance of obtaining a locally derived normal reference range. Values supplied by the manufacturer were consistently lower across the decades than those of the local Grampian population. There were also differences in BMD values between the female population of Grampian, and the German population reported recently by Butz et al (1.50). These differences are probably real, but the

77

an a character a character a state in the state of the st

following factors could also be important. Firstly, the recruitment methods for the study populations were different. The entire German population reported by Butz et al (1.50) was randomly selected, whilst about 100 of the 332 females in our population were volunteers, the remainder being randomly selected from the population. Secondly, our pQCT scanner (Stratec XCT-960) was the upgraded model from that used to scan the German population (Stratec XCT-900). Differences in performance of the scanners are not known, but are unlikely to result in the large BMD differences reported here. The finding of population differences does however support the findings of a recent study, which reported that normative female pQCT BMD data vary between a number of different European centres (5,5), Mean trabecular BMD of females aged 60 years at the different centres varied between $101-117 \text{ mg/cm}^3$, although this was statistically insignificant. Similar findings have also been shown for normative DXA spinal (5.6,5.7) and hip (5.7,5.8) Differences between a locally derived normal range and data. that provided by the manufacturer have also been found for axial DXA (5.7,5.9,5.10) and pQCT measurements (1.50). Such difference important are as they can lead to the misclassification of patients as osteoporotic or not (5.7,5.10). If must be concluded that, if possible, normative data should be derived from the local population.

그는 것이 이 방법에 가지 않는 것이 것 같아요. 이 가지 않는 것이 가지 않는 것이 가지 않는 것이 많이 가지 않는 것이 없다.

For the purposes of this thesis, the population studied in this chapter is referred to as a "normal" population. However there are several limitations of this population. Firstly, approximately 30% of females were not randomly selected. Secondly, there is a paucity of females aged 30-39 years, with BMD values in this age group generally being lower than their counterparts aged 18-29 years and 40-49 years. This suggest that the females in this age band are not representative of the population. Unfortunately, it is this age group which is used

78

an an an an

يستري المركزة المترققان

most often to create T-scores. Hence, during the remainder of the thesis, T and Z scores are quoted for DXA femoral neck and lumbar spine measurements, but not for pQCT measurements.

化分配 医内侧上颌 化氨基 化聚合物 化化氨酸化物 化可加加 化化化合物 化分子 化分子子 化分子子 医子宫下颌

18.90 0.00

- N 1 4 1

Examining the BMD values across the decades, pQCT values do not appear to alter greatly until the 6th decade for Qtot, Qscort and Qcort, and the 7th decade for Qtrab. This is a similar pattern to that observed previously for pQCT measurements (1.50,1.78). Bearing in mind the limitations of trying to determine peak bone mass with a small number of females aged 30-39 years, maximum mean BMD is achieved in the age group 40-49 years for Qtot, 18-29 years for Qtrab and Qscort, and 30-39 years for Qcort (40-49 years if the 30-39 age group is excluded). It is surprising that there does not appear to be a dramatic reduction in Qtrab during the 6th decade, bearing in mind that the accelerated phase of bone loss during the 5-10 years following the menopause preferentially affects trabecular bone. Rates of radial bone loss are examined further in chapter 7.

Weight and height are generally positively related to axial hip and spine BMD, whilst age and postmenopausal status are negatively related (1.51,4.2,5.1,5.2,5.11,5.12). The results presented here suggest that in premenopausal females, the above independent variables explain only a small amount of the variance in the various pQCT BMD values, although other local and environmental factors were been examined. This is in keeping with the previous finding that peak bone mass at the radius (1.58), and hip and spine (1.59) is largely genetically determined, with up to 80% of variance being explained by genetic and familial factors.

The variance in pQCT BMD explained by age, weight, height and menopausal status for the normative data presented here, was

79

かくいわい ひゃいこう アイレイトマイエント たいか たいしゃちょう 道い

the postmenopausal, compared much greater in to the premenopausal subgroup. Age was an important determinant of all BMD measurements, being negatively related in the whole population and the postmenopausal subgroup. It was surprising that years since the menopause was not related to any postmenopausal females, particularly measurement in as trabecular bone is preferentially affected during the earlier postmenopausal years, and is independently quantified by pQCT. However, postmenopausal status was negatively related to total and subcortical BMD measurements. Weight was also positively related to BMD measurements in the whole population, but seemed to have a greater influence upon BMD of postmenopausal females. Conversely, height was negatively related to total and trabecular BMD in the whole population, and more important in premenopausal females. This relationship of height with radial BMD is an unexpected and unusual finding which is difficult to explain, but has been noted before for trabecular BMD (1.64). It is possible that taller women have thinner radii.

コート・アリア ひょうえいしょう かんえん かくとうなな しょうしゃ ひゃくせい アアワオト

As a second state of a second state.

10 C 10 Mar

Considering the two menopausal groups together, these results are similar to previously published studies. Age, has been shown tο bе negatively related (1.50,1.51,1.63,1.68,1.69,1.70,1.78,1.79,5.3,5.13), and weight unrelated (1.50, 1.64, 5.13) or weakly positively related (1.51) to radial BMD. Height has been variably shown to be unrelated (5.13), weakly positively (1.50,1.51,5.14) or weakly negatively (1, 64)related. Postmenopausal status (1.51)and years postmenopause (5.13) have been found to be negatively related to radial BMD. The effect of the menopause upon radial pQCT BMD measurements is discussed further in chapter 7. Disparity with the some of the results of others could be explained by differing method of analysing the relationship between anthropometric and BMD variables, as some have used linear regression, others multiple regression and others a combination

80

<u>ala na districtione de la comp</u>

li ke 🗤 naak

of both. Additionally, although pQCT was used in some of the work quoted above (1.50,1.64), SPA was used by others (1.51,5.13). Accordingly, the scan site and therefore proportion of trabecular bone differs, as do the reported BMD measurements.

그 같아? 영양 것 이 제 영양이 이 지수가 이 공장한 바이지?

1.94

ne ne verse state de la constante de la consta

Table 5.1. pQCT BMD values for our normal population compared with the manufacturers reference data (Qtot

and Qtrab only).

	Age (yrs)	our	Normal da	Ita	Reference	<u>ă</u> ata	Differenc	Ð
		q	Mean	ßD	Mean	SD	mg/cm³	% age
Qtot	18-29	33	387.4	54.6	267.0	40.0	+120.4	+31.1
	30-39	9	348.4	57.7	269.2	40.0	+79.2	+22.7
	40-49	173	392.2	48.6	270.8	40.0	+121.4	+31.0
	50-59	55	370.0	70.0	268.0	40.0	+102.0	+27.6
	60-69	22	331,0	63.7	264.3	40.0	+66.7	+20.2
	70-79	29	286.8	59.4	259.6	40.0	+27.2	t.9.1
	80-90	74	257.6	54.2	I	1	ı	I
Qtrab	18-29	33	191.2	27.9	170.9	35.5	+20.3	+10.6
	30-39	9	137.3	25.0	174.1	35.5	-36.8	-26.8
	40-49	173	186.7	36.5	176.1	35.5	+10.6	+5.7
	50-59	5 10	180.4	45.5	168.2	35.5	+12.2	+6.8
	60-69	22	165.7	41.7	159.3	35.5	+6.4	9.6+
	70-79	29	154.4	51.2	156.3	35.5	9. [-	-1.2
	80-90	14	130.6	48.0	t	I	1	ı

1.1.1
Table 5.2. Subcortical (Qscort) and cortical (Qcort) BMD values for our normal population. There were no corresponding reference data from the manufacturer.

	Our	Normal di	ata		Age (yrs)	Our	Normal de	ບ ມ
	1	mean	SD			r d	mean	SD
18-29	33	542.3	86.9	Qcort	18-29	ei Ei	562.3	78.5
30-39	9	518.5	91.7		30-39	6	588.7	61.0
40-49	173	538.2	57.9		40-49	173	563.9	54.3
50-59	55	508.1	86.0		50-59	55	544.0	65.1
60-69	22	454.8	88.3		60-69	22	501.2	87.1
70-79	29	392.1	82.6		70-75	29	447.5	76.0
80-90	1 4	359.4	74.8		80-90	14	427.7	79.1

%age is the percentage difference between our normal population and the reference data. n=number of measurements in each age range.

8

- 2° -

Fig 5.1: Age dependent distribution of Qtot BMD for normal females



Fig 5.2: Age dependent distribution of Qtrab for normal females

stepwise multiple regression equations











CHAPTER 6

COMPARISON OF RADIAL PQCT BMD MEASUREMENTS WITH MEASUREMENTS AT OTHER SKELETAL SITES: IMPLICATIONS FOR THE USE OF PQCT IN SCREENING FOR LOW AXIAL BMD.

6.1 Introduction

化分离 计可继续 化二氟化化 无度的 法法法的 医外侧室 化结构 化乙基乙基乙基乙基乙基乙基乙基乙基乙基乙基乙基乙基

Bone density measurements are now possible at a variety of skeletal sites using different techniques. At the present time DXA is considered the "benchmark" method of assessment. Therefore, in the development of pQCT, it would be important to document the relationship between DXA and pQCT BMD measurements. Similarly, it would be useful to know the relationship between pQCT and other emerging scanning techniques such as ultrasound. This is particularly important regional variation in bone mass status exists within as individuals (6.1,6.2,6.3).

Osteo-densitometry has some well defined roles in the management of bone disease (1.145,1.146). However, the issue of screening and prevention for the general population remains contentious (1.101,1.194). Screening for osteoporosis has been discussed in more detail in the introduction (chapter 1.5). It has been suggested that DXA may be of value in screening perimenopausal women, targeting those considered to be "at risk" for life style and therapeutic intervention. Selection of individuals in the lowest quartile (1.88) or quintile (1.101) of aged-matched DXA hip and lumbar spine measurements is one method proposed for identifying individuals for intervention, while to target prevention in another is those with osteoporosis (T-score of \leq -2.5), or osteopenia (T-score of \leq -1.0 but > -2.5) (6.4). An ideal bone density screening tool will accurately predict subjects at risk of future osteoporotic fracture. As discussed in the introduction (chapter 1.4 & 1.5),

88

<u>e Produced Martines de la construcción de la construcción de la construcción de la construcción de la construc</u>

BMD assessment is the best method of defining fracture risk (1.101,1.194). The best predictive power is achieved by measuring BMD at the site of future fracture, ie at hip and spine (1.88). pQCT is probably cheaper and more portable than DXA, so could be used either to predict fractures or to select subjects for measurement with the more expensive DXA technique. The predictive capacity of pQCT can only be determined by long-term prospective studies, as already done for radial SPA measurements in spine (6.5) and hip (1.192) fractures. Pending publication of such studies with pQCT, it could be used in an attempt to select individuals with a high risk of low hip and spine BMD.

化硫酸盐医硫酸盐酸盐 化氨基乙基氨基乙基 网络小麦属

The relationships between BMD measurements using different techniques at various sites were therefore examined by comparing pQCT BMD measurements with DXA (hip and spine) and calcaneal ultrasound measurements in a perimenopausal female population, and whole body DXA measurements in a postmenopausal female population. The potential for pQCT to be used as a community based screening tool preliminary to axial DXA measurement was examined in the perimenopausal population that is, could it pre-select a group from the community who could be offered an axial DXA BMD measurement.

6.2 Study Populations and Bone Mass Measurements

The relationship of pQCT BMD measurements and those at other skeletal sites was examined in two groups:

Group 1. The relationships between thefour pQCT BMD measurements were examined in 216 perimenopausal females who attended an osteoporosis screening programme. The method of recruitment has been described previously (Chapter 5.5.2-point 2). The relationships between pQCT and DXA hip and lumbar

89

ga saki di Maraka shaka ka ku

the second s

spine BMD measurements, and calcaneal ultrasound measurements were also studied in this population. DXA measurements were performed on the initial screening visit, whilst pQCT and ultrasound measurements were performed on the same day at an additional visit within 2 months.

1997年,1997年,1997年,1997年,1997年,1997年初,1997年,1997年,1997年,1997年,1997年,1997年,1997年年,1997年年,1997年,1997年,1997年,1997年,

Comparison of pQCT measurements with DXA whole body Group 2. BMD (WB-BMD) was made in 45 postmenopausal females who were acting as controls in a study examining the power of different scanning modalities (DXA, CUBA and pQCT) in discriminating patients with hip fracture from a control population. This study is discussed further in chapter 8. Women within this group who had undergone hysterectomy had done so more than 2 years after cessation of their menses, and were considered to be postmenopausal. Further analysis of the whole body scan identified the non-dominant arm (ipsilateral to that scanned using pQCT) as an additional region of interest (ROI). The medial border of this ROI passed through the gleno-humeral joint and excluded all other bony areas. This is depicted in figure 2.5. The BMD of the whole arm was then derived (WB-ARM). All measurements were all done on the same day.

6.3 Statistical Analyses

Results for continuous variables are expressed as mean with standard deviation when normally distributed, median with range otherwise. Maximum and minimum values are shown for all bone density variables. The relationships between the various BMD measurements were examined by linear regression and Pearson correlation co-efficients. In the perimenopausal population (group 1), quartiles of the DXA, CUBA and pQCT measurements

90

<u>and and the second s</u>

calculated with comparison of the proportion were of guartiles individuals in the lowest of the different measurements. In view of the recently agreed WHO criteria for the diagnosis of osteoporosis and osteopenia based on bone density measurements (6.4), the proportional cut off value for each pQCT measurement which would be required to detect all women with osteopenia (T-score of \leq -1.0) at either the hip (femoral neck) or spine site was determined, and compared to ultrasound measurements.

- 2 X L

6.4 Results

6.4.1 Demographic and Bone Mass Data

人名法德尔斯 建苯基苯基酚 网络白垩色白垩 的复数形式 化苯乙基 建铁石 建制造工具 建苯乙基 网络人名

Details of age, height and weight for all individuals (groups 1 & 2) are shown in Table 6.1. In group 1, 101 (46.8%) of the study population were premenopausal, 46 (21.3%) had previously undergone hysterectomy and 69 (31.9%) were postmenopausal. All females in groups 2 were postmenopausal. Bone mass data for groups 1 and 2 are shown in tables 6.2 and 6.3 respectively. Due to scanner malfunction, it was not possible to record BUA and VOS results in 5 and 7 subjects respectively in group 1, and it proved impossible to isolate the arm as a ROI from the whole body scan in 1 female in group 2. Only the DXA hip and spine measurements have a locally acquired normal range based on 300 women known not to have osteoporosis. Locally derived T-scores are given for the group 1 study population compared to this normal range.

6.4.2 Intra-Technique BMD Correlations for pQCT

The relationships between the different pQCT measurements are shown in figures 6.1-6.3. Qtot and Qscort correlated best (r=0.917). Qtot correlated less strongly with Qtrab and Qcort (r=0.687, r=0.551 respectively). Qscort was highly correlated with Qcort (r=0.739) but only moderately with Qtrab (r=0.457). Qtrab and Qcort were very poorly and negatively correlated (r=-

91

and the stand of the second of states as

han die Stradie Gegeneren zugen

0.152).

计分词存储 医无关节的 网络马尔斯马尔斯马尔斯马尔斯

6.4.3 Correlations of PQCT BMD Measurements with Bone Mass Measurements at Other Skeletal Sites

승규는 이번 것이 가격하지 않는지 것이 가지 않는 것이 나무나 있는지 가지 않는 것이 있다. 승규는 것이 것

The relationships of Qtot with hip, spine, arm and whole body BMD measured by DXA, and calcaneal ultrasound BUA and VOS measurements are shown in figures 6.4-6.7. The correlation with hip measurements were all of a similar magnitude (FN: r=0.459, FT: r=0.443, FW: r=0.441), and slightly lower with LS (r=0.389). Correlation were comparable with WB-ARM (r=0.456), and slightly higher with WB-BMD (r=0.545). The relationship with calcaneal ultrasound measurements were generally poor (BUA: r=0.272; VOS: r=0.159).

The relationships of Qtrab with the above measurements are shown in figures 6.8-6.11. The relationships were similar to those found for Qtot, with slightly higher correlation with LS (r=0.41), FT (r=0.462), FT (r=0.525), WB-BMD (r=0.573) and BUA (r=0.306), and slightly lower correlation with FW (r=0.438), WBA-ARM (r=0.435) and VOS (r=0.097).

The relationships of Qscort with the above measurements are shown in figures 6.12-6.15. The correlations with LS (r=0.345), hip (r=0.406, 0.363, 0.403 for FN, FT and FW respectively), WB-BMD (r=0.416), WB-ARM (r=0.371) and BUA (r=0.242) were slightly poorer, and VOS (r=0.165) slightly greater than those found for Qtot.

The correlations of Qcort with other measurements (figures 6.16-6.19) were found to be extremely poor: (r: 0.047-0.207), and was best with WB-ARM.

6.4.4 Comparison of the Proportion of Females in the Lowest Quartiles of the Different Measurements for the

92

and the second second

Perimenopausal Population (Group 1)

しきむえ えいたかい せかく ひとあれ ちゃわくろ

and the second second

나는 것 같은 것 같아요. 영화는

Subsequences and subsequences

All women were placed in a quartile for each of the DXA (hip and spine), pQCT and ultrasound measurements. The percentage of females falling into the lowest quartiles (QU_4) of pQCT and DXA measurements also in the lowest quartile of other mutually exclusive hip and spine measurements are shown in Figure 6.20. Around 50% of women in QU_4 of LS, FT and FW and around 25% in QU_4 of FN were also in QU_4 of Qtot, Qtrab and Qscort. The corresponding figures for Qcort were around 30%. However, only about 60% of women in QU_4 of the hip measurements were also in OU_{4} of LS, and about 70% when comparing intra-hip measurements. Figure 6.21 shows a similar analysis comparing in QU_4 of the pQCT and ultrasound measurements. women Generally, less than 50% of women in QU_4 of all рQCT measurements were also in QU₄ of BUA, with the corresponding figures for VOS being lower.

6.4.5 Proportional Cut-Off Figures for pQCT Measurements in Detecting Low Axial BMD in the Perimenopausal Population :Comparison with Ultrasound Measurements (Group 1).

In group 1, 97 (44.9%) women had a T-score \leq -1.0 at LS, and 108 (50%) at FN. The proportional cut off values to detect all women with T scores \leq -1.0 at LS and FN and the percentage of women below this value for each of the other bone density measurements are shown in Table 6.4. The proportional cut-off values for all pQCT measurements were such that almost the whole population would have to be scanned to detect all women with hip and spine osteopenia. However, the findings were similar for ultrasound measurements. Also, to detect all women with spinal osteopenia based upon a FN measurement, 96.3% of the population would have to be scanned, and 88.4% to detect hip osteopenia based upon a lumbar spine measurement.

93

Server and the server

Real And Street Street Street

6.5 Discussion

The second second second second second

Intra-technique correlations of pQCT BMD measurements were variable. The highest correlation was found between total and subcortical BMD values, which can be explained by subcortical measurements incorporating the cortical shell and a small rim bone, thus both cortical and trabecular of trabecular components are represented in this measurement, as they are in the total BMD measurement. The correlations between other pQCT measurements were only moderately high, with BMD the relationship between trabecular and cortical measurements being negative. This suggests that radial bone at this site is inhomogeneous, as previously suggested (1.153). Also, rapid and dramatic changes in the proportion of cortical and trabecular bone occurs within the distal 3cm of the radius (1.17). The measurement site for ultra-distal pQCT measurements falls within this area so small differences in the relative scan position between individuals will result in a large scatter of results, and poorer correlation between BMD values. These correlation coefficients are similar to previously reported results for the same scanner (3.1, 6.6). It is worth noting that relationship the between trabecular anđ cortical BMD measurements at the ultra-distal radial site is very poor, a feature which has also been noted previously (3.1). This is probably related to the thinness of the measurement slice introducing sampling errors (3.1), an age related diminution in both cortical bone thickness and density (6.6), and an age related increase in the proportion of trabecular bone at the measurement site (6.6). However, the age range of the population studied (group 1) to determine the correlations of pQCT BMD measurements was narrow (45-55 years), suggesting that a dramatic change in the proportion of trabecular and cortical bone over small distances at the ultra-distal site, and thinness of the measurement slice were the main reasons for the lack of correlation between pQCT cortical and trabecular BMD.

しかしゃ かんばん かっかん たけし きょうかやく けいしん しかりしない

化氨基甲基氨基甲基甲基氨基甲基甲基

94

e na su se contra e e entre e entre de la contra de la contr

There was also only moderate correlation between pQCT BMD measurements and the total BMD of the ipsilateral arm, as identified from the DXA whole body scan. As the radius, ulna and humerus are long bones consisting predominantly of cortical bone, it is surprising that the correlation between the pQCT cortical BMD and DXA arm BMD was not higher, whilst the best correlation was found with the pQCT total BMD measurement. Once again though, this is likely to be related in part to the size of the measurement sample, relative to the size of the upper limb.

그리는 것 같은 것 같은 것 같은 물건적인 것같은 것을 하는 것이?

The correlations between total, trabecular and subcortical pQCT radial BMD measurements and DXA BMD measurements of the lumbar spine, proximal femur and whole body are all of a similar magnitude and at best only moderate, whilst those for cortical EMD are even poorer. Similar relationships between pQCT and DXA hip and spine BMD measurements have been reported previously using similar pQCT scanners with an single energy x-ray source (1.50,3.1), and a predecessor using a ¹²⁵-I energy source (6.7). To my knowledge there no reports comparing pQCT measurements to whole body EMD.

Similarly, correlations were poor between all pQCT measurements and both calcaneal ultrasound measurements, confirming previous findings (1.50). The best correlation was between BUA and radial trabecular BMD, which is perhaps not surprising as the os calcis consists of 90-95% trabecular bone (1.21). This differs from the results of a previous study where BUA correlated better with radial mid-shaft BMD (predominantly cortical bone), rather than with trabecular BMD measured at a distal site. The pQCT scanner which was specially built, was

95

n Na Ait - a a different to the Stratec 960, and the study population was much smaller, with only 24 females being studied (6.8).

그는 가슴에 가지 않는 것 같은 것 같은 것 같은 것 같은 것 같은 것 같은 것 같이 있는 것 같은 것 같이 많이 했다.

(1) 1 + (1) 1 + (1)

These results suggest that the relationship between ultradistal radial pQCT BMD measurements and those at other skeletal sites (ipsilateral arm, lumbar spine, proximal femur, whole body and calcaneus) are at best only moderate. As such, they do not reflect, and could not be used to predict with any accuracy, BMD at other skeletal sites. Whilst these results could be explained in part by our heterogeneous population, differences in the proportion of cortical and trabecular bone between skeletal sites, scanner precision, and thinness of pQCT measurement slice, it is much more likely that they further illustrate regional variation of bone density within individuals, which has been observed previously (6.1, 6.2, 6.3).

For females considered "at risk" in the lowest quartile of lumbar spine measurements, about 50% would be missed based upon pQCT (other than cortical BMD) measurements, and 70% missed based upon pQCT cortical BMD. For women "at risk" in the lowest quartile of femoral neck measurements, 70% - 80% would be missed based upon pQCT measurements. Vice versa, basing risk upon quartiles of pQCT measurements would result in many women being wrongly categorised as "at risk" at the hip and spine. To detect all women with either hip or spinal osteopenia (T score < -1.0) based upon pQCT measurements, almost the whole</pre> perimenopausal population would have to be scanned to ensure all cases were detected. The results for calcaneal ultrasound measurements were comparable to pQCT. It is also worth noting that to detect all women with femoral neck osteopenia based upon a lumbar spine scan, 88.4% of the study population would have to be scanned, and vice versa, 96.3% scanned. These data suggest that due to the poor correlation between BMD assessment at various sites irrespective of scanning modality, site

96

specific or at least two-site assessment of risk is preferable in those requiring fracture risk assessment. Two-site assessment has previously been shown to improve the predictive value for subsequent vertebral fracture (1.191,6.9,6.10).

化氨基乙基 化化化乙基化 化乙基

These results show that pQCT would be of no value in preselecting individuals for further axial DXA measurements should a screening program for osteopenia become reality. However, calcaneal ultrasound measurements have been shown to predict future hip fracture (see chapter 8) and there is convincing evidence that peripheral bone density measurements are useful in determining fracture risk. Although hip fracture risk is best determined by a site specific measurement (1.192), prospective studies have shown that radial BMD measurements, using osteo-densitometers other than pQCT, can predict fracture risk at the hip (1.192) and spine (6.5). Prospective studies are required before similar claims can be made for pQCT. Table 6.1. Details of groups 1 and 2. Group 1 is divided into premenopausal (FRE), postmenopausal (POST) and hysterectomized (HYST) subgroups. Unless otherwise stated, values are mean(SD). § is median(range).

		Group 1		Group 2
μ μ	PRE	POST	HYST	ALL POST
Number 1	101	67	48	45
Age (yrs) 4	±7.2 (1.3)	50.6 (2.6)	48.9 (2.2)	75.1 (7.7)
Height (m) 1	1.62 (0.06)	1.59 (0.06)	1.61 (0.07)	1.57 (0.06)
Weight (kg) 6	57.6 (12.5)	65.6 (11.5)	63.6 (11.2)	61.4 (9.6)
Number on HRT	co	20	14	0
Vumber on tamoxifen	н	Cł	0	0
Number on oral corticosteroid	Ч	0	r-1	0
Number on thyroxine	ň	2	ſ	Ŵ
Number with Rheumatoid Arthriti	LS 0	0	+ ^م]	H
Vumber with asthma	თ	4	4	N
Number with diabetes	7	0	0	m
ľears postmenopause §	ł	3 (0.5 - 15)	ı	30.4 (9.7)
Vears since hysterectomy §	I	I	9 (0.5 - 24)	ţ

Table 6.2. pQCT, DXA hip and spine BMD, and calcaneal ultrasound measurements of the perimenopausal population (Group 1).

nee walf for the new construction of the state of the state

1999 A.A.A.

(). .

	·····	Mean	SD	Min.	Max.
pOCT (B	MD, q/cm^3 : n=216)		····•		
Qt	ot	380.1	58.5	205.9	578.1
Qt	rab	182.3	41.0	56.8	341.7
Qs	cort	521.8	69.0	306.6	697.1
Qc	ort	554.4	61.2	380.4	705.3
DXA (BM	D,g/cm ² : n=216)				
Lu	mbar spine (LS)	1.030	0.170	0.701	1.736
	T-score	-0.81	1.46	-3.64	5.28
Fe	moral neck (FN)	0.860	0.130	0.580	1.222
	T-score	-0.90	1.08	-3.25	2.85
Tr	ochanter (FT)	0.690	0.120	0.411	1.037
Wa	urd's area (FW)	0.670	0.140	0.409	1.135
Ultrasc	ound				
At	tenuation (BUA)	77.8	17.6	39.0	144.0
(α (π	elocity (VOS) n/s: n=209)	1405	54	1151	1575

Table 6.3. pQCT total (Qtot), trabecular (Qtrab), subcortical (Qscort) and cortical (Qcort) BMD and DXA whole body (WB-BMD) and arm (WB-ARM) BMD of Group 2.

.

1.114

Sector States and the sector of the sector o

1.14

		Mean	SD	Min.	Max.
3MD, g	/cm³: n=45)	····	<u> </u>		
cot		288.8	66.1	193.5	448.1
rab		151.6	54.5	19.5	262.7
scort		397.6	86.1	258.4	557.4
cort		455.2	77.7	267.9	621.9
MD,g/c	m²)				
3-BMD	(n=45)	0.871	0.110	0.656	1.157
3-ARM	(n=44)	0.723	0.112	0.473	0.982
	BMD, g ot cot cort dD,g/c B-BMD B-ARM	BMD, g/cm ³ : n=45) tot trab scort cort MD,g/cm ²) B-BMD (n=45) B-ARM (n=44)	BMD, g/cm^3 : n=45) tot 288.8 trab 151.6 scort 397.6 cort 455.2 MD, g/cm^2) 8-BMD (n=45) 0.871 8-ARM (n=44) 0.723	BMD, g/cm^3 : n=45) tot 288.8 66.1 trab 151.6 54.5 scort 397.6 86.1 cort 455.2 77.7 MD, g/cm^2) 8-BMD (n=45) 0.871 0.110 8-ARM (n=44) 0.723 0.112	BMD, g/cm^3 : n=45)cot288.866.1193.5crab151.654.519.5scort397.686.1258.4cort455.277.7267.9MD, g/cm^2)0.8710.1100.6568-BMD (n=45)0.7230.1120.473

Table 6.4. Cut-off value and proportion of population below this value for each of the DXA, pQCT and ultrasound measurements to detect all women with T scores \leq -1 at lumbar spine (n=97) and femoral neck (n=108). Values are: bone density cut-off value (percentage of population)

. . .

5 (QC)

1.1

(a) the product of the second state of the

		Lumbar spine	Femoral neck
DXA	$(BMD,g/cm^2)$		
	LS	-	1.236 (88.4)
	FN	1.104 (96.3)	_
	FT	0.835 (89.4)	0.751 (71.3)
	FW	1.023 (98.6)	0.812 (85.6)
pQCT	$(BMD, g/cm^3)$		
	Qtot	578.1 (100)	578.1 (100)
	Qtrab	278.1 (98.6)	278.1 (98.6)
	Qscort	688.6 (99.1)	688.6 (100)
	Qcort	693.8 (99.5)	705.3 (100)
Ultra	asound		
	BUA (dB/MHz)	144 (100)	122 (98.6)
	VOS (m/s)	1496 (97.1)	1514 (99.0)





102

95% confidence limits for regression equations shown



Fig 6.2: Relationships of Qtot with Qtrab, and of Qcort with Qscort



Fig 6.3: Relationships of Qtrab with Qscort and Qcort





95% confidence limits for regression equations shown



Fig 6.5: Relationships of pQCT Qtot with DXA FT and FW



Fig 6.6: Relationships of pQCT Qtot with ultrasound BUA and VOS

107

1.50



Fig 6.7: Relationships of pQCT Qtot with DXA WB-BMD and WB-ARM



Fig 6.8: Relationships of pQCT Qtrab with DXA LS and FN



Fig 6.9: Relationships of pQCT Qtrab with DXA FT and FW



Fig 6.10: Relationships of pQCT Qtrab with ultrasound BUA and VOS



Fig 6.11: Relationships of pQCT Qtrab with DXA WB-BMD and WB-ARM















Fig 6.15: Relationships of pQCT Qscort with DXA WB-BMD and WB-ARM





95% confidence limits for the regression equations shown










Fig 6.19: Relationships of pQCT Qcort with DXA WB-BMD and WB-ARM



Fig 6.20: Proportions of females in the lowest quartiles of DXA hip (FN, FT, FW) and spine (LS) measurements also in the



Fig 6.21: Proportions of females in the lowest quartiles of pQCT measurements (Qtot, Qtrab, Qscort, Qcort), also in the lowest quartiles of ultrasound measurements (BUA, VOS)

_

AGE RELATED CHANGES IN RADIAL pQCT BMD MEASUREMENTS: COMPARISON WITH OTHER SKELETAL SITES.

7.1 Introduction

Changes in bone mass throughout the lifetime of an individual have been discussed in detail in chapter 1.1.3, and are depicted in figure 1.2. Peak bone mass (PBM) is now generally considered to be achieved by the end of the second decade (1.44, 1.60-1.65), and possibly a few years later (age 25) at the radius (1.40). PBM is generally maintained throughout the following 2-3 decades before bone loss commences. Controversies reqarding the age at which PBM is achieved, and timing of bone loss from PBM are also discussed in chapter 1.1.3. Generally, age related bone loss occurs in all individuals from all skeletal sites, and affects both trabecular and cortical bone. This type of bone loss leads to what has been termed Type 2 osteoporosis, leading eventually to hip fractures (1.67). Due to ovarian failure and subsequent oestrogen deficiency, women experience an accelerated phase of bone loss following the menopause. Although all skeletal sites are affected, there is controversy, especially at axial sites, as to the exact timing anđ degree \mathbf{of} bone loss (1.51,1.68-1.72,1.74,1.79-1.82,5.1,5.2). Trabecular bone is metabolically more active than cortical bone, and is therefore thought to be preferentially affected during this phase of accelerated bone loss. Consequently, bone loss is greater from skeletal sites rich in trabecular bone. This type of bone loss leads to what has been termed Type 1 osteoporosis, and fractures especially of the wrist and vertebrae (1.67).

Most of the work done to date examining patterns of radial bone loss have employed either SPA or DXA, which give a composite

123

assessment of trabecular and cortical BMD at the measurement site. pQCT, with its ability to quantify trabecular and cortical BMD independently may give additional information as to the timing, pattern and extent of bone loss at the radius. It should be more sensitive than radial SPA or DXA to subtle BMD change, particularly within the trabecular component. It should be borne in mind however, that for any given individual, a change in BMD of ± 2.8 x the co-efficient of variation is necessary before a BMD change can be accepted as being outwith precision error (1.88,7.1). The pattern of bone loss as determined by pQCT was therefore examined in 4 different female populations, which are detailed below. Changes in pQCT radial BMD were compared to axial measurements in three of the groups

7.2 Study Populations and Bone Mass Measurements

The following different female populations were studied.

Cross-Sectional Design

- Group 1. Changes in pQCT measurements alone were obtained from analysis of the study population used to create our normal range. This population was defined in chapter 5.2, and has an age range of 18-90 years. 213 females were premenopausal (defined as ongoing menses) and 119 were postmenopausal (defined as no menses for at least 12 months)
- Group 2. Changes in pQCT BMD measurements were compared to DXA lumbar spine and hip BMD measurements in а perimenopausal group aged 45-55 years. The study population described in chapter 6.2 (group 1) was analysed further. From the original 216 females, those with a known history of thyroid disease, diabetes, asthma and inflammatory arthritis were excluded as were those taking drugs known to affect bone metabolism and those who hađ undergone hysterectomy, leaving 119 normal women. 79 women were

124

n an tair an an an an an tair a

premenopausal, and 40 postmenopausal.

Longitudinal Design

- Group 3. 23 normal premenopausal females who were attending an osteoporosis screening programme (described in chapter 5: 5.2-point 2) were invited to return approximately one year later for re-scanning. None were known to have any condition or were taking medication which could influence bone mass. pQCT was performed within 8 weeks of DXA lumbar spine and hip measurements at the baseline visit, but both were performed on the same day at the follow-up visit. All women were menstruating regularly at both baseline and follow-up visits.
- Group 4. 26 normal postmenopausal females underwent pQCT and DXA hip and spine measurements on the same day at baseline and approximately 18 months later. None were known to have any condition or were taking medication which could influence bone mass. This study population is derived from that described in chapter 5: 5.2-point 3. Of the original 31 females, 26 females agreed to return for re-scanning, to form the basis of this study population.

For the follow-up pQCT measurements, the "trending" function, which is discussed in chapter 3, was utilised. The cut-off value used for the difference in voxel numbers between baseline and follow-up scans was set at 30. This should have ensured accurate positioning of the scanner in follow-up scans, thus minimising the influence of change in scan site on follow-up pQCT BMD measurements.

7.3 Statistical Analyses

Results for continuous variables were expressed as the mean with standard deviation if the data were normally distributed,

otherwise as the median with ranges.

7.3.1 Cross-sectional Designs

The relationships of pQCT BMD values with age in the whole population aged 18-90 years (group 1), were examined using linear and non-linear regression, exponential and logarithmic analyses. The model which best fitted the data was determined by minimising the residual sum of squares and improving the correlation co-efficient. Similar analyses were performed on premenopausal women as a function of age, and postmenopausal women as a function of years since the menopause. From the resultant equations, annual rates of change were determined, as a function of age for premenopausal women, and years postmenopause for postmenopausal females at 5, 10, 20, 30 and 40 years postmenopause.

In group 2, between group comparisons were performed using the unpaired t-test. Differences between premenopausal and postmenopausal subgroups were expressed as percentages. Taking into account the differing ranges for BMD measurements, Tscores were also generated and compared for the postmenopausal group, using the premenopausal group as the young normal population. The differences are expressed as T-scores rather than Z-scores, as obviously the populations were not aged matched, and pQCT BMD in premenopausal women studied crosssectionally showed little change (see results section of this chapter).

7.3.2 Longitudinal Designs

In the longitudinal designs, between group comparison of baseline BMD measurements were performed using the unpaired ttest. Annualized rates of change in BMD were calculated, and within group annualized percentage BMD changes from baseline were compared using the paired t-test.

7.4 Results

126

المربوع والمحادث والمراجع والمراجع والمراجع والمراجع والمراجع والمراجع والمراجع

7.4.1 Cross-sectional studies

and the second

The changes in pQCT BMD values in a female population aged 18-90 (group 1) are shown in figures 7.1 - 7.4. For the whole population, cubic regression was superior to other models in describing age related changes for all pQCT BMD measurements as function of age (see figures 7.1-7.4 for regression а equations). There was a decrease in BMD from peak bone mass to that of old age of 34.3%, 31.7%, 33.7% and 24.4% for Qtot, Qtrab, Qscort and Qcort respectively [excluding the age range 30-39 years, which is probably not representative of the general population, due to the small number of subjects studied (n=6)]. Linear regression equations of the premenopausal females showed that there was little change in any of the pQCT BMD measurements as a function of age between the end of the second and fifth decades. The relationship with age was positive for Qtot (Qtot = 376.4 + 0.3(age), r=0.053, p=NS), and negative for Qtrab (Qtrab = 186.4 - 0.01(age), r=-0.003, p=NS), Qscort (Qscort = 545 - 0.17(age), r=-0.024, p=NS) and Qcort (Qcort = 567.7 - 0.11(age), r=-0.016, p=NS). The resultant annual rates of changes in premenopausal women were small: Qtot (+0.3mg/cm³/yr; 0.08%/yr), Qtrab (-0.01mg/cm³/yr; -0.005%/yr), Qscort (-0.17mg/cm³/yr; -0.03%/yr) and Qcort (-0.11mg/cm³/yr; -0.02%/yr).

しきん さいしき ほどうどう

postmenopausal females, all For BMD measurements were negatively related to the duration since menopause, and are shown in figures 7.5 and 7.6. A cubic regression model best fitted Qtot, Qscort and Qcort BMD data, whereas a parabolic model best fitted Qtrab BMD (see figures 7.5 and 7.6 for regression equations). The annual rates of changes for postmenopausal women at 5, 10, 20, 30 and 40years postmenopause (YPM) derived from the regression equations for each of the BMD measurements are shown in table 7.1. For Qtrab, the rate of loss gradually increased from -0.53%/yr at 5 YPM to -0.76%/yr at 40 YPM. The greatest rate of loss in the early postmenopause years (5 and 10 YPM) was found for Qscort (-

127

an a bha chu<u>ch i seo a</u> seach

0.72%/yr and -1.12%/yr respectively) and Qtot (-0.79%/yr and -1.12%/yr respectively), with that for Qcort being less (-0.5%/yr and -0.95%/yr respectively). The greatest overall rates of loss for Qtot, Qscort, and Qcort were found at 20 YPM: -1.36%/yr, -1.39%/yr and -1.39% respectively. Qtot, Qscort and Qcort all increased at 40 YPM: +0.85%/yr, +1.43%/yr and +0.75%/yr respectively.

"我们就是我们的,你也不知道,你说道:我们怎么想你们的?""你们就是你的。"

Changes in radial BMD around the menopause were further examined in a perimenopausal population (group 2) and compared to DXA spine and hip BMD. 79 premenopausal females (PRE subgroup) were compared to 40 postmenopausal females (POST subgroup). Details of the study population are shown in table 7.2. The PRE group was 4.1 years younger and 3 cm taller than the POST group, but were well matched for weight. The median duration since the menopause in the POST subgroup was 35 months (range 12-180). All bone mass measurements were lower in the POST subgroup and are shown in table 7.2, and figures 7.7 and 7.8. There were statistically significant differences in all BMD measurements except Qcort and Qtrab. Percentage BMD differences for axial DXA measurements [-13.48%, -11%, -9.81% and -9.78% for FW, FT, FN and LS respectively] were greater than those for pQCT [-7.63%, -6.43%, -5.49% and -1.75% for Qtrab, Qtot, Qscort and Qcort respectively]. T-score differences between the premenopausal and postmenopausal subgroups (using the premenopausal subgroup as the reference population) followed a similar pattern to the percentage differences and are shown in figures 7.7 and 7.8. Differences [T-score(SD),p value] for axial DXA measurements: \mathbf{FT} [-] **0.81(1.29), p<0.001], FW** [-0.77 (1.01), p<0.001], FN [-] 0.76(1.09), p<0.001] and LS [-0.61(1.26), p=0.005] were greater than those found for pQCT measurements: Qscort[-0.58(1.34), p=0.018, Qtot [-0.54(1.3), p=0.013, Qtrab [– 0.35(1.2), p=0.1] and Qcort [-0.29(1.24), p=0.17].

128

and the state of the second second

1 and a disc

Construction of the annual states of the

Mr. Barren and and

7.4.2 Longitudinal Studies

Demographic and BMD data for groups 3 and 4 are shown in table 7.3. At baseline, the premenopausal group (group 3) were younger (16.3 yrs), taller (4.7cm) and heavier (2.7kg) than their postmenopausal counterparts (group 4). Accordingly, BMD for all radial, hip and spine measurements were greater in group 3, although this failed to reach significance for Qtrab and FN. Correcting for the influence of height and weight, significant differences remained for all except for Qtrab (p=0.198), FN (p=0.065) and FT (p=0.057). The variance of all BMD measurements was greater in group 4 than in group 3. There were no significant changes in height or weight in either group between baseline and the follow-up visits (group3: height 1,613 \pm 0.062m vs 1.614 \pm 0.06m; weight 65.9 \pm 12.5kg vs 66.2 \pm 12kg) (group 4: height 1.566 ± 0.063m vs 1.568 ± 0.063m; weight 63.2 ± 10.7 kg vs 62.9 ± 11 kg). Annualized rates of change in pQCT radial BMD and DXA hip and spine BMD for both groups are shown in table 7.4, and figures 7.9 and 7.10. There were statistically significant annualized decreases in Qtot in both groups, and in Qscort in group 3. Mean changes in other pQCT measurements were negative in both groups with the exception of Qtrab in group 4, but none reached significance. There were no significant changes in DXA hip or spine BMD measurements in groups 3 or 4 with the exception of FN in group 4. The variance in percentage change was large for all measurements and tended to be greater in group 4. As can be seen for figures 7.9 and 7.10, very few individual changes fell outwith the 2.8 x CV band indicating that a real change could only be confirmed in very few patients. The postmenopausal group (group 4) was further analysed according to the duration since menopause. There was no difference in the annualized rates of change between those within 10 years of their menopause (n=9), and those more than 10 years from their menopause (n=15), although the numbers in each subgroup were small.

The relationship between the annualized percentage rates of

129

and the second of the state of the second second

change in pQCT measurements and those of LS and FN for premenopausal females (group 3) are shown in table 7.5, and for postmenopausal females (group 4) are shown in table 7.6. Similar relationships were found with \mathbf{FT} and FW BMD measurements, but are not shown. There were no significant relationships between annualized rate of change between pQCT and DXA LS and FN measurements in either premenopausal or postmenopausal groups. The relationships between changes in Qtot and LS in premenopausal and postmenopausal females are shown in figure 7.11, as are those for Qtot and FN in figure 7.12. The relationships of other pQCT measurements were similar but are not shown. However, the relationships between annualised percentage rates of change at LS and FN in both group 3 and 4 were also insignificant, and are shown in figure 7.13.

网络哈伦 化二氟化化化 医无骨骨 医无关的 法法律的 化合物 化合物 医白色素

7.5 Discussion

radial pQCT BMD measurements decreased with age, the / All reduction between peak bone mass and old age being 24.4%-34.3% depending upon the measurement. Studying a population with a wide age range, as there was in group 1 (18-90 years), a cubic regression model found to best fit was all pOCT BMD measurements. This has been found previously for trabecular BMD determined by QCT at the radius (7.2) and lumbar spine (1.74,7.2). When the cross-sectional data derived from group 1 were examined by decades (see tables 5.1 & 5.2), the various BMD values seem stable until the end of the sixth decade. This method of analysing data can mask the effects of the menopause upon BMD, as its age of onset varies between women. Dividing the population by menopausal status should help clarify the effect of the menopause upon radial BMD. In the premenopausal subgroup of group 1, there must be doubt whether those females in the age group 30-40 years (n=6) are representative of the general population. This negates any meaningful attempt to determine the timing of peak bone mass in this population. Nevertheless, fitting a linear regression model should allow

130

the trend of BMD in premenopausal women to be determined, as the 10 year age bands on either side of the 4th decade are sufficiently large to be representative of the premenopausal population. Such a model showed that all radial pQCT BMD measurements were effectively stable in the premenopausal subgroup. This is in keeping with almost all of the published work done using both SPA (1.63,1.51,1.68,1.69,1.70,1.79) and pQCT (1.50,1.78). However, reduced radial trabecular BMD has been found in regularly menstruating women with evidence of declining ovarian function (7.3).

이 가지 않는 것이 되었다. 이 가지 않는 것이 가지 않는 것이 가지 않는 것이 가지 않는 것이 가지?

In contrast to the premenopausal subgroup, there was clear evidence of significant reductions in pQCT BMD measurements in the postmenopausal subgroup of group 1. A cubic regression model best fitted Qtot, Qscort and Qcort BMD data as a function of years since the menopause, whereas a parabolic model best fitted Qtrab. Surprisingly, the derived rate of loss for Qtrab gradually increased from 5 to 40 years postmenopause, which is contrary to the theory that an accelerated phase of trabecular bone loss occurs immediately following the menopause, at least at the radius. Additionally, derived rates of loss for Qtrab were much lower than those for Qtot, Qscort and Qcort in the early postmenopausal years (5 to 20 years postmenopause). The greatest rates of loss were found for Qtot, Qscort and Qcort at 20 years postmenopause(-1.36%/yr - -1.39%/yr). These findings for Qtot, Qscort and Qcort can be explained by the recent observation that postmenopausal remodelling occurs at the inner the cortical shell with trabecularisation of aspect of endosteal cortical bone. Consequently, progressive expansion of the cross-sectional area of trabecular bone occurs at the expense of cortical area in postmenopausal women (1.75,6.6). A more pronounced decrease in pQCT total BMD compared to trabecular BMD has been noted previously from cross-sectional data (1.50,6.6). The estimated rates of bone loss (from crosssectional data) for the postmenopausal subgroup of group 1 in this thesis, are comparable to other estimated rates of loss

131

ta a na sana a <u>na na na na k</u>alangan ang baranta na kalangan kana ka

(cross-sectional data) determined using pQCT (1.50,6.6), SPA (1.68,1.69), and DXA (7.4), and observed rates of loss (longitudinal data) determined using pQCT (1.75,1.78) and SPA (1.73). There was also evidence of ongoing loss for all pQCT BMD measurements in the elderly. Estimated rates of loss in women 30 years postmenopause were between -0.59%/yr and -0.93%/yr depending upon the measurement, supporting the claims of others that significant bone loss continues into old age (1.86). In contrast, the estimated rates of change at 40 years postmenopause, suggested that radial Qtot, Qscort and Qcort BMD increased in the very elderly. However, many of the women more 40 years postmenopause were recruited through than an advertising campaign requesting elderly, fracture free women to act as normal controls for hip fracture patients. They were not randomly selected from the elderly population, and were therefore liable to several biases, which could have falsely increased BMD, thus reducing or reversing estimated rates of bone loss (1.86). These biases have been discussed in more detail in chapter 1.1.3.

a service a first single state of the

Section 2000

Profound changes in bone metabolism occurs immediately following the menopause due to falling oestrogen levels. This results in increased bone resorption and subsequent bone loss. The effect of the menopause upon radial pQCT BMD was compared to DXA hip and spine BMD measurements, in a cross-sectional study. The study population was a randomly selected group of perimenopausal females aged 45-55 years of age, which was then divided into premenopausal and postmenopausal subgroups based upon menstrual history (group 2). 90% of the postmenopausal females studied in this group were within 8 years of the menopause and thus represent early postmenopausal females. In view of the differences in mean BMD values and ranges between pQCT and DXA measurements, comparison of differences is best achieved if the differences are expressed as T-scores (using the premenopausal subgroup as the young normal reference population) rather than percentages. BMD was found to be

132

A state of the second state of the second second

significantly lower at all skeletal sites in postmenopausal females. The greatest differences (based upon T-scores) were found for the three DXA hip measurements, particularly the trochanteric site and Ward's area, which are the trabecular rich sites within the hip. Differences for all pQCT BMD measurements were lower than those found at the hip and spine. At the radius, the greatest differences (based upon T-scores) found for Qtot (both and Qscort statistically were significant), whereas those for Qtrab and Qcort were insignificant. This suggests that perimenopausal and early postmenopausal bone loss is less marked from radial compared to axial sites, supporting the findings of others (1.77,7.4). These findings also suggest that perimenopausal bone loss at the radius primarily occurs from the subcortical region, supporting the findings above derived from group 1, and previously reported work (1.75,6.6). A similar insignificant reduction in radial trabecular BMD has also been reported previously from a comparable study, with a population similar but smaller to group 2 reported here (7.5).

Longitudinal studies should resolve the issues of timing and rate of bone loss. Almost all longitudinal studies have shown that there is no significant premenopausal loss in radial bone mass when assessed using SPA (1.63,1.69,1.70,1.79). The premenopausal females studied longitudinally in this thesis (group 3), are all older than 45 and are therefore likely to be nearing the menopause. Observed annualized rates of change showed mean total and subcortical BMD reductions of -1.04% and -1.35% respectively, both of which were statistically significant. Trabecular and cortical BMD changes were much smaller and insignificant. This suggest that there may be radial bone loss immediately preceding the menopause within the subcortical area, in a similar manner to that described in postmenopausal women above (Chapter 7: groups 1 & 2), and previously (1.75,6.6). It may well be that although these premenopausal women were still menstruating regularly, their

133

y dalah <u>adalah sahiji</u> sa

ي من هم المربق ال cestrogen levels were diminishing, but unfortunately they were not assessed as part of this study.

There is convincing evidence from longitudinal studies using both SPA (1.70,1.73,1.80,5.3) and pQCT (1.75) that bone loss occurs from the radius in postmenopausal females. The studies using SPA have found rates of bone loss around 1%/yr. The study using pQCT (1.75) found a greater annual rate of loss of 2.3% for total BMD and 2.8% for trabecular BMD, whilst cortical BMD remained stable. This latter study examined females within 4 years of the menopause, and supports the concept of accelerated bone loss immediately following the menopause which preferentially affects trabecular bone. In comparison, there was no significant change in trabecular BMD $(+0.3 \pm 3.04 \%/yr)$ in the group of postmenopausal females studied longitudinally in this thesis (group 4; table 7.4). Although observed mean annual rates of loss were noted for subcortical (-4.1%/yr) and cortical BMD (-4.1%/yr), these annual changes failed to reach statistical significance due to the large variances observed. The observed mean rate of loss for pQCT total BMD was 1.36%/yr which reached statistical significance. The large variance in rates of change found generally in the postmenopausal females could be related to the precision of the scanner, but also probably reflects inter-individual variation in rates of bone loss, and differing rates of bone loss in postmenopausal females at different time points from the menopause. The disparity from previous results using pQCT (1.75) can be explained by the different study populations as our population was older, and the mean duration since menopause was 13.3 years with a large variance.

The mean annual rates of change found at the radius using pQCT in all these populations (groups 1-4) would be clinically insignificant if found in an individual, bearing in mind a change of \pm 2.8 times the coefficient of variation for each BMD measurement is necessary before it can be assumed that a

134

The Alexandre prototy

and the second with the first standard

clinically relevant change in BMD has occurred, and that the change is not due to scanner error (1.88). Based upon the changes observed in group 3, an interval of around 3 years would be necessary between scans for detection of clinically important changes in total and subcortical BMD in an individual premenopausal female nearing the menopause. Ιt is more difficult to be accurate about the corresponding interval in postmenopausal females, as this would be determined by the precision of pQCT in such females. Unfortunately this has not been formally assessed by us, but the precision is likely to be that between that of normal young females, anđ of postmenopausal females with severe osteoporosis and vertebral fracture (see chapter 3). If this assumption is accepted, then, based upon the observed annual rates of change in group 4 (longitudinal data in postmenopausal women), and the estimated annual rates of change in the postmenopausal subgroup of group 1(cross-sectional data in postmenopausal women), the corresponding interval in normal postmenopausal females would again be at least 3 years, with total and subcortical BMD measurements being the most useful measures of change.

Existing evidence from the literature suggests that changes in BMD at the radius differs from that at the axial skeleton both in timing and rate. Whilst radial BMD is stable until the menopause, there is some evidence to suggest premenopausal bone loss (1.63, 1.68, 1.72, 5.1, 5.2)a.t both hip spine and (1.51,1.63,1.69,1.70,1.71,1.74,1.83). Others have found spine (1.68,1.77,1.79,5.1) bone mass stable until the menopause with loss limited to postmenopausal females. An accelerated phase of (1.81, 5.1, 5.2)bone loss at the hip and spine (1.51,1.71,1.73,1.74,1.81,5.1,5.3) immediately after the menopause has been recorded, whilst others have found a linear decline at hip (1.72,5.3) and spine (1.68,1.69,1.70). The percentage difference between PBM and BMD of old age, and rates of loss, have been found to be greater at the hip and spine compared to radius in most of the previous work. Generally,

135

and the second secon

annual rates of loss at axial sites exceed 1% and tend to be greater in postmenopausal females in both cross-sectional and longitudinal studies, with rates of loss at the spine as great as 6%/yr being found in the early postmenopausal years.

しかか ない 投入者 いいかえた ひょうかい かいぶつび

- H = 1,400

時間 秋天 化プロ・気化 いいかいかんばく かいわせい たいかいがい

Hip and spine BMD in the premenopausal females (group 3) studied longitudinally in this thesis were found to be stable, with no evidence of bone loss. Similarly, there was no evidence of spinal bone loss in the postmenopausal group. There was however a significant annual decline of 2.47%/yr in femoral neck BMD, although changes in femoral trochanter and Ward's area were insignificant. The small numbers studied in each group, the composition of the postmenopausal group (see above), the interval between repeat measurements and the precision of the DXA scanner are likely to be responsible for the largely stable DXA measurments in groups 3 and 4. As for radial pQCT measurements, the precision of DXA and rates of change recorded at axial sites are such that annual changes are not outwith scanner error, and an interval between scans greater than that studied in this thesis (median value: 15 months for group 3, 18 months for group 4) is necessary for clinically significant change to be detected in an individual.

The relationships between the observed rates of change in the longitudinal studies (groups 3 & 4) for radial pQCT BMD measurements and changes at either lumbar spine or hip were very poor. In the premenopausal group, the relationships between radial, and lumbar spine and femoral neck BMD changes were all positive except radial trabecular BMD versus lumbar none reached statistical spine. but significance. The relationships in the postmenopausal group were even poorer, such that all changes in pQCT measurements were negatively related to changes at the femoral neck. Although the corresponding relationships with lumbar spine changes were all

136

and the second secon

positive, none reached significance. A similar relationship has been found before between lumbar spine and radial BMD changes measured by DPA and SPA respectively (1.70). It is worth noting however, that the relationship between changes at the femoral neck and lumbar spine was equally poor in both premenopausal and postmenopausal populations studied in this thesis. The relationship was stronger in the postmenopausal compared to the premenopausal group, but failed to reach significance in either. This suggests that BMD changes occurring at one skeletal site are not necessarily representative of changes at another over the short-term. Change must therefore be determined from site specific measurements, and cannot be extrapolated from measurements at other sites.

÷

ž,

والمراجعة والمراجع

化二氟化二甲基氟化二化氯化二甲酸 网络经常规模 化分析机 网络

1). Rates are derived as a function of age in premenopausal (PRE) women, and years since the menopause Table 7.1. Annualized percentage rates of change of pQCT BMD measurements in women aged 18-90 years (group (5,10,20,30 & 40 years) in postmenopausal (POST) women.

	PRE			POST		
		5yrs	10yrs	20yrs	30yrs	40yrs
Qtot	+0.08	-0.79	-1.12	-1.36	- 0.77	+0.85
Qtrab	-0.005	-0,53	-0.56	-0.62	-0.68	-0.76
Qscort	-0.03	-0.72	-1.12	-1.39	-0.59	+1.43
Qcort	-0.02	-0.5	-0.95	-1.39	-0.93	+0.75

values are s/year.

i ji C

and (PRE)Table 7.2 Details of age, height, weight, radial pQCT BMD measurements for premenopausal postmenopausal (POST) subgroups of group 2. Values are mean (SD) unless stated otherwise.

	PRE (n=79)	POST (n=40)	t-test p value	\$diff	3
Age (yrs)	47.1 (1.29)	51.2 (2.46)	< 0.001		1
Height (cm)	162 (6)	159 (6)	0.013	1	
Weight (kg)	66.9 (12.3)	56.0 (11.59)	0.698	I	
pQCT (gm/cm³)					
Qtot	391.8 (52.9)	366.6 (68.6)	0.013	-6.43	
Qtrab	187.5 (38.8)	173.2 (46.6)	0.098	-7.63	
Qscort	533.6 (59.4)	504.3 (79.8)	0.018	-5.49	
Qcort	561.0 (53.4)	551.2 (66.0)	0.174	-1.75	
DXA (gm/cm^2)					
2 I	1.063 (0.16)	0.959 (0.20)	0.003	-9.78	
ЫN	0.897 (0.12)	0.809 (0.13)	<0.001	-9.81	
ΕĽ	0.718 (0.10)	0.639 (0.13)	100.0	-11.0	
ΕW	0.705 (0.13)	0.610 (0.13)	< 0.001	-13.48	

Table 7.3 Demographic data of group 3 (n=23) and 4 (n=26), with baseline pQCT and DXA spine and hip BMD data. Values are: mean (SD), § is median (range). YPM: is years postmenopause. Time: is duration in months between baseline and follow-up visit. ANOVA: is analysis of covariance controlling for height and weight

2 74 74

网络美国人名英格兰人姓氏 化合成分子 化合成分子 医白细胞的 化合成分子 化合成分子 化合成分子 化分子子 化分子子 化分子子 化分子子 化分子子 化分子子

	Group 3	Group 4	t-test	ANOVA
Age (yrs)	47.2 (1.7)	63.5 (6.8)	<0.001	
Height (cm)	1.613 (0.162)	1.566 (0.063)	0.011	_
Weight (kg)	65.9 (12.5)	63.2 (10.7)	NS	-
YPM (yrs)	-	13.3 (7.9)	-	-
Time (mo)§	15 (13-17)	18 (15-19)	-	-
pQCT (gm/cm³)				
Qtot	395.2 (47.1)	349.5 (75.3)	0.011	0.032
Qtrab	190.6 (27.2)	179.1 (44.4)	NS	NS
Qscort	542.8 (70.8)	474.1 (99.2)	0.008	0.037
Qcort	563.2 (63.4)	503.7 (86.4)	0.009	0.035
DXA (gm/cm ²)				
LS	1.08 (0.126)	0.893 (0.16)	<0.001	<0.001
F'N	0.868 (0.094)	0.799 (0.15)	NS	NS
FT	0.714 (0.093)	0.631 (0.128)	0.014	NS
FW	0.693 (0,101)	0.583 (0.170)	0.01	0.024

140

ಂಟ್ ಇರ್ಬುಟ್ಟ

ŝ	ц Ц	
ស្ព	-te	
јгог	і т	
ਮੁ	irec	
Ч Ц	ра Д	
nts	the	
ceme	ыо	
Insi	4⊶I ທ	
nea	ы. Б	
BMD	valu	
ne	ģ	
spi	5D) .	
and	<u>;</u>	
lip	mea	
XA 1	are	
П то	nes	. цо
ац	Val	ati
ial	•	ari
rad	n=26	о Ц
QCT	rj.	nt.
д. н	ausa	icie
ч С	nop	ម្ពុ ម្ពុ ម្ព
nge	tme	- O U
сћа	sod	ល -ក
ate Ate	4	CΛ
ਲ ਸ	and	0
L Z C	(23)	lin
lal:	ี่น=1	ase
Ann	al,	ត ប
۲	aus	₽ ₽
2	doua	1- MC
ble	reme	oll(
Ча Н	р Д	<u>H</u>

te tra ette da

All a Martin services and the services

· ·

. . . .

- 1944 P

		Group 3	Group 4	2 Q	ralue	2.8 x CV
pQCT	(gm/cm³)			(gp3)	(gp4)	
~	Qtot	-1.04 (2.11)	-1.36 (3.06)	0.027	0.032	ተ . 5%
يى	Qtrab	-0.19 (3.17)	0.33 (3.04)	SN	NS	± 3,7%
<u> </u>	Qscort	-1.35 (2.67)	-4.09 (12.9)	0.024	NS	± 4.48
~··	Qcort	-0.95 (2.8)	-4.09 (12.9)	SN	NS	± 5.3%
) AXO	∃m/cm²)					
	LS	-0.34 (3.17)	0.32 (2.47)	SN	SN	± 2,5%
1 1	NA	0.02 (2.97)	-2.47 (4.33)	SN	0.007	± 7.8%
г ,	ЪЭ	0.63 (2.72)	0.94 (4.59)	NS	SN	± 3.6%
1	Ш	2.15 (6.72)	5.62 (13.8)	SN	SN	± 13.7%

Table 7.5 Linear regression equations for group 3 (premenopausal females, n=23) examining the relationships between annualized percentage rates of change (Δ) in pQCT Qtot, Qtrab, Qscort and Qcort BMD measurements with DXA LS and FN BMD measurements.

ALS	SE slope	ы	, D1	ΔFN	SE slope	۶ı	Q
).406(ΔQtot) + 0.09	0.316	0.269	0.22	0.509(AQtot) + 0.55	0.287	0.360	0.09
0.057(AQtrab) - 0.35	0.218	-0.057	0.80	0.273(<u>\0</u> trab) + 0.08	0.196	0.292	0.17
).282(∆Qscort) + 0.05	0.252	0.238	0.28	0.269(∆Qscort) + 0.39	0.236	0.242	0.27
).206(ΔQcort) - 0.14	0.243	0.182	0.41	0.168(AQcort) + 0.18	0.229	0.159	0.47

SE slope = standard estimate of the mean of the slope of regression equation.

= correlation coefficient, p = significance of the regression equation

н

between annualized percentage rates of change (Δ) in pQCT Qtot, Qtrab, Qscort and Qcort BMD measurements with Table 7.6 Linear regression equations for group 4 (postmenopausal females, n=26) examining the relationships DXA LS and FN BMD measurements.

ALS	SE slope	н	<u>р</u>	ΔFN	SE slope	ы	ρ _ι
0.004(∆Qtot) + 0.33	0,164	0.005	0.98	-0.237(AQtot) - 2.79	0.284	-0.167	0.41
0.081(AQtrab) + 0.29	0.165	0.100	0.63	-0.098(AQtrab) - 2.44	0.290	-0.069	0.74
0.013(AQscort) + 0.38	0.039	0.068	0.74	-0.087(AQscort) -2.82	0.066	-0.255	0.21
).012(AQcort) + 0.31	0.059	0.041	0.84	-0.143(Qcort) - 2.350	0.098	-0.284	0.16
					:		

SE slope = standard estimate of the mean of the slope of regression equation. = correlation coefficient, p = significance of the regression equation ч



Figure 7.1: Age related changes in Qtot for normal women aged 18-90 years

Figure 7.2: Age related changes in Qtrab for normal women aged 18-90 years





















Fig 7.7: Differences in pQCT BMD values between PRE and POST females





Fig 7.9: Annualized rates of change in pQCT BMD measurements



1 ... et 1...

 • 3 • • • • • •

1.1251.042

27

Amaniagnesi % change in BMD

Fig 7.10: Annualized rates of change in DXA hip and spine BMD measurements














e di Ali Ali Ali Ali

THE ABILITY OF pQCT TO DISCRIMINATE FRACTURE AND NON-FRACTURE POPULATIONS: COMPARISON WITH OTHER BONE MASS MEASUREMENTS

かんせいだけ いうちょうかい

8.1 Introduction

Osteoporotic fracture is the ultimate event affecting individuals with low BMD. Accordingly, such individuals are categorised as suffering from severe osteoporosis (6.4). Osteoporotic fractures characteristically affects the distal radius, thoracolumbar spine and proximal femur. Such fractures cause significant morbidity and cost the NHS an estimated £740 million per year (8.1). Additionally, the latter two types of fracture are associated with a reduced 5 year survival of 0.82 (8.2) and 0.83 (1.94) respectively, when compared to age matched non-fracture controls. With our increasingly aged population, the problems and financial burden imposed by osteoporotic fractures will undoubtedly rise (1.98).

One of the most important roles of osteo-densitometry is prediction of future fracture, as individuals considered at risk based upon a bone mass measurement can be targeted for life style and therapeutic intervention. There is convincing evidence that radial bone density measurements are useful in determining fracture risk. Although hip fracture risk is best determined by a site specific measurement (1.192), prospective studies have shown that radial BMD measurements, using techniques other than pQCT, can predict fracture risk at the hip (1.192, 8.3) and spine (6.5, 8.3). It is recognised that there is an overlap in BMD assessed at axial sites between age matched non-fracture populations and both hip (1.155,8.4,8.5) and vertebral (1.54,8.6) fracture populations, with the overlap being greater for the former type of fracture. Also, lumbar spine measurements can be falsely elevated by overlying aortic

157

مريون مريد المريد

and the second second second second

الأرارية المحمور وتراري المرأب

calcification (1.168,1.169,1.170), and age related spondylosis and osteophytosis (1.171-1.177), thus limiting its discriminatory power (1.177). Inspite of these problems, spinal BMD measurements have been shown to discriminate vertebral fracture and non-fracture populations (1.173), with spinal QCT being considered the best method (8.9,8.10). Similarly, prospective studies have shown spinal BMD to predict osteoporotic fractures (1.192,6.5,6.10,8.11), with a recent study showing that spinal (and hip) DXA BMD predicts fragility fractures in perimenopausal females (1.203).

医马氏试验检尿道 医二角的现在

The state of a state of the state

There has been much recent interest in the value of calcaneal ultrasound in fracture discrimination and prediction. In the case of hip fracture, the discriminatory power of calcaneal ultrasound attenuation and velocity measurements has been established (8.12,8.13,8.14,8.15), and both prospective (8.16,8.17,8.18,8.19) and retrospective (8.20) studies have shown their predictive power. In the case of vertebral fractures, they have been shown to discriminate (8.8,8.14) and predict prospectively (8.21), and retrospectively (8.22) fracture populations. Generally ultrasound measurements are thought to be more useful in hip than vertebral fracture discrimination and prediction. A preliminary prospective study has also shown ultrasound attenuation to predict fragility fractures in perimenopausal females (1.204), although less well than DXA lumbar spine BMD.

Prospective studies in fracture prediction are required for pQCT. Such studies are clearly longterm, and not really feasible within the time scale of work for this thesis. However, an important related function is the discrimination of fracture and non-fracture populations. Almost invariably discriminatory studies have demonstrated approximately the same relationship between specific sites of BMD assessment and

158

fracture risk as subsequent longitudinal studies. As the vertebral body is predominantly trabecular bone, as is up to 50% of the proximal femur, independent assessment of trabecular and cortical bone at the radius by pQCT may allow differentiation of fracture and non-fracture populations by one or other of the pOCT BMD measurements. The ability of pOCT to discriminate groups of individuals who had suffered a previous vertebral or hip fracture from non-fracture controls was therefore studied, comparing the results with DXA anđ ultrasound measurements done at other skeletal sites.

and a state of the state of the

المتحر والانتصاف والمعاري والمراج

a sexulation established

8.2 Study Populations and Bone Mass Measurements

8.2.1 Vertebral fracture study

with the second second

45 postmenopausal females with vertebral fractures proven on lateral thoracolumbar spine x-ray were recruited from a bone clinic. None had any underlying condition or were taking known to influence medication bone mass. normal 30 postmenopausal females formed the control group. The recruitment of these females has been discussed previously (Chapter 5: 5.2 point 3). Lateral thoracolumbar spine x-rays were performed to exclude vertebral fracture in this control group (1 of the original 31 females had a vertebral fracture and was excluded from this study). All of these females had radial pQCT and DXA of hip and spine performed on a single visit. 30 of the fracture population were scanned using a Norland DXA scanner, whilst the remaining 15 females were scanned using a Lunar DPX- α scanner. The whole control population was scanned using the Norland DXA scanner. Fractured vertebral bodies within the DXA lumbar spine region of interest (L2-L4) were excluded from the analysis. As two different DXA scanners were used in the assessment of the vertebral' fracture group, and there are known to be differences in data acquisition between these scanners (8.23), DXA hip and spine

159

BMD data were standardised using a conversion equation (8.23). These equations have been derived from extensive crosscalibration experiments allowing for the expected deviations between the measured BMD's on each scanner. This allows direct comparison of data derived from the two scanners. The following equations were used for both hip and spine measurements:

in provident de la casa de la casa

 $\mathrm{sBMD}_{\mathrm{N}}$ = 1.0761 $\mathrm{BMD}_{\mathrm{N}}$

 $sBMD_{L} = 0.9522 BMD_{L}$

where s = standardised

N = Norland

and the second and with the product of the

 $_{\rm b}$ = Lunar

8.2.2 Hip fracture study

300 females who had suffered a hip fracture were recruited into a study designed to examine the power of calcaneal ultrasound and different DXA measurements in discriminating hip fracture patients from a control population. Patients who had suffered a previous hip fracture patients were obtained from a search of admission records to Aberdeen Royal NHS Trust for the five year period prior to the study. Individuals were invited by letter to attend for scanning. 64 postmenopausal female controls were recruited from advertising in the local press. None had a history of previous fracture, although thoracolumbar spine xrays were not done to exclude asymptomatic vertebral fractures. All females had the following measurements done on the same day: (1) DXA of the left hip in the control group, and the hip contralateral to the fracture site in the fracture group, (2) ultrasound of the left calcaneus. When personnel trained to perform pQCT were available, including myself, pQCT was also performed. 165 of the hip fracture group, and 45 of the control.

160

<u>All-and arrowing State</u> (and the second s

group had pQCT performed. These females formed the study populations which are reported in this thesis.

法通知 医结肠的 医结核的 网络小白色 医白白色

8.3 Statistical Analyses

Similar analytical methods were used for both vertebral and hip fracture groups. Continuous variables are expressed as mean and standard deviation when normally distributed, median with range otherwise. Comparison between groups was performed by the unpaired t-test. Bone mass measurements were examined having corrected for the influence of age, height and weight by co-variance. Percentage differences analysis of between measurements were calculated, and taking into account differing ranges for these measurements, Z-scores (number of standard deviations above or below the normal mean) were calculated. These z scores were generated using the control group as the reference population in each corresponding studies. Receiver operator curves (ROC) for the different bone mass measurements were created. These reflect sensitivity versus specificity for multiple cut off values. The area under the resultant curve (AUC) indicates the ability of that measurement to discriminate fracture from non-fracture populations. The greater the area under the curve, the more discriminatory the measurement. ROC's were generated, AUC's calculated, and subsequent comparisons made using the "ROCKER" statistical package.

8.4 Results

8.4.1 Vertebral fracture study

Details of the study populations are shown in table 8.1. The fracture group was 4.1 years older (p<0.05), 3.8 cm shorter (p<0.05), 3.2 kg lighter (p=NS) and 6.5 years more postmenopause (p<0.01) than the non-fracture group. The median number of vertebral fractures was 2 (range 1-8). All pQCT BMD values were significantly lower in the fracture group and are shown in figure 8.1: (Qtot: 341.8 \pm 76.3 vs 269.8 \pm 56.3 g/cm³;

161

n an an an an an an an ann an an an an Anna An

Qtrab: 174.7 ± 44.9 vs 114.1 ± 45.8 g/cm³, both p<0.001; Qscort: $465.1 \pm 105.6 \text{ vs } 390.5 \pm 85.4 \text{ g/cm}^3, \text{ p=}0.002)$ with the exception of Qcort (498.2 \pm 91.7 vs 475.9 \pm 87.3 mg/cm³). DXA measurements were also lower in the fracture group and are shown in figure 8.2; (standardised (S) LS: $0.958 \pm 0.174 \text{ vs} 0.726 \pm 0.133 \text{ g/cm}^2$; SFN: 0.843 \pm 0.164 vs 0.667 \pm 0.089 g/cm²; SFT: 0.667 \pm 0.138 vs 0.526 \pm 0.1 g/cm²; SFW: 0.616 \pm 0.176 vs 0.458 \pm 0.92 g/cm², all p<0.001). After correcting for differences in age, height, weight and years postmenopause between the groups, significant differences in BMD remained (see figures 8.1-8.2), with the exception of Qscort (ANOVA, p=0.063). The greatest percentage difference between groups (see figures 8.1 & 8.2) was found for Qtrab (-34.7%), with difference in Qtot, SLS, SFN, SFT and SFW being of a similar magnitude (-21.1%, -24.2%, -26.4%, -21.1% and -25.6% respectively). A smaller difference was found in Qscort (-16%) and Qcort (-4.5%). Taking into account differing measurement ranges, z-score differences between fracture and non-fracture groups (see figures 8.1 & 8.2) were greatest for standardised LS [mean(SD): -1.37(0.78)] and Qtrab [--] 1.35(1.02)]. Z-score differences were of a similar magnitude for standardised DXA hip BMD measurements and Qtot (-0.9 - -1.08), whilst that for Qscort [-0.71(0.81)] was lower, and that for Qcort [-0.24(0.95)] lowest.

The receiver operator (ROC) curves generated for pQCT and DXA measurements, with the corresponding area under the curves (AUC), are shown in figure 8.3. Comparisons of the AUC's for each of the measurements are shown in table 8.2. The largest AUC was for Qtrab (0.853), which was significantly greater than that for Qscort (0.715, p=0.035) and Qcort (0.572, p<0.002), but not Qtot (0.782, p=0.139). The AUC for SLS (0.847) was greater than that for SFN, SFT and SFW (0.822, 0.787 and 0.784 respectively). However, there were no significant differences between the AUC's for the DXA measurements. There was a slight

numerical difference between the AUC for Qtrab and SLS in favour of the former, but this was insignificant (p=0.976). The AUC for Qcort was significantly less than all other pQCT and DXA measurements. Otherwise, there were no significant differences in the AUC's comparing pQCT with DXA measurements, except for SLS which was greater than Qscort (p=0.035).

化二甲酸 医斯勒氏试验检尿道 化偏离化 医小脑的 医黄疸 医手术神经炎的 医肺炎的 化乙基酸甲酸乙基乙基甲酸

8.4.2 Hip fracture study

A start graduate sector of a

Details of the hip fracture and non-fracture groups are shown in table 8.1. The fracture group was 1.2 years younger, 0.8cm shorter, 2.5 kg lighter and 1.4 year less postmenopausal than their non-fracture counterparts, none of which were statistically significant. The mean duration between fracture and study was 3 years. 1 of the control group, and 26 of the hip fracture group self-reported a history of rheumatoid arthritis (χ^2 test, p=0.02). Their exclusion did not change the results, so they are included in the subsequent analyses. Differences in bone mass measurements between groups are shown in figures 8.4-8.6. DXA hip BMD data is missing on 17 of the fracture group due to bilateral hip prostheses. Ultrasound data are missing from 1 control and 11 fracture subjects due to machine malfunction. All pQCT measurements were significantly lower in the fracture group and are shown in figure 8.4: (Otot: 288.8 \pm 66.1 vs 243 \pm 60.7 g/cm³ and Qscort: 397.6 \pm 86.1 vs 334.7 \pm 81 g/cm³, both p<0.001; Qcort: 455.2 \pm 77.7 vs 403.4 \pm 94.1 g/cm³, p=0.001; Qtrab: 151.6 \pm 54.5 vs 126.1 \pm 47.7 g/cm³, p=0.002). All DXA hip measurements were significantly lower and are shown in figure 8.5: (FN: 0.686 ± 0.106 vs 0.576 ± 0.095 g/cm^2 ; FT 0.549 ± 0.114 vs 0.458 ± 0.113 g/cm^2 ; FW: 0.462 ± 0.105 vs 0.39 \pm 0.096 g/cm², all p<0.001), as were ultrasound measurements which are shown in figure 8.6; (BUA: 62.3 \pm 15.8 vs 52.1 \pm 19.6 dB/MHz, p=0.002; VOS: 1485 \pm 49 vs 1423 \pm 64 m/s, p<0.001). Correction for differences in age, height, weight and years postmenopause between the groups did not alter

163

化化物理学院 化结合输出 化氟化物 法法法 化化化物

Att is a constant

ing and a second

A DE ANT TANK

the results. The percentage differences were of a similar magnitude for all pQCT (-16.8%, -15.8% -15.6% and -11.4%, for Qtrab, Qscort, Qtot and Qcort respectively), DXA hip (-16.6%, -16% and -15.6% for FT, FN and FW respectively) and BUA (-16.4%) measurements. There was only a -4.2% difference for VOS. Again, taking into account differing measurement ranges, z-score differences between fracture and non-fracture groups (see figures 8.4 - 8.6) were greatest for FN [-1.04 (0.86)] and VOS [-0.91(4.68)], followed by FT [-0.83 (1.03)] and Qscort [-0.73 (0.94)]. Those for the remaining pQCT measurements, FW and BUA were all of a similar magnitude [-0.64 - -0.68], with Qtrab differences being lowest [-0.48 (0.88)].

197 No. 6 To A. N.

The receiver operator (ROC) curves generated for pQCT, DXA and ultrasound measurements, with the corresponding area under the curves (AUC), are shown in figure 8.7. Comparisons of the AUC's for each of the measurements are shown in table 8.3. The AUC for the ROC curve was greatest for FN (0.796), which was significantly greater than those for all pQCT measurements: (Qtot: AUC=0.693, p=0.028; Qtrab: AUC=0.648, p<0.002; Qscort: AUC=0.698, p=0.037; Qcort: AUC=0.675, p=0.027), FW (AUC=0.738, p=0.01) and BUA (AUC=0.663, p=0.007). Although the AUC for FN was greater than those for FT (0.751, p=0.062) and VOS (AUC=0.783, p=0.818), the differences were statistically insignificant. The AUC for Qscort was greater than other pQCT measurements, but not statistically superior. The AUC for VOS was greater than all of the pQCT measurements, but this reached significance only against Qtrab (p=0.026). The AUC for BUA was similar to those for pQCT, but significantly less than that for VOS (p=0.009).

8.5 Discussion

the second constrained and the second second

These results show that trabecular BMD measured by pQCT at the ultra-distal radial site was the most powerful measurement in

164

Service and a classic sector.

And A Walnut S.

discriminating patients with vertebral fracture from nonfracture controls, by virtue of the greatest AUC of the ROC curve. However there was no statistical superiority over the discriminatory power of DXA lumbar spine and hip measurements for which the AUC's were only slightly less. In contrast, the discriminatory power of the cortical BMD measurement was low, and significantly poorer than all other pQCT, as well as DXA hip and spine measurements. When the discriminatory power of pQCT measurements were compared, that for trabecular BMD was significantly better than that of cortical and subcortical, but not total BMD measurements. These finding largely mirror the zscore differences between fracture and non-fracture groups for each of the BMD measurements - the greatest difference being found for radial trabecular BMD, and the smallest for radial cortical BMD.

化结晶化合物 医结核 医鼻腔 建苯基苯基 化结构 化合物 网络加拿大的人名法尔

ちょうりん ちょうち

In contrast, the power of pQCT to discriminate a hip fracture population from a non-fracture population was less than it was for a vertebral fracture population, as shown by the smaller AUC's. This was true for all individual pQCT BMD measurements comparing the AUC's in the vertebral fracture and hip fracture populations, with the exception of the cortical BMD. The most discriminatory pQCT measurement in the hip fracture study was subcortical BMD, whilst the least was trabecular BMD. Overall, the best discriminatory measurement for hip fracture was DXA femoral neck BMD, which was statistically superior to all pQCT measurements, and calcaneal ultrasound attenuation, but not velocity. In general, the discriminatory power of the pQCT and DXA was poorer for hip fractures than it was for vertebral fractures. This could be explained in part by the smaller zscore differences between fracture and non-fracture groups for corresponding BMD measurements in the hip fracture study compared to the vertebral fracture study (except for Qscort and Qcort). It also suggests that although BMD is an important

165

discriminant of fracture, it is more important in vertebral fracture than in hip fracture.

化二硫酸盐 化氯化化 化热热 化化化物 化化物化物 化化物化物 化分子

It is apparent that factors other than BMD are important in determining fracture risk. More recently, other features of the proximal femur have been cited as being important in fracture predisposition namely femoral neck --length (8.24,8.25,8.26,8.27,8.28), and distribution and architecture of bone (8.24,8.28,8.29,8.30), as assessed on x-ray and DXA Bone turnover rate also predicts hip fracture images. independently of BMD (8.31). The mechanics of a fall are also important for hip fracture (1.26,1.27,1.94,8.32,8.33). The orientation and energy of the fall, the location of impact, and protective mechanisms are all important in determining whether a hip fracture will occur or not (8.32). The importance of falls, which is the most important risk factor for hip fracture, is less important in vertebral fractures (1.94,8.1). For the latter type of fracture, architectural factors have also been found to be of some importance. A smaller crosssectional area of unfractured vertebrae results in an increased mechanical strain for equivalent loads (8.34), and morphometric differences in spinal trabecular bone predispose to fracture (8.35).

To date, very little data has been published on the discriminatory or predictive power of pQCT in fracture populations, especially hip fracture. The superiority of pQCT derived trabecular BMD over DXA lumbar spine BMD in the discrimination of a vertebral fracture population has been cited previously (8.36). In contrast, Grampp et al (8.37) found pQCT to be inferior to spinal BMD measured by QCT, and radial total BMD superior to radial trabecular BMD. The cross-sectional area of radial cortical bone was found to be equally discriminatory as total radial BMD. Several workers have

166

<u>an mine an tao ao amin' am</u>in' amin' ami Amin' amin reported pQCT BMD in vertebral fracture populations (1.78,3.1,6.7,8.38). Grampp et al (3.1) found differences of 10% for trabecular, 8% for total and 1.5% for cortical BMD, which are lower than those reported in this thesis. There was also a greater age difference between their groups than ours. Schneider et al (6.7) found a difference of 21% between groups, but they compared a postmenopausal fracture group (mean age 59 years) with a premenopausal control group (aged 20-40) which are clearly not comparable. However, Ruegsegger et al have found differences of 15-33% in age matched controls (1.78, 8.38) which are comparable to those reported in this thesis.

in a spectral contract of the second second second

. . -

To date, no prospective cohort study exist in which fracture risk has been determined by pQCT. Numerous studies have shown that radial BMD, as measured by SPA, has predictive value for hip (1.192,8.3,8.39,8.40) and vertebral (6.5, 6.10, 8.3)fractures. One standard deviation decrease in radial BMD results in a 1.6-2.6 increased risk o£ hip fracture (1.192,8.3,8.11), and a 1.3-2.4 increased risk of vertebral fracture (6.5, 6.10, 8.3, 8.11). It is however generally accepted that fracture risk is best determined by a site specific measurement (1.192,8.11,8.41). If this applies to the radius, then radial pQCT measurements should be useful in predicting distal forearm fracture, although this was not formallv assessed as part of the work for this thesis. The discriminatory power of pQCT trabecular and total BMD measurements in vertebral fracture suggests that pQCT may also be of value in the prediction of vertebral fracture. pQCT may be of particular value when lumbar spondylosis is evident, when the discriminatory power of spinal DXA is diminished (8.52). The power of pQCT to discriminate hip fracture was less than that of hip DXA and calcaneal ultrasound, suggesting that pQCT would be of less value in the prediction of hip fracture.

167

80. Ber seden om en en er befordet storeten of hårer er en her er om en en en en en en er en er en er en er en er e Prospective studies are now required to determine the clinical usefulness of pQCT in fracture prediction.

The fact that BUA was a poor discriminator of hip fracture was surprising and difficult to explain, as it is contrary to the findings of others (8.12-8.15), who found differences between groups of a similar magnitude to that presented here. BUA has also been shown to predict future hip fracture (8.16-8.19), and correlate with the failure load of human cadaveric femurs during simulation of a fall onto the trochanter (8.42). The finding that ultrasound velocity discriminates hip fracture from non-fracture populations as well as hip BMD, inspite of smaller percentage differences, is in keeping with previous work (8.12,8.14). When the differences in mean values and the range of absolute values are taken into account, and fracture and non-fracture groups are compared using z-scores, differences in DXA femoral neck BMD [-1.04 (0.86)] and VOS [-0.91(4.68)] are not dissimilar. The power of ultrasound velocity was superior to all pQCT measurements in. the discrimination of hip fractures, although this reached statistical significance only against pQCT trabecular BMD. The finding that ultrasound attenuation was no better than pQCT, and significantly poorer than DXA hip and ultrasound velocity was surprising, and contrary to previous findings comparing DXA and BUA. One possible explanation is the interval between fracture and scanning in the work presented in this thesis (mean: three years), as the fracture and subsequent immobility, could have resulted in greater bone loss at the hip than at the calcaneus. A recent, large, prospective study showed that calcaneal ultrasound attenuation and velocity predicted hip fracture as well as, and independently of, DXA femoral neck EMD in 7575 (5662 underwent ultrasound assessment) community dwelling, elderly (at least 75 years of age) women (8.19). Of the ultrasonic measurements, BUA was found to be the most

168

<u>en el ser en ser de la compaña de la serie de la compaña de la serie de la serie de la serie de la serie de la</u>

independent of femoral neck BMD, yet the most predictive of hip fracture. Additionally, for women in whom both BUA and femoral neck BMD was low (less than the median), the fracture risk was much greater than in those with only one low measurement. This suggests that fracture risk is best determined by measuring both femoral neck BMD and calcaneal BUA.

There are several confounding factors in the studies presented in this thesis. In the vertebral fracture study, the groups were not entirely comparable, with the fracture group being older, shorter and further beyond the menopause than the nonfracture group. However, correcting for these differences did not greatly change the highly significant differences for most BMD measurements between groups. Also, as discussed in chapter 7, age related rates of bone loss are greater at the spine and hip compared to the radius. Therefore any age difference between groups would affect DXA more than pQCT measurements, and as such should not detract from the finding that pQCT trabecular BMD discriminates best between vertebral fracture and non-fracture groups. An additional problem was performing axial measurements on two DXA scanners manufactured by different companies, which is less than ideal. However, correction equations have been extensively validated to allow comparison of data acquired from these different scanners (8.23). Additionally, there was no quantification of lumbar spondylosis on the spinal x-rays. If significant osteoarthritic changes were present, this would have diminished the discriminatory power of the DXA lumbar spine measurement.

In the hip fracture study, the control group is much smaller than the fracture group, whilst the reverse is the ideal. Unfortunately it proved impossible to recruit more non-fracture controls. Also, only a subgroup of the population originally recruited underwent pQCT as discussed in chapter 8.2.2. There

169

<u>na serie de la companya de la compa</u>

was no preselection of this subgroup, as pQCT was performed whenever personnel trained in pQCT were available, and this is unlikely to have biased the subgroup which formed the populations presented here. The differences in bone mass measurements between hip fracture and non fracture groups for hip DXA and ultrasound measurements are comparable to those previously published (8.12,8.13,8.15), which supports the concept that both groups were representative.

(1) S. C. M. Mark Math. Astrophys. 1998 (1997) 181 (1997).

A statistic statistic sector statistical

100.63

ł

a sea en s

Finally, the ability of pQCT to predict hip, vertebral and distal forearm fractures must now be determined by prospective studies.

and the second second

corresponding	
with the	
ure groups	
-F) fract	
ral (VERT	
ud verteb	. squ
2 (A-4IH)	ntrol gro
f the hip	ERT-C) CO
nic data c	tebral (V
Demograpl) and ver
Table 8,1.	hip (HIP-C

	HIP-C (n=45)	HIP-F (n=165)	VERT-C (n=30)	VERT-F (n=39)
Age (yrs)	75.1 (7.7)	73.9 (9.1)	64.1 (7.1)	68.2 (6.9) ^a
Height (m)	1.565 (0.064)	1.557 (0.074)	1.569 (0.06)	1.531 (0.061) ^a
Weight (kg)	61.4 (9.6)	58.9 (11.4)	62.9 (10.1)	59.7 (10.2)
Years Postmenopause	30.4 (9.7)	29.0 (10.2)	15.3 (9.3)	21.8 (6.8) ^b
Interval between hip fracture	ı	3.0 (1.5)	ı	t
and study (yrs)			c	
NO. OF VERTEDIAL LIACTURES &	I	ł	Ð	(A-T) 2
	1000 - 1000 - 1000 - 1000 - 1000 - 1000 - 1000 - 1000 - 1000 - 1000 - 1000 - 1000 - 1000 - 1000 - 1000 - 1000 -			

(a):0.05>p>0.01; (b):0.01>p>0.001, for unpaired t-test versus controls. Otherwise not significant Values are mean (SD), except § which is median (range).

discriminate ve	ertebral fr	acture from	non-fractu	re popul	ations.	Comparison	against	mutually	exclusive
measurements ar	e also show	n.							
	Qtot	Qtrab	Qscort	Qaart	LS	EN	Б	НW	
AUC	0.782	0.853	0.715	0.572	0.847	0.822	0.787	0.784	
Qtrab	0.139								
Qscort	0.134	0.035							
Qcort	<0.002	<0.002	<0.002						
LS	0,211	0.976	0,035	<0.002					
FN	0.395	0.631	0.057	<0.002	0.582				
ĹŀĦ	0.842	0.294	0.159	<0.002	0.246	0.395			
H ^W	0.905	0.308	0.254	0.003	0.234	0.368	0.952		

8.2 Area (AUC) under the receiver operator curves (ROC) indicating the power of pQCT and DXA to

Table

Values are p-values for comparison of AUC of ROC.

Table 8.3 Area	(AUC) under	the receiv	ver operato	r curves	(ROC) indi	cating the	power of p	QCT, hip D	XA and
calcaneal ultras	ound to disc	riminate hi	p fracture	from non-f	racture po	opulations.	Comparison	against mu	tually
exclusive measu	rements are	also shown.							
	Qtot	Qtrab	Qscort	Qcort	FN	<u>[-1</u> [2 4	ME	BUA	VOS
AUC	0.693	0.648	0.698	0.675	0.796	0.751	0.738	0.663	0.783
Qtrab	0.201								
Qscort	0.841	0.271							
Qcort	0.653	0.689	0.39						
NA	0.028	<0.002	0.037	0.027					
ЪД	0.254	0.042	0.284	0.184	0.062				
МЛ	0.358	0.07	0.424	0.276	10.0	0.624			
BUA	0.589	0.787	0.542	0.857	0.007	0.082	0.144		
VOS	0.142	0.026	0.162	0.093	0.818	0.582	0.447	0,009	
Values are p-val	lues for com	parison of	AUC for RO	υ					



Fig 8.1: Differences in pQCT BMD measurements between vertebral fracture and non-fracture groups

Fig 8.2: Differences in DXA BMD measurements between vertebral fracture and non-fracture groups



p value is for covariate analysis controlling for age, height, weight and years postmenopause. Z score: mean (SD) standard deviation difference of the fracture group relative to the control group

Fig 8.3: Receiver operator curves (ROC) showing the power of DXA and pQCT to discriminate vertebral fracture and non-fracture populations.



Fig 8.4: Differences in pQCT BMD measurements between hip fracture and non-fracture groups



p value is for covariate analysis controlling for age, height and weight differences. Z score: mean (SD) standard deviation difference of the fracture group relative to the control group

Fig 8.6: Differences in DXA BMD measurements between hip fracture and non-fracture groups



p value is for covariate analysis controlling for age, height and weight differences. Z score: mean (SD) standard deviation difference of the fracture group relative to the control group





Fig 8.6: Differences in ultrasound measurements between hip fracture and non-fracture groups

Fig. 8.7: Receiver operator curves (ROC) showing the power of hip DXA, pQCT and calcaneal ultrasound to discriminate hip fracture and non-fracture populations.



CHANGES IN RADIAL POCT BMD MEASUREMENTS IN RESPONSE TO DRUG THERAPY: COMPARISON WITH MEASUREMENTS AT OTHER SKELETAL SITES.

9.1 Introduction

One of the most important functions identified for bone is monitoring of drug therapy densitometry (1.145,1.146,1.153,1.154,9.1). This is as equally applicable to drugs which have a beneficial effect upon bone mass and are used in the treatment of osteoporosis, as it is to those which have an adverse effect upon bone mass, such as corticosteroids. Both hormone replacement therapy (HRT) and cyclical etidronate with calcium are used in the treatment of osteoporosis. HRT has been shown to reduce fracture rates at the hip, spine and radius in postmenopausal women (1.198,1.199,9.2-9.5), increase axial (9.6-9.13) and maintain radial (9.8,9.11,9.14) BMD. Cyclical etidronate and calcium increase spinal BMD and is of proven value in the treatment of established postmenopausal osteoporotic vertebral fracture by reducing vertebral fracture (9.15, 9.16, 9.17)and increasing rates spinal BMD (9.7,9.13,9.15,9.16,9.17). The desired interval between repeat measurements for monitoring response is largely determined by the precision of the scanner and the expected rate and magnitude of change. As the precision of pQCT is comparable to DXA lumbar spine and femoral trochanter measurements, and better than femoral neck and Ward's area measurements (see chapters 2 and 3), the response to these treatments may be better monitored at the radius by following changes in trabecular BMD. This is especially attractive as both HRT and etidronate preferentially affect trabecular bone which has an increased metabolic rate. As DXA measures a composite of trabecular and cortical bone unlike pQCT which allows direct quantification of trabecular BMD, DXA may be less sensitive to

181

nana 2014 - Antonio Jampin, Angelanda angelana na sakara bana bana angelana angelana angelana angelana angelan

smaller changes in trabecular bone than pQCT. Whilst spinal QCT is also a technique which allows measurement of trabecular bone independently, its use, especially when repeat measures are necessary, is limited by radiation considerations (see chapter 1). The use of pQCT for monitoring purposes would however be dependent upon drug induced change at the radius being of a similar timing and magnitude to that at other skeletal sites. At present, BMD assessment every 2 years is recommended (1,145). A hypothesis can be raised pQCT will allow earlier detection of responders, and equally importantly, nonresponders to HRT and etidronate. The work in this chapter addresses this hypothesis, the issue of timing of bone mass changes at different skeletal sites in response to HRT and etidronate during the first year of treatment, and whether changes in radial trabecular BMD measured by pQCT can predict axial changes measured by DXA.

.

9.2 Study Populations and Bone Mass Measurements

11 postmenopausal women (defined as no menses for at least 2 years) were treated by their general practitioners with HRT preparations (Prempak-C, n=4; Livial and Estracombi n=2; Trisequens, Estraderm and Cycloprognova each n=1) for the prevention of osteoporosis (HRT Gp). 6 of these women were identified from the ongoing osteoporosis screening programme which has been described previously (Chapter 5: 5.2-point 2). They either had lumbar spine or femoral neck BMD in the lowest quartile of age matched controls and were targeted for intervention with HRT assuming no contra-indications, a cut-off point which has been proposed previously (1.88). The remaining women were referred directly by their general practitioner's for a DXA scan and were also found to have low bone mass for which HRT was recommended. All women were either osteopenic (Tscore \leq -1; n= 1) or osteoporotic (T-score \leq -2.5; n= 10) as defined by the WHO (6.4), at either the femoral neck or lumbar

182

l'ant die setui e s

spine. None of these women had suffered an osteoporotic fracture and none had vertebral fractures on lateral thoracolumbar spine x-ray.

10 older postmenopausal women were referred by their GP's to a specialised bone clinic because of symptomatic osteoporotic vertebral fractures for which cyclical etidronate and calcium supplements was prescribed (ETD gp). All women were osteoporotic (T-scores \leq -2.5 at either lumbar spine or femoral neck) with atraumatic vertebral osteoporotic fractures.

None of the women in either group had other conditions or were taking other medications which could affect bone mass or metabolism. Baseline pQCT and DXA of hip and spine measurements were made before treatment was started, and then 4, 8 and 12 months thereafter. All measurements were done on the same day except the baseline visit, where DXA preceded pQCT by no more than 8 weeks in some women - DXA was repeated if a longer delay was encountered. For the follow-up pQCT measurements, the "trending" function, which is discussed in chapter 3, was utilised. The cut-off value used for the difference in voxel numbers between baseline and follow-up scans was set at 30, thus ensuring accurate re-positioning of the scanner in followup scans and minimising the influence of change in scan site on repeat pQCT BMD measurements.

The co-efficients of variation (CV) for pQCT (see chapter 3) are generally poorer in postmenopausal women with vertebral osteoporotic fractures compared to young normal women (Qtot: 1.73% vs 1.24%, Qtrab: 2.96% vs 1.33%, Qscort: 1.57% vs 1.58%, Qcort: 5.18% vs 1.88%). The precision of axial DXA measurements was determined in 6 postmenopausal women [median (range) age: 68 (60 -71) years, and years postmenopause: 21 (13 - 25) years] with vertebral fractures on lateral thoracolumbar x-ray [median

183

and the second second

(range) number of fractures: 2.5 (1-3)], by performing repeat hip and spine measurements on the same day with repositioning of the subject between scans. These women were different to those studied to determine the precision of pQCT in the a vertebral fracture group, and those studied in the etidronate treatment group. The CV's in the vertebral fracture group compared to young normals were: LS: 2.32% vs 0.9%, FN: 1.3% vs 2.8%, FT: 1.42% vs 1.3%, FW: 2.16% vs 4.8%. Fractured vertebral bodies within the DXA lumbar spine region of interest (L2-L4) were excluded from all relevant analyses.

والمراجع والمعاد فالمناب المراجع والمراجع والمراجع والمراجع والمراجع

9.3 Statistical Analyses

Results for continuous variables are expressed as median with ranges. Baseline lumbar spine and femoral neck BMD results are also expressed as T-scores and Z-scores (number of standard deviations above or below the mean relative to young, and age matched normal controls respectively). Between group comparison at baseline are performed using the Mann-Whitney U test. Within group percentage changes in bone mass measurements from baseline are compared using the Wilcoxon matched pairs test and one factor repeat measurement analysis. The relationships between annual rates of changes in BMD at different sites were examined using linear regression and correlation co-efficients.

9.4 Results

Baseline clinical and bone mass details of both groups are shown in Table 9.1. Although 4 women had undergone hysterectomy (hyst) previously [HRT gp: n=1(age 63,hyst 20 yrs previously)], [ETD gp: n=1(age 71,hyst 16 yrs previously), n=2(aged 60 & 73, hyst 30 yrs previously)], none had FSH, LH or oestradiol levels done to confirm their assumed postmenopausal status. The HRT group were younger and fewer years postmenopausal than the ETD group. Consequently all baseline BMD measurements were greater

ومراجع والمستعدين المرجع

in the HRT group with the exception of LS.

the second se

There were no significant changes in median height or weight values during the 1 year follow-up period in either HRT or ETD groups (see table 9.2). Percentage BMD changes for the pQCT radial and DXA hip and spine BMD measurements are shown in tables 9.3 and 9.4 respectively, and depicted in figures 9.1 and 9.2. There were no overall significant changes in any pQCT BMD measurements for either group. There was however a trend towards a fall in BMD at month 8 in the HRT group, which reached significance for Qtot and Qscort, with some recovery by month 12. Significant overall increase was found at LS in both groups, which was statistically significant within 4 months, and further increased after 12 months of treatment. Although there was a trend for all hip BMD measurements to increase, overall significance was only achieved for FN in the HRT group. The results remained unchanged after correction for changes in height and weight. There were no significant relationships between the percentage changes in any of the pQCT BMD measurements and DXA hip or spine BMD changes with either treatment. Comparisons of the percentage changes in pOCT measurements with LS BMD after 12 months of treatment with HRT (r = 0.19, -0.35, 0.26 and 0.19 for Qtot, Qtrab, Qscort and)Qcort respectively; p = NS for all) and etidronate (r = 0.14, 0.35, 0.01 and -0.01 for Qtot, Qtrab, Qscort and Qcort respectively; p = NS for all) were typical of comparisons.

9.5 Discussion

These results shows that changes in BMD induced by both HRT and etidronate during one year of treatment varies between skeletal sites. The most significant and greatest changes occurred at the lumbar spine for both treatments, although there was considerable inter-individual variation. Also, for several

185

a server there are a server have been been as a server the

women in either group, lumbar spine BMD did not increase during the first year of treatment. Median increases of +6.6% and of treatment +5.3% after one year with HRT (9.6, 9.9, 9.10, 9.11, 9.18)etidronate and (9.15, 9.16)respectively are in keeping with previous work. The work presented here suggests that statistically significant changes were detectable at the lumbar spine after only 4 months of either etidronate. treatment with HRT \mathbf{or} Present recommendations are that BMD measurements should not be repeated until a patient has been on therapy for 1-2 years (1.145). This is based upon a change 2.8 times that of the coefficient of variation (CV) being necessary before it can be regarded as real and not due to measurement error (1.88,9.1). The CV for DXA lumbar spine measurements in our centre is 0.9% in young healthy women, but 2.32% in older osteoporotic women. The median change in lumbar spine BMD after 4 months of treatment with HRT was +3.3% which could only be accepted as significant if the precision value for young normal women was applied (ie CV = 0.9% with significant change greater than \pm 2.52%). If this figure was applied, then a repeat lumbar spine DXA scan as early as 4 months may indicate response to therapy. However, although the HRT group did not suffer from vertebral fractures, and major spondylosis was not a problem, it is unlikely that the precision value would be as low as 0.9%, due to the lower BMD values which would result in a higher CV. If the higher CV figure was applied, then a follow up scan after 1 year would be the earliest possible to detect significant. change. In the etidronate group, the higher CV for DXA lumbar spine scans would have to be applied (CV = 2.32%, with significant change greater than \pm 6.4%). Consequently, a repeat scan no earlier than 1 year would be indicated. This implies that the optimum length of time between scans when assessing the response to therapy is dependent upon both the age of the patient (which can affect the precision of the measurement),

consider and a new protocol

化化化物 化过去式分词 法法律法 使过去分词 网络小花花 法法法保证证券

and the expected magnitude and timing of the response to treatment.

Contraction of the All Contraction of

1970 - N. M. M. M.

It is surprising that the baseline lumbar spine BMD of the ETD group was greater than that of our HRT group bearing in mind they were older and more years postmenopausal. This could be due to age related spinal osteoarthrosis (1.171-1.177) and aortic calcification (1.168,1.169,1.170) elevating spinal BMD results in the ETD group. This should not influence the rate of response at the lumbar spine, although a state of low bone turnover has recently been observed in patients with lumbar spondylosis and osteoarthritis (1.174,1.178,1.179,9.19). Also, we cannot exclude a spurious rise in BMD from further fractures occurring within the area of interest (L2-4) during the first year of treatment, as lumbar spine X-rays were not repeated. This attempt to minimise radiation was an exposure, particularly as the purpose of this work was not to examine the effect of treatment upon fracture rates.

Although there was a trend towards an increase in all hip measurements for both groups, this only achieved overall significance for femoral neck BMD in the HRT group. Similar changes in proximal femur BMD have been documented previously with HRT (9.6,9.10,9.11,9.18) and etidronate (9.20,9.21), and tend to be lower than those found at the lumbar spine, presumably due to a smaller proportion of trabecular bone at the former site. Bearing in mind the poorer precision of DXA hip measurements, they would be of less value than spinal measurements for detecting early response to HRT and etidronate. If however, certain situations prevailed, such as heavy overlying aortic calcification, severe spondylosis or multiple vertebral fractures within the spinal region of interest, then hip measurements may be the better choice for monitoring response, with an appropriately longer interval

187

between scans.

Previous work has shown that radial BMD, as measured by single photon absorptiometry (SPA) is maintained but not increased HRT (9.8, 9.11, 9.14) and significantly with etidronate (9.7.9.15). Theoretically, by quantifying trabecular BMD independently, pQCT should be more sensitive than SPA in detecting change at appendicular sites, and may mirror or even antedate axial changes measured by DXA. Although radial BMD was maintained, there was no significant increase, even in trabecular BMD in either treatment group. This, and the fact that there were no meaningful relationships between percentage changes at the radius and spine or hip, suggests that the response to treatment at the radius, even within the trabecular component, is different from that at the lumbar spine. It is possible that trabecular BMD may change after more than 1 year of treatment, but previous studies using SPA have shown no significant increase in radial BMD after treatment with HRT for 2 years (9.22), and etidronate for 3 years (9.15).

Early demonstration of response to treatment is important because it may help improve patient compliance (1.201,1.202,9.1), and allow early detection of non-responders which have been found previously with both etidronate (9.20,9.23) and HRT (9.10,9.18,9.22). For such non-responders, compliance with therapy could be checked, the importance of treatment emphasised, or the treatment changed if necessary. Also, proof of response to therapy is important as there is now some evidence that the anti-fracture efficacy of treatment is determined by both a reduction in bone turnover and an increase in BMD (9.24). The work presented in this thesis suggests that a repeat measurement after one year would evaluate response to treatment with HRT and etidronate in the majority of cases. When changes smaller than 2.8x the CV are recorded, such

188

changes could also be interpreted as showing an improvement if there was a consistent upward trend, although this would have to be judged in the context of a given individuals treatment (9.1). In clinical practice, the interval between follow up scans is likely to be determined largely by individual clinical situations and resources. In most instances, due to the differential effect of HRT and etidronate on axial and appendicular bone, response to treatment would be best determined by a spinal BMD measurement.

.

.....

See a second water to a second state of the second state of the second state of the second state of the second

	HRT gp (n=11)	ETD gp (n=10)
Age	53 (46-63)	71 (60-79)
Years Postmend	opause § 4 (2-10)	23 (14-30)
No. of fractu	res -	2 (1-4)
pQCT (g/cm³)		
Qtot	340.0(261.6-389.1)	279.6(198.3-397.8)
Qtrab	147.8(102.1-200.4)	125.1(87.8-225.7)
Qscort	476.3(383.9-562.3)	376.6(288.6-538.3)
Qcort	523.0(466.7-627.0)	463.3(399.1-565.3)
DXA (g/cm²)		
\mathbf{LS}	0.755(0.632-0.959)	0.783(0.526-0.934)
T-score	-3.17(-4.231.41)	-2.93(-5.021.63)
Z-score	-1.46(-1.840.43)	-0.70 (-2.55 - +0.36)
FN	0.723(0.540-0.905)	0.606(0.471-0.858)
Tscore	-2.06(-3.580.54)	-3.03 (-4.160.94)
Z-score	-1.01(-1.88 - +0.69)	-1.26 (-2.09 - +0.89)
FT	0.577(0.441-0.813)	0.497(0.276-0.619)
FW	0.533(0.373-0.760)	0.387(0.304-0.554)

Table 9.1 Baseline clinical, pQCT and DXA data for hormone replacement therapy (HRT) and etidronate (ETD) groups.

· · · ·

a star where set where s

. . . .

the second second second second second second

Values are median (range) unless otherwise stated.

§ for non-hysterectomized women.

190

and the second second
8 and 12 in hormone replacement therapy (HRT), and etidronate Table 9.2. Height and weight at months 0, 4,

(ETD) groups.

	· • • • • • • • • • • • • • • • • • • •				
	Month 0	Montīn 4	Month 8	Month 12	Q
Height	(m)				-
HRT	1.615m(1.485-1.708)	1.614(1.472-1.702)	1.61 (1.476-1.703) 1	L.615(1.471-1.705)	NS
ETD	1.547(1.451-1.621)	1.551(1.453-1.602)	1.541(1.452-1.621) 1	1.549(1.448-1.601)	NS
Weight	(kg)				
HRT	66 (51-89.5)	66 (52-91.5)	65.5 (51-89)	64 (53-89)	NS
ETD	61.5 (44.5-83.5)	60 (45-78)	62.8 (45.5-82.5)	61.3 (47.5-82)	SN
					-
Values	are median (range)				
p-valu(e is for one factor repeat	measurement analysis.			

191

5 X 4

etidro	a.s.re nate (En	rcentage on D gp) group	auges III pyc. 18, compared 1	I at 4, 8 aut 12 months 1. with baseline.	п потшоле тертасешени	γαρι	(d6 1911)
		Month	4 ¹	∞	12	Overall-	Sig
pQCT							
Qtot	HRT	-0.7 (-7.0	1 - +10.5	-2.0 (-4.8 - +1.7) ^{a.0.}	-1.1 (-9.8 - +4.9)	NS	
	ETTD	-0.4 (-2.7	, - +7.4́)	+0.5 (-2.8 - +8.6)	+0.2 (-3.7 - +7.8)	NS	
Qtrab	HRT	+0.3 {-14.	0 - +4.2)	-1.3 (-18.2 - +3.9)	-0.7 (-9.3 - +2.8)	NS	
	ETD	+0.9 (-8.5	: - +14.2)	-0.5 (-16.4 - +8.2)	-1.0 (+12.9 - +13.1) NS	
Qscort	HRT	-1.0 (-6.7	- +14.5)	-1.8 (-5.6 - +3.3) ^{0.04}	-1.1 (-9.9 - +7.1)	SN	
	ETD	-0.8 (-4.1	- +11.5)	+0.8 (-5.2 - +11.3)	+0.6(-4.1 - +9.8)	SN	
Qcort	HRT	-1.2 (-6.7	- +4.6)	-2.4 (-7.8 - +2.8)	-2.2 (-4.9 - +5.5)	SN	
	$\mathbf{E}\mathbf{T}\mathbf{D}$	-0.3 (-3.0	- +11.5)	+0.4 (-5.4 - +11.3)	-0.2 (-6.0 - +9.2)	SN	
Values	are med	ian (range)	p-vzlua for Wilcoxo	n paired test versus mosth o			

and

Overall significance is for one factor repeat measurement analysis.

hormone	
년 - 년	
months	
7	
and	line.
ω	ase]
, 4	ğ
ц а	wit]
easurements	, compared ,
BMD m	groups
did	(db
and	ETD
spine	onate (
n DXA	etidro
-11 -11	nd
changes	Гgp) а
e E	(HR.
Percentaç	therapy
4.	ent
ດັ	cem
Table	repla

.

DXA					
LS	HRT	+3.3 (-4.8 - +9.8) ^{0.02}	+4.3 (-2.7 - +8.8) ^{3.01}	+6.6 (-3.2 ~ +13.1) ^{a.c1}	< 0.001
	ETD	+3.8 (-1.5 - +10.2) ^{0.01}	$+5.0(-2.8 - +8.4)^{0.01}$	$+5.2 (-4.3 - +12.8)^{3.02}$	0.013
FN	HRT	+1.6 (-4.1 - +5.6)	+1.2 (-3.2 - +7.9)	+3.1 (-5.7 - +7.7)	0.026
	ETD	-0.6 (-5.1 - +13.1)	+1.0 (-4.0 - +6.8)	+2.1 (-3.7 - +7.6)	NS
ЕŢ	HRT	+2.4 (-4.9 - +12.7)	+1.2 (-6.1 - +13.7)	+2.3 (-8.8 - +19.5) ^{3.026}	NS
	ETD	+4.2 (-17.7 - +29.3)	+4.2 (-8.0 - +33.3)	$+5.3 (+0.5 - +35.2)^{3.005}$	NS
FW	HRT	+4.3 (-7.9 - +22.4)	+3.1 (-15.1 - +22.9)	+3.8 (-1.8 - +20.8) ^{5.01}	SN
	ETD	+5.9(-6.7 - +45.1)	+7.8 (-14.5 - +50.3)	+5.4(-8.1 - +44.2)	NS

Values are median (range) p-value for Wilcoxon paired test verbus month D

Overall significance is for one factor repeat measurement analysis.

193



Fig 9.1: Percentage changes in pQCT and DXA BMD measurements with HRT



Fig 9.2: Percentage changes in pQCT and DXA BMD measurements with etidronate

THE EFFECT OF LONGTERM WARFARIN THERAPY ON RADIAL AND AXIAL BMD.

÷ 2

10.1 Introduction

It is well recognised that heparin can cause bone loss (10.1). Warfarin is known to affect the development of foetal bone with stippled epiphyses and hypoplasia of the nose and extremities being observed in the off-spring of mothers treated during prequancy (10.2,10.3). The effect of warfarin on adult bone is less well documented, but there is a suggestion that courmarins do have an adverse effect (10.4,10.5). In these previous studies reduced bone mass was found, but assessment was limited to the radius using SPA, and the more clinically important sites of hip and spine were not examined. Warfarin is thought to affect bone metabolism through its effect on osteocalcin, the non-collagenous bone matrix protein which is synthesised by osteoblasts and considered to be a marker of bone formation. Osteocalcin contains gamma-carboxyglutamic acid (gla), which is derived from glutamate through enzymic carboxylation (10.6), a process which requires vitamin K as a co-factor. Warfarin, by antagonising this vitamin K dependent process, results inincreased decarboxylated osteocalcin, which is not incorporated into the bone matrix, but released into the circulation (10.7).

As warfarin is a commonly prescribed drug, any adverse effect on bone deserves further investigation. As previously stated, metabolic changes affect trabecular bone more rapidly than they do cortical bone. Therefore any effect of warfarin should be greatest within the trabecular component, and skeletal sites rich in trabecular bone. pQCT should therefore be valuable in assessing its effect on bone. The influence of warfarin upon BMD at the radius, was compared to that at the hip and spine in

196

L State Contraction

<u>. X. e. e.</u>

a group of men taking longterm warfarin.

10.2 Study Population and Bone Mass Measurements

A state specified in protocol to the state

40 men taking longterm warfarin were recruited from the hospital anticoagulant clinic (WARFARIN group). The indications for anticoagulation were as follows: prosthetic heart valve replacement (HVR; n=16), atrial fibrillation (AF; n=13), bypass surgery for peripheral vascular disease (PVD; n=7), and recurrent venous thrombo-embolism (VTE; n=4). The degree of anticoagulation was determined by averaging the INR values at the anticoagulant clinic visits before and after the day of study. Controls were obtained from a search of hospital discharge summaries, clinic lists and surgical registers (CONTROL group). Each individual in the WARFARIN group was matched to a control for age (within 5 years). They were also matched for underlying disease so that individuals in the WARFARIN group with AF and PVD were matched to controls with the same condition but not requiring warfarin. Likewise those taking warfarin for HVR were matched to controls with HVR not requiring warfarin, although a small number of controls had undergone open heart coronary artery by-pass grafting rather than HVR. Those taking warfarin because of VTE were matched to controls who had undergone lower limb varicose vein surgery. All men were Caucasian and ambulatory. Individuals with underlying conditions or who were taking drugs (with the exception of thiazide diuretics) known to influence bone metabolism or bone mass were excluded. The controls for individuals in the WARFARIN group taking thiazide diuretics (n=4) were also matched for thiazide diuretic use. Thus the WARFARIN and CONTROL groups were stringently matched for factors which could influence BMD other than warfarin therapy.

All individuals attended for assessment on a single visit within a three week period. At this visit all medications,

197

including warfarin dose, were checked. Height, weight, smoking habit and alcohol consumption were recorded. A questionnaire about physical activity was completed which gave a composite score with a range of 5(most active) to 20 (least active). All men had BMD measurements at the radius by pQCT, and hip and spine by DXA.

은 사람이

1.11

10.3 Statistical Analyses

and the state of the state

get all construction of

Results for continuous variables were expressed as the mean with standard deviation if the data were normally distributed, otherwise as the median with ranges. Between group comparison performed using the unpaired t-test. Within was group comparison of BMD data was performed using the Kruskal-Wallis examine the influence indication test to of for anticoagulation. In the WARFARIN group, the relationships between bone mass measurements and duration and degree of anticoagulation were examined using univariate regression analyses.

10.4 Results

Anthropometric, lifestyle and warfarin related data are shown in table 10.1. As might be expected from the matching process, there were only slight, and insignificant differences between the groups for anthropometric and lifestyle variables. One man had received warfarin therapy for 4 months, the remainder for more than 12 months. LS $(1.061 \pm 0.182 \text{ vs } 0.95 \pm 0.139 \text{ g/cm}^2,$ p=0.003) and Qtrab $(219.8 \pm 37.1 \text{ vs } 200 \pm 39.8 \text{ g/cm}^3, p=0.024)$ were significantly lower in the WARFARIN group. Whilst the remaining pQCT (Qtot: $398.7 \pm 64.2 \text{ vs } 376.7 \pm 61.3 \text{ g/cm}^3,$ Qscort: $533.3 \pm 86.5 \text{ vs } 509.8 \pm 77.3 \text{ g/cm}^3,$ and Qcort: $529.6 \pm$ $70 \text{ vs } 527.6 \pm 61.3 \text{ g/cm}^3)$ and DXA hip (FN: $0.817 \pm 0.122 \text{ vs}$ $0.801 \pm 0.142 \text{ g/cm}^2,$ FT: $0.794 \pm 0.147 \text{ vs } 0.748 \pm 0.13 \text{ g/cm}^2,$ and FW $0.587 \pm 0.106 \text{ vs } 0.571 \pm 0.139 \text{ g/cm}^2)$ were also lower in the WARFARIN group, the differences failed to reach statistical

significance. Figures 10.1 and 10.2 show the pQCT and DXA BMD data for both groups respectively. There were mean percentage differences of -10.4% and -9.0% for LS and Otrab respectively. The percentage differences for all other measurements were lower: -5.5%, -4.4% and -0.4% for Qtot, Qscort and Ocort respectively; and -2%, -5.8% and -2.9% for FN, FT and FW respectively. Controlling for height, age and weight differences between groups did not significantly change the findings. There were no differences in BMD measurements within the subgroups of the WARFARIN group based upon indication for warfarinisation. Linear regression analyses of the WARFARIN group showed that there were no significant relationships between any of the BMD values and dose or duration of warfarin therapy, or with the degree of anticoagulation as indicated by the INR. These relationships with Qtrab and LS BMD are shown in figures 10.3 and 10.4 respectively. Unexpectedly, the slopes of the regression lines of duration of warfarin therapy versus LS and Qtrab BMD were both positive, but failed to reach statistical significance.

10.5 Discussion

These results show that males treated with longterm warfarin have lower BMD at the radius, spine and hip than age matched male counterparts. The greatest difference was found at the lumbar spine, where trabecular bone accounts for about 65% of the vertebral body bone mass (1.20). Trabecular BMD at the radius was affected to a similar extent, whilst there was very little difference in radial cortical BMD. This suggests that trabecular bone is preferentially affected by warfarin, no doubt due to its increased metabolic rate compared to cortical bone. This is in keeping with the effects of other drugs which have an adverse effect upon bone such as corticosteroids (10.8), and those which have a beneficial effect, such as HRT and etidronate (see chapter 9). Bearing in mind that the DXA

1**9**9

lumbar spine measurement is a composite measure of both trabecular and cortical bone, it is surprising that the difference at the spine was greater than that for the purely trabecular measurement at the radius. Differences in trabecular bone at the lumbar spine, which could be measured by QCT, would probably be greater than the difference detected by DXA in this thesis, and therefore also greater than that found at the radius. This suggest that bone at the radius is affected less than that at the spine. At the hip, the greatest difference was found for the femoral trochanter site which has a greater proportion of trabecular bone than the femoral neck (1.20). Whilst the femoral Ward's area is supposed to represent the most trabecular rich area within the proximal femur, poorer precision would explain why the difference at this site is less than other hip sites.

and the state of the

10.000 (10.00

12

There were no significant relationships between duration or dose of warfarin, or degree of anticoagulation with BMD measurements at any site. The relationship between duration of warfarin therapy and lumbar spine and radial trabecular BMD appears to be a positive. However, the majority of men had been on warfarin for less than 160 months (range 4-302 months), with only 6 being on treatment for longer. It is possible that these 6 individuals are not representative of this population and may skewed the have data and resultant regression slope. Alternatively, the bone losing effect of warfarin may occur early in t:he course o£ treatment, as it does with corticosteroids (10.8). There are differences between the vitamin K antagonism on carboxylation in bone and liver (10.9) and it is possible that the antagonistic effect on bone occurs at doses lower than is necessary for anticoagulation. This may explain the lack of relationship between dose and INR with BMD measurements. The issues relating to duration, dose and INR would be better addressed in a prospective study with larger

200

numbers of subjects.

Men were chosen for the study group because the matching of individuals in the warfarin group with controls (for age, thiazide diuretic use and underlying disease) was considered to be very important. This would have been much more difficult had women been studied, as it would have been necessary to match for menopausal status and duration since the menopause as well. Matching individuals for underlying condition proved to be imperfect, as is illustrated by men taking warfarin for venous thromboembolism being matched to controls who had undergone varicose vein surgery. Such difficulties are probably of minor importance in this work, and are inevitable in cross-sectional designs.

The clinical importance of these findings are unclear at the present time, as there is no fracture incidence data available to date. However, a 10% reduction in BMD is equivalent to about 10 years of age related bone loss (8.40), and around 50% of the difference between postmenopausal and premenopausal women (10.10). It is also comparable to a fall in BMD of almost 1SD within the general population (10.11), which is associated with up to а three fold increase in fracture risk (6.10,10.12,10.13). In view of the increasing indications for anticoagulation, such as non-rheumatic atrial fibrillation and poor left ventricular function, perhaps the adverse effect on bone should also be considered when deciding to anticoagulate individuals, particularly those already at increased risk of osteoporosis, such as postmenopausal women.

201

a la creación de la tablicación de secur

	WARFARIN GD	CONTROP 3D	5 4	
Age (years)	64.2 (7.1)	64.9 (6.7)	NS	
Height (m)	1.702 (0.062)	1.703 (0.081)	NS	
Weight (kg)	76.5 (12.4)	77.1 (11.7)	NS	
Smoking (pack-years) §	26.5 (0-108)	23 (0-130)	SN	
Alcohol (units/week) §	4 (0-30)	2.5 (0-28)	SN	
Physical Activity Index	9.1 (2.9)	8.4 (2.7)	NS	
Warfarin dose (mg)	4.9 (1.9)	E		
INR	2.7 (0.7)	·		
Warfarin Duration (months) §	40.5 (4-302)	·		

Table 10.1 Anthropometric, lifestyle and warfarin related data for the WARFARIN and CONTROL groups.

p-value is for the unpaired t-test, or the Mann-Whitney U test when the median (range) is expressed.



Fig 10.1: pQCT radial BMD measurements for WARFARIN and CONTROL groups



Fig 10.2: DXA hip and spine BMD measurements for WARFARIN and CONTROL groups





Fig 10.4: Relationships between LS BMD and warfarin duration, dose and INR





THE EFFECTS OF DISEASE AND CORTICOSTEROIDS ON APPENDICULAR BONE MASS IN POSTMENOPAUSAL WOMEN WITH RHEUMATOID ARTHRITIS: COMPARISON WITH AXIAL MEASUREMENTS

11.1 Introduction.

Reduced bone mass is a well recognised feature of rheumatoid arthritis (RA), with many studies showing reductions at both axial and appendicular sites (11.1). The effect of oral corticosteroid therapy on bone mass in RA is contentious especially at appendicular sites (11.2,11.3). There is evidence to support the concept of generalised bone loss (11.4,11.5), but controversy exists about the distribution and degree of bone loss, and the relative contribution of immobility, low dose corticosteroids and the systemic effects of RA (11.3-11.8).

The pQCT measurement scan occurs at the 'ultra-distal radius in close proximity to the carpus. As the carpus is frequently involved in RA, and periarticular bone loss is a characteristic feature, pQCT seems an ideal technique to study periarticular bone changes related to RA. Additionally, trabecular bone is metabolically more active than cortical bone, and therefore may be more responsive to stimuli such as corticosteroid therapy (11.9), which is used to treat a proportion of patients with RA. Accordingly, by quantifying trabecular bone, pQCT may be able to differentiate the effects of RA disease activity and corticosteroid therapy on bone at the radial site.

The ankle and subtalar joints are also frequently involved in RA. One might therefore expect the os calcis, which is also a trabecular rich bone (1.21), to be affected in a similar manner to the distal radius by local inflammatory processes and

207

here and a second se

corticosteroid therapy. It would of interest to determine how calcaneal ultrasound measurements are affected by RA, and how such changes relate to those found with pQCT.

이 가장 그는 것은 것 같아. 이 것은 사람이 한 것을 위해 한 것

The relative importance of corticosteroids, disease parameters and life style on bone mass in postmenopausal women with RA, were examined using pQCT, and compared with ultrasound, and DXA of hip and spine.

11.2 Study Population, Clinical Assessments and Bone Mass Measurements

11.2.1 Study Population

75 ambulatory, Caucasian, postmenopausal women were studied. The control group comprised of 29 previously normal postmenopausal women, the recruitment of which is discussed in chapter 5 (5.2.point 3). The remaining 46 women had suffered from RA (as defined by the American College of Rheumatology) (11.10) for at least 6 months and were recruited from outpatient clinics and a casenote review. 25 of the RA women had been on longterm low dose prednisolone therapy (< 10mg/day) for at least 6 months (RA+P group). The remaining 21 had never received oral corticosteroids (RA group). Patients who had received intra-articular steroid were not excluded. Subjects with abnormal renal function (serum creatinine >140 μ mol/1) or other conditions known to affect bone metabolism were excluded individuals taking drugs as were such as HRT and bisphosphonates. Patients on thiazide diuretics were not excluded. Each individual had all assessments performed on the same day.

11.2.2 Clinical Data

Data collected included age, height, weight, years since

208

and an example alternation of the second state of the second state of the second state and the second state of

menopause, smoking history (cigarettes/day) and alcohol consumption (units/week). Dietary calcium intake (mg/day) was determined by a locally derived questionnaire, and level of physical activity by the Framingham questionnaire. All individuals underwent lateral thoracolumbar spine x-ray (T4-L5) and fractures were noted (defined as a reduction of 25% in anterior, mid or posterior vertebral body height).

1.1

RA patients completed a health assessment questionnaire (HAQ). Articular index (painful joint count only), mean grip strength of three measurements of right and left hands (mm Hg), visual analogue pain score (150mm horizontal line) and the time taken to walk 15 meters on level ground were recorded. X-rays of hands and wrists were scored using the Larsen method. Serum Creactive protein and rheumatoid factor were measured using standard laboratory methods.

11.2.3 Bone Density Measurements

pQCT was performed at the ultra-distal radius of the nondominant forearm. The effect of RA on in-vivo short-term precision is discussed in chapter 3. Briefly, precision measurements [expressed as the coefficient of variation (CV)] are poorer in the RA population compared to young normals (1.24% vs 1.6%, 1.33% vs 4%, 1.58% vs 1.84% and 1.88% vs 2.8% for Qtot, Qtrab, Qscort and Qcort respectively). pQCT was not possible in one RA subject due to bilateral wrist surgery. Ultrasound of the non-dominant os calcis was performed, but due to machine servicing, ultrasound data is missing from 7 controls and 1 RA subject. DXA hip and spine measurements were performed using a single scanner. Hip measurements were not possible on one RA+P subject due to bilateral hip replacements. Fractured vertebrae within the DXA spinal region of interest were excluded from the analysis. The effect of RA upon precision values of ultrasound and DXA hip and spine

209

1947 – Norden Jahren and State (1997) – State (1997)

measurements was not studied.

11.3 Statistical Analyses

Results for continuous variables were expressed as the mean with standard deviation if the data were normally distributed, otherwise as the median with ranges. Categorical data were examined using the Chi squared (χ^2) test. Variables common to all three groups were compared using the oneway analysis of (ANOVA) Kruskal-Wallis where variance or appropriate. Subsequent comparisons among groups were performed through the use of Scheffe's multiple comparison test. Variables relevant only to the RA groups were compared using the unpaired t-test or Mann-Whitney U test. Bone measurements across groups were examined having corrected for the influence of age, height, weight and years postmenopause by analysis of covariance with subsequent multiple comparison testing among groups. The independent relationships between corticosteroid use, RA. variables, life-style variables and bone mass measurements were examined using Pearson correlation co-efficients and stepwise multiple regression analysis.

11.4 Results

ere and the first of the

Detailed clinical data for the three groups are given in Table 11.1. There were no significant differences in age, years postmenopause, height, weight, smoking habit or dietary calcium among groups, although the RA+P group was slightly heavier and more years postmenopause than the others. A greater proportion of the control group consumed alcohol than either RA group, although the amount consumed was not significantly different (p=0.16; Kruskal-Wallis). The physical activity index was significantly higher in the control group compared to either RA group, but not significantly different between RA groups. Thiazide diuretic use was similar for the combined RA groups and control group. 17 (81%) of the RA group and 19 (76%) of the

210

and a second second

modifying taking disease agents RA+P group were (sulphasalazine, auranofin, hydroxychloroquine, penicillamine, myocrisin, methotrexate or azathioprine). The median disease duration was 4 years greater in the RA+P group (p=0.24, Mann-Whitney U test) and although there was a trend toward worse disease parameters in the RA+P group, significance was reached only for the HAQ. There were more vertebral fractures in the RA groups than controls, although the difference was not significant.

the second second second second second second

Bone mass measurements are shown in figures 11.1 - 11.3. There were significant reductions in both RA groups compared to the control group for all measurements except the lumbar spine and forearm subcortical and cortical BMD. Although Qscort was lower in both RA groups compared to the control group, this reached significance only for the RA+P group. pQCT BMD measurements were (control vs RA vs RA+P; overall ANOVA p value: Qtot: 340.6 +/- 67.1 vs 289 +/- 67 vs 285.9 +/- 89.7 g/cm³; p=0.016, Qtrab: 174.2 +/- 40 vs 111.8 +/- 52.6 vs 111 +/- 63 g/cm³; p=0.001, Qscort: 466.9 +/~ 92.8 vs 413.4 +/- 72.6 vs 393 +/- 102.3 g/cm³; p=0.009, Qcort: 495.2 +/- 89.3 vs 518.6 +/- 66.8 vs 498 +/- 97.2 g/cm³; p=NS). Ultrasound measurements were (control vs RA vs RA+P; overall ANOVA p value: BUA: 68.7 +/- 13 vs 49.5 +/-17 vs 44.6 +/- 20.5 dB/MHz; p=0.001, VOS: 1517 +/- 51.2 vs 1428 +/- 36.9 vs 1408 +/- 40.8 m/s; p=0.001). DXA BMD measurements were (control vs RA vs RA+P; overall ANOVA p value: LS: 0.89 +/- 0.16 vs 0.85 +/- 0.21 vs 0.82 +/- 0.18 g/cm²; p=NS, FN: 0.78 +/- 0.14 vs 0.68 +/- 0.15 vs 0.65 +/- 0.11 g/cm²; p=0.002, FT: 0.61 +/- 0.12 vs 0.55 +/- 0.15 vs 0.52 +/- 0.11 g/cm²; p=0.012, FW: 0.56 +/- 0.14 vs 0.45 +/- 0.14 vs 0.47 +/- 0.08 g/cm^2 ; p=0.007). Differences persisted after adjustment for the influence of age, height, weight and years postmenopause. There were no significant differences between the RA groups for any bone mass measurement. Combining the RA groups, the percentage

"我们,你们不可以一些我的问题,我们还是你的你们,你不是你们的你?""你?""你你们,你们不是你的问题。""你不是你?""你……""你……""你……""你……""

difference in bone mass compared to controls was -36.1%, -15.6%, -8.4% and +2.4% for forearm trabecular, total, subcortical and cortical measurements respectively, -31.7% and -6.6% for calcaneal BUA and VOS respectively, -15.4%, -13.1% and -17.9% for femoral neck, trochanter and Ward's area respectively, and -6.7% for the lumbar spine.

11 - 196A

Stepwise multiple linear regression analyses on the whole population are shown in table 11.2. Weight was positively correlated with lumbar spine and hip BMD, radial trabecular BMD and ultrasound measurements. Physical activity was positively correlated with lumbar spine and hip BMD. The independent variable RA (coded 0=no, 1=yes) was negatively correlated to radial total, trabecular and subcortical BMD, calcaneal ultrasound measurements and femoral neck and trochanter BMD. Years postmenopause was negatively correlated with radial total, subcortical and cortical BMD. Age was negatively correlated with femoral trochanter BMD. Introducing RA disease variables into the regression analyses for the combined RA group showed that the Larsen score (LARS) was negatively correlated with BUA [BUA = 86.25 - 1.04(LARS); R²=0.485, F Sig <0.0001], VOS [VOS = 1472.1 - 1.45(LARS); R²=0.185, F Sig = 0.0037], Qtot [Qtot = 347.2 - 1.852(LARS); R²=0.093, F Sig =0.0348] and Qtrab [Qtrab = 205.6 - 2.21(LARS) - 0.06(RhF), $R^2=0.503$, F Sig <0.0001], with rheumatoid factor titre (RhF) being negatively correlated with Qtrab. Table 11.3 shows the correlation coefficients of individual RA disease variables with pQCT and ultrasound measurements for the combined RA groups. The Larsen score for the non-dominant wrist and whole hand (wrist + MCP's + PIP's) was negatively correlated with Qtrab, BUA and VOS, whilst the Larsen score of the non-dominant wrist was positively correlated with Qcort. Rheumatoid factor titre was negatively correlated with Qtrab, BUA and VOS, whilst grip strength was positively correlated with Qtrab and BUA. The

significant linear regression relationships of Qtrab, BUA and VOS with RA disease variables are shown in figures 11.4 - 11.6 respectively.

11.5 Discussion

This study has demonstrated marked reductions in appendicular bone mass measurements in patients with RA using both pQCT and ultrasound. Some appendicular measurement changes were more marked than those noted at the femoral neck and lumbar spine, with the changes at the latter site being insignificant in this small study.

pQCT measurements were difficult in a small number of patients with RA, and the precision of pQCT is poorer in this population. There are several reasons for this, which are discussed in chapter 3 - namely a lower mean BMD value in RA patients, and difficulty visualising the radio-carpal joint during the scout scan in patients with more severe RA where the normal anatomical landmarks are distorted. However, unless there is severe destruction and resorption of the distal radius and ulnar, measurement scans in the RA population should be in a position similar to those of the control group. In this study, pQCT trabecular and total BMD measurements were marked reduced in the RA groups (36.1% and -15.6% respectively) compared with controls, with very little difference between the RA groups. These data are in keeping with previous studies examining radial bone loss in patients with non-steroid treated RA using pQCT (11.11,11.12). There are however some important differences between the studies. The previous longitudinal studies examined premenopausal and postmenopausal women with early (< 3 years) RA, whilst the cross-sectional study population reported in this thesis is purely postmenopausal with longer and variable disease duration. The pQCT scanners used also differed. An ¹²⁵I rather than x-ray radiation source

<u>an la caminata da cana da sera da cana da sera </u>

was used previously, trabecular BMD was measured at the 8-10%, and cortical BMD at the distal 1/3 site (rather than the 4% site here). As the proportion of trabecular and cortical bone changes dramatically over short distances at the distal radius (1.17,1.19), these are important differences. Nevertheless, annual rates of loss up to 25% were recorded for trabecular BMD, whilst cortical BMD changes were similar to the control group (11.11), results similar in distribution and magnitude to ours, Rapid trabecular bone loss at the radius occurs early in the disease process with subsequent slowing (11.12). In view of the substantially lower trabecular BMD found in the RA groups, data presented here would suggest that there the is considerable ongoing bone loss beyond this initial period of rapid loss, although this would be better addressed by a longterm longitudinal study. The strong correlation of whole hand and wrist Larsen scores with trabecular BMD suggests that local inflammatory mechanisms responsible for joint damage, and resultant wrist immobility have important adverse effects on ultra-distal radial bone mass, as found previously (11.13). An increasing rheumatoid factor titre was negatively correlated to trabecular bone, which may be related to more aggressive local disease in such individuals. The correlation of grip strength with trabecular BMD supports the concept that mechanical loading and stimulation has a beneficial effect upon local bone mass (4.2). In this case, it is detectable within the trabecular component, which being more metabolically active, is more responsive to such stimuli.

As corticosteroid induced bone loss preferentially affects trabecular bone, one might expect pQCT to be a sensitive technique for detecting such an effect. However, it is apparent that there is no significant corticosteroid effect on any of the forearm BMD measurements, even within the trabecular component. The subcortical measurement was significantly lower

214

ligh shellow an index of the second of the sequence proves a second state of the

to the test of the second

in the steroid treated RA group, but not the non-steroid treated RA group compared to controls, but there was no significant difference between the RA groups. The lack of corticosteroid effect is almost certainly due to the juxtaarticular, inflammatory driven bone loss (11.14) overwhelming other mechanisms affecting trabecular bone at this site. The is radial measurement site therefore an important consideration. That for SPA is more proximal than pQCT, and as such is further removed from the inflammatory process occurring at the wrist. The effect of corticosteroid could therefore be more pronounced at a more proximal site, which might explain why some have found significantly lower radial bone mass in patients with corticosteroid treated RA (11.2). In keeping with the results reported above, others found no significant corticosteroid effect at the radius (11.3,11.15).

Similarly, immediate postmenopausal bone loss preferentially affects trabecular rather than cortical bone, yet the duration since menopause was negatively correlated with total, subcortical and cortical, but not trabecular BMD. Presumably this is also due to the periarticular inflammatory process overwhelming menopausal effects on trabecular bone, but having much less effect upon cortical bone, which is metabolically less active. Cortical bone comprises 55% of the pQCT total BMD measurement which explains the relationship of duration since the menopause with the latter BMD. Cortical bone is also known to undergo postmenopausal as well as age related change (1.51), as is the pQCT subcortical measurement as discussed in chapter 7.

There were no difficulties in performing ultrasound of the os calcis in these RA patients as the foot recess was wide enough to accommodate broad feet, and application of the transducers was not impaired by the degree of subtalar deformity. There

215

(a) A set of the se

were reductions in ultrasound attenuation and velocity of -31.7% and -6.6% respectively in the combined RA group, again with no additional significant steroid effect. Weight was an important determinant of ultrasound measurements as was the index in the RA groups. The latter finding is Larsen particularly intriguing as the Larsen index was measured at the hand and wrist. The same inflammatory mechanisms causing bone loss at the wrist may well effect the trabecular rich os calcis, explaining the correlation between the Larsen index at the wrist and the ultrasound measurements at the os calcis. As for radial trabecular BMD, and presumably for the same reason, calcaneal ultrasound measurements were negatively both correlated with a rising titre of rheumatoid factor. It is difficult to explain the correlation between grip strength and BUA, but this may well reflect similar long-term disease activity in the hands and feet in this group of patients.

10 IN 11 TO 12 AV

There have been many previous studies addressing axial BMD in and these have been reviewed recently (11.1). DXA RA, measurements were included here to allow comparison with pQCT and ultrasound within the same population. A significant reduction in femoral neck BMD of 12.8% (0.71 SD) and 15.4% (0.93 SD) in the RA and RA+P groups respectively, with no significant corticosteroid effect is in keeping with most of the present literature (11.6,11.7,11.8,11.15,11.16). Such reductions are associated with an increased risk of future hip fracture (1.192), which has been recently demonstrated in the RA population (11.17). Weight and physical activity were important determinants of hip BMD in keeping with previous work (11.7,11.8), suggesting that patient mobility helps maintain hip BMD.

The 4.5% and 7.9% reductions in lumbar spine BMD for the RA and RA+P groups respectively failed to reach significance and are

a the second second

smaller than those recorded at the hip, radius and calcaneus. Although the groups were well matched for age, we cannot age and fracture related spinal osteoarthrosis exclude influencing our spinal BMD results (1.171 - 1.177), Interpretation of the lumbar spine results in the light of previous work has to be cautious because of differences in measurement techniques, study populations and dose of corticosteroid. In those studies which examined postmenopausal women, Hall et al studied larger numbers and found a larger and significant difference of 7.5% between control and non-steroid treated RA groups (11.8). Although a significant difference of 6.5% was found between steroid and non-steroid treated RA groups, the difference was smaller (4.3%), and non-significant when a subgroup with a lower cumulative dose of corticosteroid was examined. Verstraetaen et al studied the effect of higher dose prednisolone in women with much longer disease duration than their non-steroid treated counterparts (11.3). Also, rather unusually, lumbar spine BMD was much higher in the nonsteroid RA group compared to control group, suggesting that either the RA or control group was non-representative. Using QCT, Laan et al found a reduction of 34.7% in lumbar spine trabecular BMD in a low dose steroid-treated RA group compared to a non-steroid treated group (11.18), which is much greater than that reported here. However, QCT measures purely trabecular bone, whereas DXA measures a composite of cortical and trabecular bone, and as such, the techniques are clearly not comparable. In patients with RA, one would not expect differences in radial trabecular BMD (as measured by pQCT) to be comparable to differences in spinal trabecular BMD (as measured by QCT), as the spine is far removed from the periarticular inflammatory process which profoundly influences radial trabecular BMD. Also, there is poor correlation between radial and axial BMD in the general population (11.19, see

217

chapter 6). Compared to controls, Lane et al (11.15) found DXA

lumbar spine BMD 8.4% lower in non-steroid treated RA patients, but a non-significant difference for steroid treated patients, although the steroid treated group was small (n=13 for LS BMD measurements). The greater difference in lumbar spine BMD for the non-steroid treated group in comparison to the results presented here, can be explained by the longer duration of RA (mean: 24.5 yrs) in the group studied by Lane et al. Others have examined populations which differed in gender and menopausal status (1.131,11.6,11.7,11.16), or studied the effect of higher dose corticosteroid (11.3,11.7,11.8,11.16) which makes direct comparison with the results reported here difficult. An increased prevalence of vertebral fracture has been demonstrated in steroid treated postmenopausal women with RA (11.20), although there was no non-steroid treated RA group for comparison. The increased prevalence was greater than expected from DXA spinal BMD differences, suggesting a steroid effect on bone architecture as well as mass.

This study has highlighted important changes in appendicular bone mass in RA. There is now good evidence that bone loss occurs early in RA at the radius (11.11,11.12), hip and spine (11.4), with axial bone loss being related to disease and physical activity (11.4). Recently there has been interest expressed in measuring appendicular BMD as a method of monitoring disease activity in RA (11.1,11.21). pQCT and calcaneal ultrasound, by quantifying trabecular bone adjacent to the carpus and ankle respectively, offer novel and unique opportunities to study the effect of RA on periarticular bone, and may be valuable methods of monitoring response to treatments such as disease modifying anti-rheumatic drugs and corticosteroid. Further work is necessary to examine the clinical use of pQCT and ultrasound in RA.

218

and here a property and a second second second and a second second second second second second second second se

Table 11.1. Characteristics of the control, rheumatoid arthritis (RA) and steroid treated rheumatoid arthritis (RA+P) groups.

	4 4 4	RA	Controls	Overal1
	(n = 25)	(n = 21)	(n = 29)	p-value
Age (years)	65.1 (6.3)	64.1 (6.4)	63.8 (6.3)	n.s.
Height (m)	1.59 (0.07)	1.58 (0.05)	1.57 (0.06)	п. г.
Weight (kg)	66.2 (12.2)	63.5 (11.2)	62.6 (10.2)	n.s.
Years since menopause	18.0 (8.3)	17.5 (7.2)	14.9 (9.2)	n.s.
Physical activity index	28.7 (2.7) **	29.2 (2.5) *	31.2 (1.5)	0.0002
Dietary calcium intake (mg/day)	706.5 (369.1)	835 (165.6)	836 (232.8)	n.s.
Number (%) of smokers	8 (32)	6 (28.6)	5 (17.2)	n.s.
Number (%) who drink alcohol	5 (20)	2 (9.5)	19 (65.5) 0.0	-
Number (%) on thiazide diuretics	2 (8)	l (4.8)	3 (10.3)	n.s. ⁽¹⁾
Number (%) with vertebral fractures	5 (20)	5 (23.8)	1 (3.4)	п. г.
Disease duration, years §	12 (2-30)	8 (2-35)		n.s.
Health assessment questionnaire	1.95 (0.69)#	1.4 (0.77)		0.014
15m Walking time, secs	16.9 (6.6)	16.8 (13.2)		n.s.
Ritchie articular index	11.0 (8.3)	9.6 (8.1)		n.s.

Table 11.1	(cont).	Characteristics of the control,	rheumatoid arthritis	(RA)	and steroid	treated	rheumatoid
artartas	(A+FX)	groups.					

	RA + P	RA	Controls	Overall
	(n = 25)	(n = 21)	(n = 29)	p-value
Mean grip strength, mmHg	90.5 (41.4)	111.3 (44.4)		n.s.
<i>J</i> isual analogue paín score, mm	54.0 (32.3)	55.4 (43.4)		п, 5.
dean Larsen score	41.4 (13.8)	34.1 (13.5)		п.в.
Number (%) Rheumatoid Factor +ve	21 (84)	10 (47.6)		0.006
C-reactive protein, mg/dl	1.94 (1.83)	1.41 (1.79)		п.s.
Daily prednisolone dose, mg §	6.5 (2.5-10)			
lumulative prednisolone dose, grams §	8.21 (1.8-65.	2)		
Prednisolone duration, years 5	4.0(1-21)			

Unless otherwise stated, values are mean (SD) and § median (range)

Continuous variables compared by ANOVA (* p<0.05, ** p<0.001 versus controls: Scheffe's multiple comparison test) and unpaired t-test (# p<0.05 versus RA group).

Categorical variables compared by χ^2 test (⁽¹⁾ = Fisher exact χ^2 test after compression of (RA) & (RA+P) groups into a single group)

spine and hip as	letermined by stepwise multiple linear regression of data	from all subjects.
Qtot BMD	= -48.48(RA) - 2.123(PMYR) + 372.2	$(R^2=0.145, Sig F = 0.0017)$
Qtrab BMD	= 1.098 (wt) - 63.83 (RA) + 105.5	$(R^2=0.283, Sig F < 0.0001)$
Qscort BMD	= -57.88 (RA) - 3.36 (PMYR) + 516.9	$(R^{2}=0.190, Sig F = 0.0002)$
Qcort BMD	= -4.96(PMYR) - 10.44(PAQ) + 894.59	$(R^2=0.184, Sig F = 0.0003)$
BUA	= 0.748(wt) - 24.55(RA) + 22.99	$(R^2=0.403, Sig F < 0.0001)$
NOS	= 1.316(wt) - 106.03(RA) + 1436	$(R^2=0.571, Sig F < 0.0001)$
L'S BMD	= 0.007(wt) + 0.015(PAQ) - 0.05	$(R^3=0.228, Sig F < 0.0001)$
FN BMD	= 0.003(wt) + 0.022(PAQ) - 0.073(RA) - 0.157	(R ² =0.385, Sig F <0.0001)
FT BMD	<pre>= 0.005(wt) + 0.018(PAQ) - 0.057(RA) - 0.264</pre>	(R ² =0.387, Sig F<0.0001)
FW BMD	= 0.024 (PAQ) - 0.21	(R ² =0.178, Sig F=0.0001)
wt = weight, PAQ	physical activity, RA = Rheumatoid Arthritis (coded 0=n	o, 1=yes)
PMYR = years post	lenopause ,	
R^2 = adjusted R^2 ,	lig F = significance of regression equation	

والمحافظة المحال والمحاجر المحافظين

. . . .

rahaat laaf ti

. .

Server also

122.6

1.00

ere al properties and the second s

State of Action and State of Action and Action and Action

221

and the second secon Second second

10

12.1

Table 11.3. Fearson correlation co-efficients for RA disease parameters and radial pQCT total (Qtot), trabecular (Qtrab), subcortical (Qscort) and cortical (Qcort) BMD measurements and calcaneal ultrasound attenuation (BUA) and velocity (VOS) for the combined steroid (RA+P) and non-steroid (RA) treated RA groups.

	Qtot	Qtrab	Qscort	Qcort	BUA	VOS
Years with RA	-0.06	-0.18	10.01	0.07	-0.09	-0.14
Years on prednisolone	0.01	0.06	-0.11	-0.13	-0.01	-0.23
Daily prednisolone dose	-0.08	-0.12	-0.13	-0.08	-0.2	-0.28
Cumulative prednisolone dose	-0.08	-0.02	-0.17	-0.15	-0.05	-0.21
Ritchie tenderness score	0	0.05	-0.02	-0.1	0.14	0.02
Grip strength (non-dominant hand)	0.1	0.35*	-0,08	0.02	0.33*	0.22
Visual analogue pain score	0.13	-0.11	0.21	0.22	-0.14	-0.08
Health assessment score	0.11	-0.19	0.15	0.23	-0.29	-0.23
Physical activity score	-0.04	0.19	-0.07	-0.19	0.29	0.19
C reactive protein	0.01	-0.18	0.02	0.06	-0.27	-0.12
Rheumatoid factor	-0.11	-0.44**	0.03	0.24	-0.45**	-0.36*
Larsen score (non-dominant hand)	-0.18	-0.63***	-0.07	0.25	-0.66***	+65.0-
Larsen score (non-dominant wrist)	-0.08	-0.55***	0.02	0.33*	-0.59***	-0.4**

Significance values are: * 0.05>p>0.01, **0.01>p>0.001, ***p<0.001, otherwise not significant.



Figure 11.1: DXA hip and spine measurements for control, non-steroid (RA)



224



Figure 11.2: pQCT measurements for control, non-steroid (RA)











energi in traductione energia de la constructione en energia de la construction de la construction de la constr


Figure 11.5: Relationships between BUA and disease variables for combined RA groups





(1494) (1

CHAPTER 12

OVERALL CONCLUSIONS AND DISCUSSION.

Briefly, the aims of this thesis were to examine the use of pQCT in the field of osteoporosis. The precision of pQCT was studied in different populations. The effect upon precision of minimising the difference between scan voxel numbers in followup scans was also investigated. The effect of dominance in individuals of normal day-to-day activity was studied. Normative data from the local female population was obtained, important anthropometric determinants from which, were determined, and estimated rates of change derived. pOCT was variably compared to DXA measurements and calcaneal ultrasound in a number of different situations. These were: perimenopausal changes in BMD; the relationships with other bone mass measurements and the implications for screening; discrimination of vertebral and hip fracture populations from non-fracture controls; the effect of HRT and etidronate therapy upon BMD and the implications for monitoring treatment; the effect of warfarin upon BMD; and the effect of disease and low dose corticosteroid in postmenopausal women with rheumatoid arthritis.

しゃんきょうりょうがない たいしたい

14 C 4 C 14 C 1

The precision of the Stratec XCT 960 pQCT scanner was comparable to other scanning modalities such as DXA with precision values of 1-2%. Although less precise than the multislice pQCT technique, it is cheaper and more ubiquitous. The precision was found to be poorer in older postmenopausal women with vertebral fracture, and in women with rheumatoid arthritis It is likely that the precision of the Stratec XCT 960 pQCT system could be improved by increasing the number of scan slices, thus increasing the measurement sample size. However, this would result in a longer scan time, and consequently a greater chance of movement artifact. The local radiation dose

229

in and and the second and a start of the start of the second start of the second start of the second start of t

would also be increased, although this is unlikely to be a problem, as the radiation dose associated with the present system is so low.

이 집은 사망한 것은 것은 것은 것이 가지 않는 것이 것이다.

Precision was shown to be influenced by the difference between voxel numbers of repeat scans, suggesting that scan voxel numbers can be used as a surrogate marker of cross-sectional area, and an aid to positioning during repeat scanning. This is an important consideration in repeat scanning of the ultradistal radius, as absolute BMD, and the proportion of cortical and trabecular bone change significantly over short distances (1.16 - 1.19). With the present Stratec system, bevonä standardization of the scanning technique, further improvement in precision is possible by limiting the difference in voxel numbers between repeat scans. At the present time, a difference of 30 voxels or less is recommended from the work presented in chapter 3. However, this should be further validated, for example, by an in-vitro cadaveric study. Dominance influenced radial BMD by only 1-2% in individuals of normal physical activity. Although not statistically significant, it does have important clinical implications. A lack of consistency with respect to which forearm is scanned, would introduce a further 1-2% variance in BMD measurements. It was our policy therefore, to always scan the non-dominant forearm unless there had been a previous Colles fracture, and to always scan the same forearm in longitudinal studies.

The importance of acquiring a normal range based upon the local population was shown in chapter 5. BMD values of the normal range derived from the Grampian region of Scotland were greater than those of the manufacturers and a previously reported German population (1.50). The population sample was somewhat lacking in numbers in the 30-39 years age group, which unfortunately negates any meaningful attempt to address the

230

issue of timing of peak bone mass. Not surprisingly, age was negatively correlated with all pQCT BMD measurements in multiple regression analyses. Weight was found to be positively correlated, inspite of the radius being a non-weight bearing site. The negative correlation for height with total and trabecular BMD measurements is surprising, but may indicate that taller women have thinner radii.

internet with the particular

10.0

1277 1 1 1 1

The cross-sectional pQCT data presented in this thesis indicates that radial BMD was stable from the age of 18 years until the menopause, which is inkeeping with almost all of the previous literature(1.50,1.51,1.63,1.68,1.69,1.70,1.78,1.79). However, the longitudinal data from 23 normal premenopausal women (mean age 47.2 year, range 46-54 years) showed a significant decrease in total and subcortical BMD (-1.04% and -1.35% respectively) after follow up for one year, inspite of ongoing regular menses. The differences in results between estimated and observed rates of change illustrate that population variability hides small changes in BMD in crosssectional data, which can be better determined in precise, longitudinal studies. Premenopausal changes in radial BMD should be clarified with more long term, prospective studies involving greater numbers of participants.

The menopause is considered to induce profound bone loss. The pQCT data in chapter 7 of this thesis, derived from the crosssectional studies supports this finding in part. When all postmenopausal women were considered as a single group, estimated rates of bone loss up to -1.39%/yr were found. Surprisingly however, the rate of trabecular BMD loss for postmenopausal women was generally lower than that of other pQCT BMD measurements. It also gradually increased with the duration since the menopause, which is contrary to the accepted theory that trabecular bone is preferentially affected

immediately following the menopause (1.49,1.64,1.66,1.67,1.74-1.77), at least at the radius. In contrast, the work in chapter 7 showed that subcortical bone was preferentially affected in the early postmenopausal years. This supports the recent observation that during this period, remodelling results in trabecularization of endosteal cortical bone and expansion of the intramedullary radial cavity at the expense of the cortical shell, whilst there is little change in trabecular BMD (6.6). This was confirmed when changes in pQCT BMD around the comparing examined further menopause were by early postmenopausal and late premenopausal women in a crosssectional study, where a greater difference was found in subcortical than trabecular BMD. In this perimenopausal group, it was also apparent that difference in radial BMD measurements were less than those found at the hip and spine as measured by DXA.

It is appreciated that bone mass at one skeletal site does not necessarily reflect that of another. This was borne out in the work presented in chapter 6, where there was at best, only moderate correlation between radial pQCT BMD measurements, calcaneal ultrasound and axial hip and spine measurements. Additionally, the longitudinal studies which examined annualized rates of change in premenopausal and postmenopausal women in chapter 7, albeit in small numbers, and only after 12-18 months of follow-up, showed that there was no relationship between changes at the radius and those at either the hip or spine. Further study of these relationships involving a greater number of individuals followed for a longer duration are required. Pending these studies and prospective studies addressing fracture prediction using pQCT, these preliminary findings suggest that pQCT has little role as either a screening or pre-screening tool (eg. for axial DXA assessment), should population based screening for osteoporosis risk ever

232

and the second second

become a reality. Essentially a single radial measurement did not adequately predict axial BMD, and serial measurements did not reflect changes in axial BMD in the short-term.

and the second second second

316

n in the single end of the North

1. 1. 1. M.

At the present time, fracture prediction is best achieved by BMD assessment. In chapter 8, a retrospective study comparing axial DXA with radial pQCT measurements showed that radial trabecular BMD was superior to all other measurements in discriminating vertebral fracture from non-fracture populations. In contrast, the discriminatory power of all pQCT measurements was poor in a retrospective study of hip fractures, and significantly worse than DXA femoral neck BMD. This suggest that pQCT may be useful in the prediction of vertebral fractures, but less so for hip fractures. Prospective studies are now required to confirm this suggestion.

Monitoring response to drug therapy is an established indication for BMD assessment (1.145,1.146,1.153,1.154,9.1). Preliminary work presented in chapter 9 compared changes in pQCTradial BMDmeasurements with DXA hip anđ spine measurements in response to HRT and cyclical etidronate during the first year of treatment. The results showed that the response to both drugs follows a similar pattern, with a significant increase in spinal BMD evident within 4 months and further increase thereafter. In contrast, there was a more gradual increase in hip BMD, while radial BMD, including the trabecular measurement, remained virtually unchanged. This suggests that in the majority of cases, response to these therapies is best assessed at the lumbar spine. This needs to be confirmed by larger studies with longer follow-up. It should be borne in mind however, that as lumbar spine DXA measurements can be influenced by severe spondylosis (1.171-1.177), multiple fractures overlying and aortic calcification (1.168,1.169,1.170), it is not always the best measurement for

233

and the second second second second

monitoring purposes. Based upon the data presented here, there is no suggestion that pQCT could be used as an alternative, although further work, as suggested above, is necessary before its use in this area can be completely discounted.

Monitoring the effect of drugs with an adverse effect upon bone equally important. Such an effect is epitomised by is corticosteroid therapy. Unfortunately, it was not possible to examine the role of pQCT in monitoring steroid induced bone loss per se, but this area is undoubtedly deserving of future study. It was however possible to examine the potential adverse effect of warfarin upon bone, a side-effect seldom considered. In a case control study of males presented in chapter 10, lumbar spine and radial trabecular BMD were almost one standard deviation lower at each site in those men on longterm warfarin compared to the control group. Bearing in mind the increasing use of warfarin, this is an important adverse effect to be aware of. In considering the effect of HRT, etidronate and warfarin on radial BMD, there appears to be disparity in the results. It seems strange that radial trabecular BMD was unchanged with both HRT and etidronate, yet significantly decreased by warfarin. The study designs must be borne in mind however. The HRT/etidronate effects were studied for only one year in a longitudinal study, whereas the warfarin effect was examined in a cross-sectional design where the drug had been given for a median duration of 40.5 months (range 4-302). The disparity in results could be related to the duration of drug therapy, warfarin having had a much longer time to exert its effect upon radial bone, than HRT or etidronate. With longer follow up, a significant increase in radial BMD may have been found with HRT and etidronate.

Rheumatoid arthritis (RA) is one of the many conditions which is known to induce bone loss and cause osteoporosis (11.1,11.4-

234

الرواف المترجع المراجع

an and the second second second to the second s

11.8,11.11,11.12,11.15-11.17,11.20). The pathogenesis is multifactorial, and the effect of low dose prednisolone-at all sites controversial. In chapter 11, the effect of A and low dose prednisolone upon radial pQCT BMD measurements wag investigated in a case control study, with comparison of BMD at other skeletal sites. Trabecular BMD was reduced by more than 30% in patients with A compared to controls, with very little additional corticosteroid effect. In contrast cortical BMD was unaffected. A similar significant difference was detected at the os calcis, a smaller but significant difference at the hip, and an insignificant difference at the lumbar spine. This study supports the growing body of evidence suggesting that the use of low dose corticosteroid in patients with A does not have a deleterious effect upon BMD. The effect upon periarticular trabecular BMD at the wrist in this study (as measured by pQCT) was dependent upon articular damage, which is the longterm consequence of uncontrolled inflammatory joint disease. Previous work by others has shown loss of radial trabecular BMD (as measured by pQCT) early in the natural history of the disease (11.11,11.12), and the data presented here suggests further loss in the longer term. This begs the question, could assessment of trabecular BMD by pQCT be used to monitor disease progression, and possibly response to therapy? Further work, preferably in interventional, prospective studies of early A are needed to address this issue.

It is apparent from the work presented here that radial BMD behaves differently from that at the hip and spine. There was poor correlation between radial and axial measurements. There was no correlation between the rate of change at the radius and that at the hip or spine. The response to drugs such as HRT and etidronate differed between sites. Although pQCT has the advantage of assessing trabecular and cortical BMD independently, radial bone seems less active, and responds

235

differently and more slowly than does axial bone. This may well be due to differences in remodelling at the radius as discussed above in this chapter, and in differences in the bone marrow present between sites. Yellow marrow is present in the radial medullary cavity throughout adulthood, whereas haematopoietic spine and proximal marrow predominates at the femur. Osteoclasts and osteoblasts are derived from haematopoietic marrow. The signalling and control of osteoclasts/osteoblasts and remodelling is extremely complex anđ not fully characterised, but thought to involve numerous cytokines, hormones and matrix factors (1.2,1.3,1.8,12.1). The close proximity to, and interaction with haemopoietic rather than yellow marrow may well explain some of the different behaviour of axial and radial bone which was apparent from the work presented in this thesis.

Osteo-densitometry is a relatively new concept and significant advances have been made in this field in recent years. Although present day osteo-densitometers are the best method of assessing "osteoporosis" in its broadest context, research into pQCT and other modalities, both established and novel, must continue to ensure future progress. Two and three dimensional QCT systems have been developed (1.149,1.150,8.35,12.2-12.8) which allow structural analysis of trabecular bone, providing effect, a non-invasive bone biopsy. in This structural information along with BMD may further improve the performance of pQCT (12.7). Scanners with a higher spatial resolution than that of the Stratec XCT-960 system have largely been used for this, although an attempt at structural analysis by enhancing the Stratec system has been reported to show an increased number of perforations in early postmenopausal compared to late premenopausal women inspite of an insignificant difference in BMD (7.5).

236

<u>1. A second se second sec</u> The work for this thesis evaluating the role of pQCT in osteoporosis was done within a limited timescale. Consequently it has its limitations and is far from being all inclusive. However, it has addressed some of the important issues which must be covered in the development of an osteo-densitometer, and has highlighted areas where pQCT may, and may not have a role in the management of patients with osteoporosis. Further work is necessary to confirm some of these findings, and address areas not covered by this work. Like most research, issues hitherto not realised, or poorly documented, have been raised which are worthy of future investigation. The most likely fruitful areas for further research and development into pQCT would be:

1.1.1.1

- The use of scan voxel numbers to improve precision, and aid repositioning in repeat scanning.
- The use of pQCT to study age related changes in BMD and bone remodelling.
- The use of pQCT in fracture prediction, more especially vertebral fracture. It may have additive predictive value in conjunction with axial DXA, as has calcaneal ultrasound in hip fracture prediction (8.19).
- The use of pQCT in determining BMD changes in response to corticosteroid therapy
- The use of pQCT as an assessment tool for monitoring disease progression and response to therapeutic intervention in rheumatoid arthritis, and possibly other inflammatory arthritides.
- Further assessment of structural considerations, both radial geometric indices, and trabecular microstructure.

237

have the first of the second state of the seco

REFERENCES

1.1 Department of Health. Report from the Advisory Group on Osteoporosis. Barlow DH: Department of Health, Nov 1994.

a second a second second a second second

- 1.2 McDonald BR, Gowen M. Cell biology of bone. In: DM Reid, editor. Osteoporosis. Baillieres Clin Rheumatol 1993:7(3);421-39.
- 1.3 Manolagas SC, Jilka R. Bone marrow, cytokines and bone remodelling. N Engl J Med 1995;332:305-11.
- 1.4 Risteli L, Risteli J. Biochemical markers of bone metabolism. Ann Med 1993;25:385-93.
- 1.5 Wahner HW, Dunn WL, Riggs BL. Assessment of bone mineral. J Nucl Med 1984;25:1134-41.
- 1.6 Smith MW, Walker RR. Femoral expansion in aging women: Implications for osteoporosis and fracture. Science 1964;145:156-7.
- 1.7 Fournier PE, Rizzoli R, Slosman DO, Buchs B, Bonjour JP. Relative contribution of vertabral body and posterior arch in female and male lumbar spine peak bone mass. Osteoporosis Int 1994;4:264-72.
- 1.8 Parfitt AM. Osteonal and hemi-osteonal remodelling: the spatial and temporal framework for signal traffic in adult human bone. J Cell Biochem 1994;55:273-86.
- 1.9 Mosekilde L. Sex differences in age-related loss of vertebral bone mass and structure-biomechanical consequences. Bone 1989;10:425-32.
- 1.10 Parfitt AM. Bone remodelling: relationship to the amount and structure of bone, and the pathogeneesis and prevention of fractures. In: Riggs BL, Melton LJ, editors. Osteoporosis: Etiology, Diagnosis and Management. New York: Raven Press, 1988:45-93.
- 1.11 Lindsay R. Pathogenesis of postmenopausal osteoporosis. In: Reid DM, editor. Osteoporosis. Baillieres Clin Rheumatol 1993;7(3):499-513.

238

يمجي المحرور المحري مع المحر المحرور <u>المحرور المحرور المحرور المحرور المحرور المحرور المحرور المحرور المحرور ا</u>

1.12 Dempster DW, Shane E, Horhert W, Lindsay R. A simple method for correlative light and scanning electron microscopy of human iliac crest bone biopsies: qualitative observation in normal and osteoporotic subjects. J Bone Miner Res 1986;1:15-21.

化化学 化化学性化化 法自己的现在分词

1999 N. 1998

(1) (2) (2) (2) (2)

- 1.13 Lang SM, Moyle DD, Berg EW et al. Correlation of mechanical properties of vertebral trabecular bone with equivalent mineral density as measured by computed tomography. J Bone Joint Surg 1988;70A:1531-8.
- 1.14 Ganong WF. Circulating body fluids. In: Ganong WF, editor. Review of medical physiology. Los Altos: Lange Medical Publications, 1985:421-41.
- 1.15 De Bisschop E, Luypaert R, Louis O, Osteux M. Fat fraction of lumbar bone marrow using in vivo proton nuclear magnetic resonance spectroscopy. Bone 1993;14:133-6.
- 1.16 Wahner HW, Eastell R, Riggs BL. Bone mineral density of the radius: where do we stand? J Nucl Med 1985;26:1339-41.
- 1.17 Schlenker RA, Von Seggen WW. The distribution of cortical and trabecular bone mass along the lengths of the radius and ulna and the implications for in vivo bone mass measurements. Calcif Tissue Res 1976;20:41-52.
- 1.18 Wahner HW, Eastell R, Riggs BL. Bone mineral density of the radius: Where do we stand? J Nucl Med 1985;26:1339-41.
- 1.19 Takada M, Engelke K, Hagiwara S, Grampp S, Genant HK. Accuracy and precision study in vitro for peripheral quantitative computed tomography. Osteoporosis Int 1996;6:207-12.
- 1.20 Riggs BL, Wahner HW, Seeman E et al. Changes in bone mineral density of the proximal femur and spine with aging J Clin Invest 1982;70:716-23.
- 1.21 Vogel JM, Wasnich RD, Ross PD. Clinical relevance of calcaneus bone mineral density measuremeths: a review. Bone Miner 1988;5:35-58.

239

r<u>en an an teach ta sao an</u> an teach

and the second secon

1.22 Melton LJ III, Chao EY, Lane J. Biomechanical aspects of fractures. In: Riggs BL, Melton LJ III, editors. Osteoporosis: Etiology, Diagnosis, and Management. New York: Raven Press, 1988:111-31

and strength of the

- 1.23 Hansson T, Roos B, Nachemson A. The bone mineral content and ultimate compressive strength of lumbar vertebrae. Spine 1980;5:46-55.
- 1.24 Dempster DW, Ferguson-Pell MW, Mellish RW et al. Relationship between bone structure in the iliac crest and bone structure and strength in the lumbar spine. Osteoporosis Int 1993;3:90-6.
- 1.25 Moro M, Hecker AT, Bouxsein ML, Myers ER. Failure loads of thoracic vertebtae correlates with lumbar bone mineral density measured by DXA. Calcif Tissue Int 1995;56:206-9.
- 1.26 Lotz J, Hayes W. The use of quantitative computed tomography to estimate risk of fracture of the of the hip from falls. J Bone Joint Surg 1990;72A:689-700.
- 1.27 Courtney A, Wachtel EF, Myers ER, Hayes WC. Age releted reductions in the strength of the femur tested in a fall loading configuration. J Bone Joint Surg 1995;77A:387-95
- 1.28 Beck T, Ruff CB, Warden KE, Scott M, Rao G. Predicting femoral neck strength from bone mineral data: a structural approach. Invest Radiol 1990;25:6-18.
- 1.29 Horsman A, Currey JD. Estimation of mechanical properties of the distal radius from bone mineral content and cortical width. Clin Orthop 1983;176:298-304.
- 1.30 Myers ER, Sebeny EA, Hecker AT et al. Correlation between photon absorption properties and failure loads of the distal radius in vitro. Calcif Tissue Int 1991;49:292-7.
- 1.31 McBroom RJ, Hayes WC, Edwards WT, Goldberg RP, White AA. Prediction of vertebral body compressive fracture using quantitative computed tomography. J Bone Joint Surg 1985;67A;1206-14.

240

مهم ومسوحها والحاصر والمحصي وولعية فواد والقورة والقورة والأورون والع

1.32 Carter DR, Hayes WC. Bone compressive strength: the influence of density and strain rate. Science 1976;194:1174-6.

and the second second

- 1.33 Carter DR, Hayes WC. The compressive behaviour of bone as a two phase porous structure. J Bone Joint Surg 1977;59A:954-62.
- 1.34 Rockoff SD, Sweet E, Bluestein J. The relative contribution of trabecular and cortical bone to the strength of human lumbar vertebrae. Calcif Tissue Res 1969;3:163-75.
- 1.35 Mazess RB. Fracture risk: a role for compact bone. Calcif Tissue Int 1990;47:191-93.
- 1.36 Spadaro JA, Werner FW, Brenner RA, Fortino MD, Fay LA, Edwards WT. Cortical and trabecular bone contribute strength to the osteopenic distal radius. J Orthop Res 1994;12:211-8.
- 1.37 Louis O, Boulpaep F, Willnecker J, Van den Winkel P, Osteaux M. Cortical mineral content of the radius assessed by peripheral QCT predicts compressive strength on biomechanical testing. Bone 1995;16:375-9.
- 1.38 Specker BL, Brazerol W, Tsang RC, Levin R, Searcy J, Steichen J. Bone mineral content in children 1 to 6 years of age. Am J Dis Child 1987;141:343-44.
- 1.39 Glastre C, Braillon P, David L, Cochat P, Meunier PJ, Delmas PD. Measurement of bone mineral content of the lumbar spine by dual energy x-ray absorptiometry in normal children: correlations with growth parameters. J Clin Endocrinol Metab 1990;70:1330-3.
- 1.40 Hui SL, JohnstonCC, Mazess RB. Bone mass in normal children and young adults. Growth 1985;49:34-43.
- 1.41 Faulkner RA, Bailey DA, Drinkwater DT, McKay HA, Arnold C, Wilkinson AA. Bone densitometry in Canadian children 8-17 years of age. Calcif Tissue Int 1996;59:344-51.

generation de la companya

<u>, na serie de la serie de s</u>

- 1.42 De Schepper J, Derde MP, Van den Broeck M, Piepsz A, Jonckheer MH. Normative data for lumbar spine bone mineral content in children: influence of age, height, weight and pubertal stage. J Nucl Med 1991;32:216-20.
- 1.43 Gilsanz V, Gibbens DT, Roe TF et al. Vertebral bone density in children: effects of puberty. Radiology 1988;166:847-50.
- 1.44 Sabatier JP, Guaydier-Souquieres G, Laroche D et al. Bone mineral acquisition during adolescence and early adulthood
 a study in 574 healthy females 10-24 years of age. Osteoporosis Int 1996;6:141-8.
- 1.45 Theintz G, Buchs B, Rizzoli R et al. Longitudinal monitoring of bone mass accumulation after 16 years of age at the levels of lumbar spine and femoral neck in female subjects. J Clin Endocrinol Metab 1992;75:1060-5.
- 1.46 Bonjour J, Theintz G, Law F, Slosman D, Rizzoli R. Peak bone mass. Osteoporosis Int 1994:4(Suppl 1):7-13.
- 1.47 Johnston CC, Slemenda CW. Peak bone mass, bone loss and risk of fracture. Osteoporosis Int 1994;4(Suppl 1):43-5.
- 1.48 Nordin BEC. Bone mass, bone loss, bone density and fractures. Osteoporosis Int 1993;3(Suppl 1):1-7.
- 1.49 Sambrook P, Kelly P, Eisman J. Bone mass and aging. In: Reid DM, editor. Osteoporosis. Baillieres Clin Rheumatol 1993;7(3):479-98.
- 1.50 Butz S, Wuster C, Scheidt-Nave, Gotz M, Ziegler R. Forearm BMD as measured by peripheral quantitative computed tomography (pQCT) in a German reference population. Osteoporosis Int 1994;4:179-84.
- 1.51 Geusens P, Dequeker J, Verstraeten A, Nijs J. Age, sex, and menopause related changes of vertebral and peripheral bone: population study using dual and single photon absorptiometry. J Nuc Med 1986;27:1540-9.

<u>en an de l'estados de l'astronomias de la constance de la constance de la constance de la constance de la const</u>

1.52 Kelly PJ, Eisman JA, Twomey L, Sambrook PN. Sex differences in peak adult bone mass. J Bone Miner Res 1990;5:1169-75.

.

- 1.53 Karantanis AH, Kafel-Ezra JA, Glaros DC. Quantitative computed tomography for bone mineral measurement: technical aspects , dosimetry, normal data and clinical applications. Br J Radiol 1991;64:298-304.
- 1.54 Cann CE, Genant HK, Kolb FO, Ettinger B. Quantitative computed tomography for prediction of vertebral fracture risk. Bone 1985;6:1-7.
- 1.55 Gilsanz V, Boechat MI, Roe TF, Loro ML, Sayre JW, Goodman WG. Gender differnces in vertebral body size in children and adolscents. Radiology 1994;190:673-7.
- 1.56 Gilsanz V, Boechat MI, Gilsanz R, Loro ML, Roe TF, Goodman WG. Gender differences in vertebral sizes in adults: biomechanical implications. Radiology 1994;190:678-82.
- 1.57 Faulkner RA, McCulloch RG, Fyke SL et al. Comparison of areal and estimated volumetric bone mineral density values between older men and women. Osteoporosis Int 1995;5:271-5.
- 1.58 Smith DA, Nance WE, Won Kang K, Christian JC, Johnston CC. Genetic factors in determining bone mass. J Clin Invest 1973;52:2800-8.
- 1.59 Pocock NA, Eisman AJ, Hooper JL, Yeates MG, Sambrook PN, Eberl S. Genetic determinants of bone mass in adults: a twin study. J Clin Invest 1987;80:706-10.
- 1.60 Hansen MA. Assessment of age and risk factors on bone density and bone turnover in healthy premenopausal women. Osteoporosis Int 1994;4:123-8.
- 1.61 Mazess RB, Barden HS, Eberle RW, Denton M. Age changes of spine density in posterior-anterior and lateral projections in normal women. Calcif Tissue Int 1995;56:201-5.

243

<u>na ana amin'ny sorana amin'ny sorana dia kaodim-paositra dia kaodim-paositra dia kaodim-paositra dia kaodim-pao</u>

- 1.62 Matkovic V, Jelic T, Wardlaw GM et al. Timing of peak bone mass in Caucasian females and its implication for the prevention of osteoporosis. Inference from a crosssectional model. J Clin Invest 1994;93:799-808.
- 1.63 Jelic T, Wardlaw GM, Ilich JZ et al. Timing of peak bone mass in causasian females. J Bone Miner Res 1992;7(1):187.
- 1.64 Smith DAS, Hosie CJ, Deacon AD, Hamblen DL. Quantitative γ -ray computed tomography of the radius in normal subjects and osteoporotic patients. Br J Radiol 1990;63:776-82.
- 1.65 Slosman DO, Rizzoli R, Pichard C, Donath A, Bonjour JP. Longitudinal measurements of regional and whole body bone mass in young healthy adults. Osteoporosis Int 1994;4:185-90.
- 1.66 Dempster DW, Lindsay R. Pathogenesis of osteoporosis. Lancet 1993;341:797-805.
- 1.67 Riggs BL, Melton LJ III. Evidence for two distinct syndromes of involutional osteoporosis. Am J Med 1983; 75:899-901.
- 1.68 Pocock NA, Eberl S, Eisman JA et al. Dual photon absorptiometry in normal Australian women: a crosssectional study. Med J Aust 1987;146:293-7.
- 1.69 Riggs BL, Wahner HW, Dunn WL, Mazess RB, Offord KP, Melton LJ III. Differential changes in bone mineral density of the appendicular and axial skeleton with ageing. J Clin Invest 1980;67:328-35.
- 1.70 Riggs BL, Wahner HW, Melton LJ III, Richelson LS, Judd HL, Offord KP. Rates of bone loss in the appendicular and axial skeletons of women. Evidence of substantial vertebral bone loss before the menopause. J Clin Invest 1986;77:1487-91.
- 1.71 Krolner B, pors Nielsen S. Bone mineral content of the lumbar spine in normal and osteoporotic women: crosssectional and longitudinal studies. Clin Sci 1982;62:329-36.

- 1.72 Schaadt O, Bohr H. Different trends in age-related diminution of bone mineral content in the lumbar spine, femoral neck and femoral shaft. Calcif Tissue Int 1988; 42:71-6.
- 1.73 Hansen MA, Overgaard K, Christiansen C. Spontaneous postmenopausal bone loss in different skeletal areas followed up for 15 years. J Bone Miner Res 1995;10:205-10.
- 1.74 Block LE, Smith R, Gluer CC, Steiger P, Ettinger B, Genant HK. Models of spinal trabecular bone loss as determined by quantitative computed tomography. J Bone Miner Res 1989;4:249-57.
- 1.75 Ruegsegger P, Durand E, Dambacher MA. Localization of regional forearm bone loss from high resolution computed tomographic images. Osteoporosis Int 1991;1:76-80.
- 1.76 Ruegsegger P, Durand EP, Dambacher MA. Differential effects of aging and disease of trabecular annd compact bone density of the radius. Bone 1991;12:99-105.
- 1.77 Nilas L, Christiansen C. Rates of bone loss in normal women: evidence of accelerated trabecular bone loss after the menopause. Eur J Clin Invest 1988;18:529-34.
- 1.78 Ruegsegger P, Dambacher MA, Ruegsegger E, Fischer JA, Anliker M. Bone loss in premenopausal and postmenopausal women. J Bone Joint Surg 1984;66A:1015-23.
- 1.79 Recker RR, Lappe JM, Davies M, Kimmel DB. Changes in bone mass immediately before the menopause. J Bone Miner Res 1992;7:857-62.
- 1.80 Sowers MR, Clark MK, Hollis B, Wallace RB, Jannausch M. Radial bone mineral density in pre and perimenopausal women: a prospective study of rates and risk factors for loss. J Bone Miner Res 1992;7:647-57.
- 1.81 Szejnfeld VL, Atra E, Baracat EC, Aldrighi JM, Civetelli R. Bone density in white Brazilian women: rapid loss at the time around the menopause. Calcif Tissue Int 1995;56:186-91.

March & Spill and share a second of the second states of the second second

أجهدت

- 1.82 Baran D, Sorensin A, Grimes J et al. Dietary modification with dairy products for preventing vertebral bone loss in premenopausal women: a three year prospective study. J Clin Endocrinol Metab 1990;70:264-70.
- 1.83 Prior JC, Vigna YM, Barr SI, Kennedy S, Schulzer M, Li DK. Ovulatory premenopausal women lose cancellous spinal bone: five year prospective study. Bone 1996;18:261-7.
- 1.84 Baran DT. Magnitude and determinants of premenopausal bone loss. Osteoporosis Int 1994;4 (Suppl 1):31-4.
- 1.85 Haddaway MJ, Davie MWJ, HcCall IW. Bone mineral density in healthy normal women and reproducibility of measurements in spine and hip using dual energy x-ray absorptiometry. Br J Radiol 1992;65:213-7.
- 1.86 Kanis JA, Adami S. Bone loss in the elderly. Osteoporosis Int 1994;4 (Suppl 1):59-65
- 1.87 Parfitt AM. Interpretation of bone densitometry measurements: disadvantages of a percentage scale and a discussion of some alternatives. J Bone Miner Res 1990;5:537-40.
- 1.88 Peel N, Eastell R. Measurement of bone mass and turnover. In: Reid DM, editor. Osteoporosis. Baillieres Clin Rheumatol 1993;7(3)479-98.
- 1.89 Schapira D, Shapira C. Osteoporosis: the evolution of a scientific term. Osteoporosis Int 1992;2:164-7.
- 1.90 Nordin BEC. The definition and diagnosis of osteoporosis. Calcif Tissue Int 1987;40:57-8.
- 1.91 Concensus Development Conference. Prophylaxis and treatment of osteoporosis. Osteoporosis Int 1991;1:114-7.
- 1.92 World Health Organisation. Assessment of osteoporotic fracture risk and its role in screeing for postmenopausal osteoporosis. Kanis JA. World Health Organisation, Geneva, 1994.
- 1.93 Melton LJ. How many women have osteoporosis now? J Bone Miner Res 1995;10:175-7.

(2) Service Service Report and the service of th

the is a

1.94 Melton LJ III. Epidemiology of fractures. In: BL Riggs, LJ Melton III, editors. Osteoporosis: Etiology, Diagnosis, and Management. New York: Raven Press, 1988:133-54.

the second for

- 1.95 Cooper C. Epidemiology and public health impact of osteoporosis. In: Reid DM, editor. Osteoporosis. Baillieres Clin Rheumatol 1993;7(3):459-77.
- 1.96 Melton LJ, Chrischilles EA, Cooper C, Lane AW, Riggs BL. How many women have osteoporosis? J Bone Miner Res 1992;7:1005-10.
- 1.97 Kanis JA, McCloskey EV. Epidemiology of vertebral osteoporosis. Bone 1992;13:S1-S10.
- 1.98 Kannus P, Parkkari J, Sievanen H, Heinonen A, Vuori I, Jarvinen M. Epidemiology of hip fractures. Bone 1996;18(Suppl 1):57S-63S.
- 1.99 Maggi S, Kelsey JL, Litvak J, Heyse SP. Incidence of hip fractures in the elderly: a cross-sectional analysis. Osteoporosis Int 1991;1:232-41.
- 1.100 Kanis JA. The incidence of hip fracture in Europe. Osteoporosis Int 1993;3(Suppl 1):10-5.
- 1.101 Kanis JA and the WHO study group. Assessment of fracture risk and its application to screening for postmenopausal osteoporosis: a synopsis of the WHO report. Osteoporosis Int 1994;4:368-81.
- 1.102 Kenzora JE, McCarthy RE, Lowell JD, Sledge CB. Hip fracture mortality. Relationship to age, treatment, preoperative illness, time of surgery, and complications. Clin Orthop 1984;186:45-6.
- 1.103 Clayer MT, Banze RJ. Morbidity and mortality following fractures of the femoral neck and trochanteric regions: analysis of risk factors. J Trauma 1989;29:1673-8.
- 1.104 Davies KM, Recker R, Heaney RP. Normal vertebral dimensions and normal variation in serial measurements of vertebrae. J Bone Miner Res 1989;4:341-9.

247

and a standard management of the standard standard standard standard standards.

1.105 Melton LJ, Kan SH, Frye MA, Wahner HW, O'Fallon M, Riggs BL. Epidemiology of vertebral fractures in women. Am J Epidemiol 1989;129:1000-11.

1. 1. 1. 1. 1.

- 1.106 Black DM, Cummings SR, Stone K, Hudes E, Palermo L, Steiger P. A new approach to defining normal vertebral dimensions. J Bone Miner Res 1991;6:883-92.
- 1.107 Eastell R, Cedel SL, Wahner HW, Riggs BL, Melton LJ. Classification of vertebral fractures. J Bone Miner Res 1991;6:207-15.
- 1.108 Minne HW, Leidig G, Wuster C et al. A newly developed spine deformity index (SDI) to quantitate vertebral crush fractures in patients with osteoporosis. Bone Miner 1988;3:335-49.
- 1.109 Genant HK, Wu CY, Van Kuijk C, Nevitt MC. Vertebral fracture assessment using a semiquantitative technique. J Bone Miner Res 1993;8:1137-48.
- 1.110 Cooper C, Atkinson EJ, Jacobsen SJ, O'Fallon WM, Melton LJ. Survival following vertebral fractures: a populationbased study. J Bone Miner Res 1992;7 (Suppl 1):174.
- 1.111 Boyle IT. Secondary osteoporosis. In: Reid DM, editor. Osteoporosis. Baillieres Clin Rheumatol 1993;7(3): 515-34.
- 1.112 Dempster DW, Arlot MA, Meunier PJ. Mean wall thickness and formation periods of trabecular bone packets in corticosteroid induced osteoporosis. Calcif Tissue Int 1983;35:410-7.
- 1.113 Brescot C, Meunuer PJ, Chapuy MC, Lejeune E, Darby AJ. Histomorphometric profile, pathophisiology & reversibility of corticosteroid induced osteoporosis. Metab Bone Dis Rel Res 1979;1:303-11.
- 1.114 Wajchenberg BL, Pereira VG, Kieffer J, Uric S. Effects of dexamethasone on calcium metabolism and Ca kinetics in normal subjects. Acta Endocrinol(Copenh) 1969;61:173-92.

248

enter a construction de tradicional de la construction de la construction de la construction de la construction

and the second second second

1.115 Suzuki Y, Ichikawa Y, Saito E, Homma M. Importance of increased calcium excretion in the development of secondary hyperparathyroidism of patients under glucocorticoid therapy. Metabolism 1983;32:151-6.

1.1.1.1.1.1.1

化乙酸乙酸乙酸乙酸乙酸乙酸乙酸乙酸乙酸乙酸乙酯

- 1.115 Eastell R. Management of corticosteroid-induced osteoporosis. J Intern Med 1995;237:439-47
- 1.117 Sakakura M, Takebe K, Nakagawa S. Inhibition of luteinising hormone secretion induced by synthetic LRH by long term treatment with glucocorticoids in human subjects. J Clin Endocrinol Metab 1975;40:774-9.
- 1.118 Hsueh AJ, Erickson GF. Glucocorticoid inhibition of FSHinduced oestrogen production in cultured rat granulosa cells. Steroids 1978;32:639-48.
- 1.119 Crilly RG, Marshall DH, Nordin BEC. Metabolic effects of corticosteroid therapy in post-menopausal women. J Steroid Biochem 1979;11:429-33.
- 1.120 MacAdams MR, White RH, Chipps BE. Reduction in serum testosterone levels during chronic glucocorticoid therapy. Ann Intern Med 1986;104:648-51.
- 1.121 Danisova A, Sebokova E, Kolena J. Effect of corticosteroids on oestradiol and testosterone secretion by granulosa cells in culture. Exp Clin Endocrinol 1987;89:165-73.
- 1.122 Sambrook PN, Eisman JA, Champion GD, Pocock NA. Sex hormone status and osteoporosis in post-menopausal women with rheumatoid arthritis. Arthritis Rheum 1988;31:973-8.
- 1.123 Sambrook PN, Birmingham J, Champion GD, Kelly F et al. Post-menopausal bone loss in Rheumatoid Arthritis: Effect of oestrogens and androgens. J Rheumatol 1992;19:357-61.
- 1.124 Crilly RG, Cawood M, Marshall DH, Nordin BE. Hormonal status in normal, osteoporotic and corticosteroid treated post-menopausal women. J R Soc Med 1978;71:733-6.

Charles and the proceeding

20. 18. 20. 16. 1

الهوري المراجع أربعهم والأفريج ومراجع والمراجع والمراجع والمراجع والمراجع والمراجع

1.125 Seeman E, Kumar R, Hunder GG, Scott M, Heath H, Riggs BL. Production, degradation and circulating levels of 1,25dihydroxyvitamin D in health and chronic glucocorticoid excess. J Clin Invest 1980;66:664-9.

1.77

しょうかい とうめいとうかい かいかた ひつうり しかいそのがく 読み

(a) A₂ = 5 A₂

- 1.126 Colette C, Monnier L, Pares Herbute N, Blotman F, Mirouze J. Calcium absorption in corticoid treated subjects: Effects of a single oral dose of calcitrol. Horm Metab Res 1987;19:335-8.
- 1.127 Luckert BP, Adams JS. Calcium and phosphorus homeostasis in man: effects of corticosteroids. Arch Intern Med 1976;136:124-53.
- 1.128 Gennari C, Civitelli R. Glucocorticoid-induced osteoporosis. Clin Rheum Dis 1986;12:637-54.
- 1.129 Reid IR, Heap SW. Determinants of vertebral mineral density in patients receiving long-term glucocorticoid therapy. Arch Intern Med 1990;150:2545-8.
- 1.130 Sambrook PN, Kelly PJ, Keogh A et al. Bone loss after cardiac transplantation: a prospective study. J Heart Lung Transpl 1994;13:116-21.
- 1.131 Garton MJ, Reid DM. Bone mineral density of the hip and anteroposterior and lateral dimensions of the spine in men with rheumatoid arthritis: effects of low-dose corticosteroids. Arthritis Rheum 1993;36:222-8.
- 1.132 Baylink DJ. Glucocorticoid-induced osteoporosis. N Eng J Med 1983;309:306-8.
- 1.133 Adinoff AD, Hollister JR. Steroid induced fractures and bone loss in patients with asthma. N Eng J Med 1983;309:265-8.
- 1.134 Sambrook P, Birmingham J, Kempler S et al. Corticosteroid effects on proximal femur bone loss. J Bone Miner Res 1990;12 1211-6.
- 1.135 Sambrook PN, Jones G. Corticosteroid osteoporosis. Br J Rheumatol 1995;34:8-12.

250

- 1.136 LoCascio V, Bonucci E, Imbimbo et al. Bone loss after glucocorticoid therapy. Calcif Tissue Int 1984;36:435-43.
- 1.137 Montemurro L, Fraioli P, Riboldi A, Delpiano S, Zanni D, Rizzato G. Bone loss in prednisolone treated sarcoidosis: a two year follow up. Ann Ital Med Int 1990;5:164-8.
- 1.138 de Deuxchaisnes CN, Devogelaer JP, Esselinckx W et al. The effect of low dose glucocorticoids on bone mass in rheumatoid arthritis: a cross-sectional and longitudinal study using single photon absorptiometry. Adv Exp Med Biol 1984;171:210-39.
- 1.139 Ruegsegger P, Medici TC, Anliker M. Corticosteroidinduced bone loss. A longitudinal study of alternate day therapy in patients with bronchial asthma using quantitative computerised tomography. Eur J Clin Pharmacol 1983;25:615-20.
- 1.140 Avioli L. Osteoporosis in rheumatoid arthritis. Arthritis Rheum 1987;30:830-1.
- 1.141 Reid DM, Nicoll JJ, Smith MA, Higgins B, Tothill P, Nuki G. Corticosteroids and bone mass in asthma: comparisons with rheumatoid arthritis and polymyalgia rheumatica. Br Med J 1986;293:1463-6.
- 1.142 Packe GE, Douglas JG, McDonald A, Robins SP, Reid DM. Bone density in asthmatic patients taking high dose inhaled beclomethasone dipropionate and intermittent systemic steroids. Thorax 1992;47:414-7.
- 1.143 Medici TC, Ruegsegger P. Does alternate-day cloprednol therapy prevent bone-loss? A longitudinal double blind, controlled clinical study. Clin Pharmacol Ther 1990;48:455-66.
- 1.144 American College of Rheumatology Task Force on Osteoporosis Guidelines. Recommendations for the prevention and treatment of glucocorticoid-induced osteoporosis. Arthritis Rheum 1996;1791-1801.

والمراجع والأوراق والمراجع والمنابع والمتراجع المتراجع والمرور المتحاد المتحاد المراجع والمراجع المراجع المراجع

CONTRACT STREET, NO. 1

1.145 Compston JE, Cooper C, Kanis JA. Bone densitometry in clinical practice. Br Med J 1995;310:1507-10.

the second second second second

- 1.146 Kanis JA, Devogelaer JP, Gennari C. Practical guide for the the use of bone mineral density measurements and assessment of treatment of osteoporosis. Osteoporosis Int 1996;6:256-61.
- 1.147 Sartoris DJ, Resnick D. Current and innovative methods for noninvasive bone densitometry. Radiol Clin North Am 1990;28:257-78.
- 1.148 Wilson CR, Fogleman I. Densitometry and bone mineral measurement. Curr Opin Rheumatol 1990;2:369-78.
- 1.149 Ruegsegger P. The use of peripheral QCT in the evaluation of bone remodelling. Endocrinologist 1994;4:167-76.
- 1.150 Gluer CC, Engelke K, Lang TF, Grampp S, Genant HK. Quantitative computed tomography (QCT) of the lumbar spine and appendicular skeleton. Eur J Radiol 1995;20:173-8.
- 1.151 van Daele PLA, Burger H, De Laet CEDH, Pois HAP. Ultrasound measurement of bone. Clin Endocrinol 1996;44:363-9.
- 1.152 Majumdar S, Genant HK. A review of the recent advances in magnetic resonance imaging in the assessment of osteoporosis. Osteoporosis Int 1995;5:79-92.
- 1.153 Lang P, Steiger P, Faulkner K, Gluer C, Genant HK. Current techniques and recent developments in quantitative bone densitometry. Radiol Clin North Am 1991;29:49-76.
- 1.154 Jergas M, Genant HK. Current methods and recent advances in the diagnosis of osteoporosis. Arthritis Rheum 1993;36:1649-62.
- 1.155 Cummings SR, Kelsey JL, Nevitt MC, O'Dowed KJ. Epidemiology of osteoporosis and osteoporotic fractures. Epidemiol Rev 1985;7:178-208.
- 1.156 Garton MJ, Robertson EM, Gilbert FJ, Gomersall L, Reid DM. Can radiologists detect osteopenia on plain radiographs? Clin Radiol 1994;49:118-22.

252

1.157 Singh M, Nagrath AR, Maini PS. Changes in trabecular pattern of the upper end of the femoral neck as a measure of osteopenia. J Bone Joint Surg 1970;52A:457-67.

- 44 C. S. C.

- 1.158 Koot VCM, Kesselaer SMMJ, Clevers GJ, de Hooge P, Weits T, van der Werken. Evaluation of the Singh index for measuring osteoporosis. J Bone Joint Surg 1996;78B:831-4.
- 1.159 Masud T, Jawed S, Doyle DV, Spector TD. A population study of the screening potential of assessment of trabecular pattern of the femoral neck (Singh index): the Chingford Study. Br J Radiol 1995;68:389-93.
- 1.160 Yates AJ, Ross PD, Lydick E, Epstein RS. Radiographic absorptiometry in the diagnosis of osteoporosis. Am J Med 1995;98(2A):418-478.
- 1.161 Cosman F, Herrington B, Himmelstein S, Lindsay R. Radiographic absorptiometry: a simple method for determination of bone mass. Osteoporosis Int 1990;2:34-8.
- 1.162 Ravn P, Overgaard K, Huang C, Ross PD, Green D, McClung M. Comparison of bone densitometry of the phalynges, distal forearm and axial skeleton in early postmenopausal women participating in the EPIC study. Osteoporosis Int 1996;6:308-13.
- 1.163 Adami S, Zamberlan N, Gatti D et al. Computed rediographic absorptiometry and morphometry in teh assessment os postmenopausal bone loss, Osteoporosis Int 1996;6:8-13.
- 1.164 Swezey R, Draper D, Swezey AM. Bone densitometry: comparison of dual energy x-ray absorptiometry to radiographic absorptiometry. J Rheumatol 1996;23:1734-8.
- 1.165 Wishart JM, Horowitz M, Bochner M, Need AG, Nordin BEC. Relationships between metacarpal morphometry, forearm and vertebral density and fractures in postmenopausal women. Br J Radiol 1993;66:435-40.

253

and the second second

and several and the second second

and the second states

1.166 Kuiper JW, van Kuijk C, Grashuis JL, Ederveen AGH, Schutte HE. Accuracy and the influence of marrow fat on quantitative CT and dual-energy x-ray absorptiometry measurements of the femoral neck in vitro. Osteoporosis Int 1996;6:25-30.

- 1.167 Hangartner TN, Johnston CC. Influence of fat on bone measurments with dual-energy absorptiometry. Bone Miner 1990;9:71-81.
- 1.168 Drinka PJ, DeSmet AA, Bauwens SF, Rogot A. The effect of overlying calcification on lumbar bone densitometry. Calcif Tissue Int 1992;50:507-10.
- L.169 Frye MA, Melton LJ III, Bryant SC et al. Osteoporosis and calcification of the aorta. Bone Miner 1992;19:185-94.
- 1.170 Hoshino H, Kushida R, Takahashi M, Ohishi T, Sugiyama E, Inoue T. The influence of aortic calcification on spinal bone mineral density in vitro. Calcif Tissue Int 1996;59;21-3.
- 1.171 Masud T, Langley S, Wiltshire P, Doyle DV, Spector TD. Effect of spinal osteophytosis on bone mineral density measurements in vertebral osteoporosis. Br Med J 1993;307:172-3.
- 1.172 Jones G, White C, Nguyen T, Sambrook PN, Kelly PJ, Eisman J. Prevelent vertbral deformities - relationship to bone mineral density and spinal osteophytosis in elderly men and women. Osteoporosis Int 1996;6:233-9.
- 1.173 Yu W, Gluer CC, Fuerst T, Grampp S, Li Y, Genant HK. Influence of degenerative joint disease on spinal bone mineral measurements in postmenopausal women. Calcif Tissue Int 1995;57:169-74.
- 1.174 Peel NFA, Barrington NA, Blumsohn A, Colwell A, Hannon R, Eastell R. Bone mineral density and bone turnover in spinal osteoarthritis. Ann Rheum Dis 1995;54:867-71.

254

化盐酸盐盐 化化化合物 化合物合物 网络拉拉人名 法公司法 化乙烯酸盐 化乙烯酸盐

and the second second

- 1.175 Peacock DJ, Egger P, Taylor P, Cawley MID, Cooper C. Lateral bone density measurements in osteoarthritis of the lumbar spine. Ann Rheum Dis 1996;55:196-8.
- 1.176 Yu W, Gluer CC, Fuerst T et al. Influence of degenerative joint disease on spinal bone mineral measurements in postmenopausal women. Calcif Tissue Int 1995;57:169-74.
- 1.177 von der Recke P, Hansen MA, Overgaard K, Christiansen C. The impact of degenerative conditions in the spine on bone mineral density and fracture risk prediction. Osteoporosis Int 1996;6:43-9.
- 1.178 Dequeker J, Mokassa L, Aerssens J. Bone density and OA. J Rheumatol 1995;22(suppl):98-100.
- 1.179 Nevitt MC, Lane NE, Scott JC, and the Study of Osteoporotic Fractures Research Group. Radiographic osteoarthritis of the hip and bone mineral density. Arthritis Rheum 1995;38:907.
- 1.180 Goh JCH, Low SL, Bose K. Effect of femoral rotation on bone mineral density measurements with dual energy x-ray absorptiometry. Calcif Tissue Int 1995;57:340-3.
- 1.181 Langton CM, Evans GP, Hodgkinson R, Riggs CM. Ultrasonic, elastic and structural properties of cancellous bone. In: Ring EFG, editor. Current Research in Osteoporosis and Bone Mineral Measurement. London: British Institute of Radiology, 1990:10-11.
- 1.182 Gluer CC, Wu CY, Genant HK. Broadband ultrasound attenuation signals depends on trabecular orientation. An in vitro study. Osteoporosis Int 1993;3:185-91.
- 1.103 Gluer CC, Wu CY, Jergas M, Goldstein SA, Genant HK. Three quantitative ultrasound parameters reflect bone structure. Calcif Tissue Int 1994;55:46-52.
- 1.184 Han S, Rho J, Medige J, Ziv I. Ultrasound velocity and broadband attenuation over a wide range of bone mineral density. Osteoporosis Int 1996;6:291-6.

the second second states and second second

PJ.185 Hans D, Arlot ME, Schott AM, Roux JP, Kotzki PO, Meunier PJ. Do ultrasound measurements on the os calcis reflect more the bone microarchitecture than the bone mass?: a two dimentional histomorphometric study. Bone 1995;16:295-300.

the second second second second

- 1.186 Brooke-Wavell K, Jones PR, Pye DW. Ultrasound and dual xray absorptiometry measurments of the calcaneus: influences of region of interest location. Calcif Tissue Int. 1995;57:20-4.
- 1.187 Petley GW, Roins PA, Aindow JD. Broadband ultrasonic attenuation: are current measurement techniques inherently inaccurate? Br J Radiol 1995;68:1212-14.
- 1.188 Wu CY, Gluer CC, Jergas M, Bendavid E, Genant HK. The impact of bone size on broadband ultrasound attenuation. Bone 1995;16:137-41
- 1.189 Ruegsegger P, Elsasser U, Anliker M, Gnehm H, Kind H, Prader A. Quantification of bone mineralization using computed tomography. Radiology 1976;121:93-7.
- 1.190 McClean BA, Overton TR, Hangartner TN, Rathee S. A special purpose x-ray fan-beam CT scanner for trabecular bone density in the appendicular skeleton. Phys Med Biol 1990;35:11-9.
- 1.191 Cummings SR, Black D. Bone mass measurements and risk of fracture in Caucasian women: a review of findings from prospective studies. Am J Med 1995;98(Suppl 2A):24S-28S.
- 1.192 Cummings SR, Black DM, Nevitt MC et al. Bone density at various sites for prediction of hip fracture Lancet 1993;341:72-5.
- 1.193 Mallmin H, Ljunghall S. Distal radius fracture is an early sign of general osteoporosis: bone mass measurements in a population-based study. Osteoporosis Int 1994;4:357-61.
- 1.194 School of Public Health. Effective Heath Care: Screening for osteoporosis to prevent fractures. School of Public Health, 1992 (No 1).

256

3.47

al an an an ann a bhair a bha ann a she ann a chann an ann an an an an tha ann an an ann ann ann an 1875. An a

1.195 Torgerson DJ, Garton MJ, Donaldson C, Russell IT, Reid DM. Recruitment methods for screening programmes: trial of an improved method within a regional osteoporosis study. Br Med J 1993;307:99.

1.1

- 1.196 Garton MJ, Torgerson DJ, Donaldson C, Russell IT, Reid DM. Recruitment methods for screening programmes: trial of a new method within a regional osteoporosis study. Br Med J 1992;305:82-4.
- 1.197 Black DM. Why elderly women should be screened and treated to prevent osteoporosis. Am J Med 1995;98(2A)67S-75S.
- 1.198 Maxim P, Ettinger B, Spitalny GM. Fracture protection provided by long-term estrogen treatment. Osteoporosis Int 1995;5:23-9.
- 1.199 Lufkin EG, Wahner HW, O'Fallon WM et al. Treatment of postmenopausal osteoporosis with transdermal estrogen. Ann Int Med 1992;117;1-9.
- 1.200 Larcos G. What factors influence general practitioners commencement of hormone replacement in perimenopausal women. Br J Clin Pract 1996;50:6-8.
- 1.201 Rubins SM, Cummings SR. Results of bone densitometry affects women's decisions about taking measures to prevent fractures. Ann Intern Med 1992;116:990-5.
- 1.202 Ringa V, Legesert B, Breart G. Determinants of hormone replacement therapy among postmenopausal women enrolled in the French GAZEL cohort. Osteoporosis Int 1994;4:16-20.
- 1.203 Kroger H, Huopio J, Honkanen R et al. Prediction of fracture risk using axial bone mineral density in a perimenopausal population: a prospective study. J Bone Miner Res 1995;10:302-6.
- 1.204 Stewart A, Torgerson DJ, Reid DM. Prediction of fractures in perimenopausal women: a comparison of dual energy x-ray absorptiometry and broad band ultrasound attenuation. Ann Rheum Dis 1996;55:140-2.

257

<u>na ana amin'ny solana dia kaodim-paositra dia kaodim-paositra dia kaodim-paositra dia kaodim-paositra dia kaodi</u>

- 1.205 Earnshaw SA, Hosking DJ. Clinical usefulness of risk factors for osteoporosis. Ann Rheum Dis 1996;55:338-9.
- 1.206 Raisz LG. Assessment of the risk of osteoporosis at the menopause: therapeutic consequences. Osteoporosis Int 1994;4 (Suppl 1):53-7.
- 1.207 Ribot C, Tremollieres F, Pouilles JM. Can we detest women with low bone mass using clinical risk factors. Am J Med 1995;98(Suppl 2A):528-558.
- 1.208 Kleerekoper M, Peterson E, Nelson D et al. Identification of women at risk for developing postmenopausal osteoporosis with vertebral fractures: role of history and single photon absorptiometry. Bone Miner 1989;7:171-86.
- 1.209 Cooper C, Shah S, Hand DJ et al. Screeing for vertebral osteoporosis using individual risk factors. Osteoporosis Int 1991;1:48-53.
- 2.1 Abbaszadegen H, Adolphson P, Dalen N, Jonsson U, Sjoberg H, Kalen S. Bone mineral loss after Colles' fracture. Acta Orthop Scand 1991;62:156-8.
- 2.2 Massie A, Reid DM, Porter RW. Screening for osteoporosis: Comparison between dual energy X-ray absorptiometry and broadband ultrasound attenuation in 1000 perimenopausal women. Osteoporosis Int 1993;3:107-10.
- 2.3 Tothill P, Avenell A, Reid DM. Precision and accuracy of measurements of whole-body mineral: comparison between Hologic, Lunar and Noralnd dual-energy X-ray absorptiometers. Br J Radiol 1994;67:1210-7.
- 3.1 Grampp S, Lang P, Jergas M et al. Assessment of the skeletal status by peripheral quantitative computed tomography of the forearm: short-term precision in vivo and comparison to dual x-ray absorptiometry. J Bone Miner Res 1995;10:1566-76.
- 3.2 Bland M. An introduction to medical statistics. Oxford: Oxford Medical Publications, 1990.

South States and the second states and states and states and states

4.1 Gutin B, Kasper MJ. Can vigorous exercise play a role in osteoporosis prevention? A review. Osteoporosis Int 1992;2:55-69.

医马克氏 化二氟基苯基乙基 医乙酰氨酸乙基乙基

- 4.2 Mauhan R, Abel R, Watson J, Weir J. Forearm composition and muscle function in trained and untrained limbs. Clin Physiol 1986;6:389-96.
- 4.3 Pirnay F, Bodeux M, Crielaard J, Franchimont P. Bone mineral content and physical activity. Int J Sports Med 1987;8:331-5.
- 4.4 Huddleston AL, Rockwell D, Kulund D, Harrison B. Bone mass in lifetime tennis athletes. JAMA 1980;244:1107-9.
- 4.5 Jacobson P, Beaver W, Grubb S, Taft T, Talmage R. Bone density in women: college athletes and older athletic women. J Orthop Res 1984;2:328-32.
- 4.6 Sandler R, Cauley J, Hom D, Sashin D, Kriska A. The effects of walking on the cross-sectional dimensions of the radius in postmenopausal women. Calcif Tissue Int 1987;41:65-9.
- 4.7 Ayalon J, Simkin A, Leichter I, Raifmann S. Dynamic bone loading exercises for postmenopausal women: effect on density of the distal radius. Arch Phys Med Rehabil 1987;68:280-3.
- 4.8 Simkin A, Ayalon J, Leichter I. Increased trabecular bone density due to bone-loading exercises in postmenopausal osteoporotic women. Calcif Tissue Int 1987;40:59-63.
- 4.9 Beverly M, Rider T, Evans M, Smith R. Local bone mineral response to brief exercise that stresses the skeleton. Br Med J 1989;299:233-5.
- 4.10 Rikli R, McManus B. Effects of exercise on bone mineral content in postmenopausal women. Res Q Exerc Sport 1990;61:243-9.

259

- 4.11 Rico H, Gonzales-Roila J, Revilla M, Villa LF, Gomez-Castresana F, Escribano J. Cortical versus trabecular bone mass: influence of activity on both bone compartments. Calcif Tissue Int 1994;54:470-2.
- 4.12 Faulkner KG, Genant HK, McClung M. Bilateral comparison of femoral bone density and hip axis length from single and fan beam DXA scans. Calcif Tissue Int 1995;56:26-31.
- 4.13 Bonnick SL, Nichols DL, Sanborn CF, Payne SG, Moen SM, Heiss CJ. Right and left proximal femur analyses: is there a need to do both? Calcif Tissue Int 1996;58:307-10.
- 4.14 Dretakis E, Damilakis J, Kontakis G, Gourtsoyiannis N. Variation between os calces as measured by broadband ultrasound attenuation. Br J Radiol 1994;67:636~8.
- 5.1 Mazess RB, Barden HS, Ettinger M et al. Spine and femur density using dual photon adsorptiometry in US white women. Bone Miner 1987;2:211-9.
- 5.2 Hedlund LR, Gallager JC. The effect of age and menopause on bone mineral density of the proximal femur. J Bone Miner Res 1989;4:639-42.
- 5.3 Harris S, Dawson-Hughes B. Rates of change in bone mineral density of the spine, heel, femoral neck and radius in healthy postmenopausal women. Bone Miner 1992;17:87-95.
- 5.4 Silman A. European Vertebral Osteoporosis Study: an overview. In: EFJ Ring, DM Elvins, AK Bhalla, editors. Current research in osteoporosis and bone mineral measurement III: London, British Institute of Radiology, 1994:16
- 5.5 Reeve J, Kroger H, Nijs J et al. Radial cortical and trabecular bone densities of men and women standardized with the European forearm phantom. Calcif Tissue Int 1996;58:135-43.

the state of the second se

5.6 Dequeker J, Pearson J, Reeve J et al. Dual x-ray absorptiometry: cross-calibration and normative reference ranges for the spine. Results of a European Community Concerned Action. Bone 1995;17:247-54.

and the second sec

1124 012

- 5.7 Simmons A, Barrington S, O'Doherty MJ, Cakley AJ. Dual energy X-ray absorptiometry normal reference range use within the UK and the effect of different normal ranges on the assessment of bone density. Br J Radiol 1995;68:903-9.
- 5.8 Pearson J, Dequeker J, Reeve J et al. Dual x-ray absorptiometry of the proximal femur: normal European values standardised with the European Spine Phantom. J Bone Miner Res 1994;10:315-24.
- 5.9 Lehmann R, Wapniarz M, Randerath O et al. Dual energy Xray absorptiometry at the lumbar spine in German men and women: a cross-sectional study. Calcif Tissue Int 1995;56:350-4.
- 5.10 Petley GW, Cotton AM, Murrills AJ et al. Reference ranges of bone density for women in southern England: the impact of local data on the daignosis of osteoporosis. Br J Radiol 1996;69:655-60.
- 5.11 Parra-Cabrera S, Hernandez-Avila M, Tamayo-y-Orozco J, Lopez-Carrillo L, Meneses-Gonzalez F. Exercise and reproductive factors as predictors of bone density among osteoporotic women in Mexico City. Calcif Tissue Int 1996;59:89-94.
- 5.12 Michaelsson K, Bergstrom R, Mallmin H, Holmberg L, Wolk A, Ljunghall S. Screening for osteopenia and osteoporosis: selection by body composition. Osteoporosis Int 1996;6:120-6.
- 5.13 Pocock NA, Eisman JA, Yeates MG, Sambrook PN, Eberl S. Physical fitness is a major determinant of femoral neck and lumbar spine bone mineral density. J Clin Invest 1986;78:618-21.

261

<u>na ang s</u>ana sa bang <u>na</u> sali

5.14 Davis JW, Ross PD, Wasnich RD. Relation of height and weight to the regional variations in bone mass among Japanese-American men and women. Osteoporosis Int 1995;5:234-8.

in the second state of the

- 6.1 Lai K, Rencken M, Dinkwater BL, Chestnut III CH. Site of bone density measurement may affect therapy decision Calcif Tissue Int 1993;53:225-8.
- 6.2 Pouilles JM, Tremollieres F, Ribot C. Spine and femur densitometry at the menopause: Are both sites necessary in the assessment of the risk of osteoporosis. Calcif Tissue Int 1993;52:344-7.
- 6.3 Greenspan SL, Maitland-Ramsey, Myers E. Classification of osteoporosis in the elderly is dependent on site specific analysis. Calcif Tissue Int 1996;58:409-14.
- 6.4 Kanis JA and the WHO study group. Assessment of fracture risk and its application to screening for postmenopausal osteoporosis: a synopsis of the WHO report. Osteoporosis Int 1994;4:368-81.
- 6.5 Wasnich RD, Ross PD, Davis JW, Vogel JM. A comparison of single versus multiple BMC measurements for assessment of spine fracture probability. J Nucl Med 1989;30:1166-71.
- 6.6 Gatti D, Rossini M, Zamberlan N, Braga V, Fracassi E, Adami S. Effect of aging on trabecular and compact bone components of proximal and ultradistal radius. Osteoporosis Int 1996;6:355-60.
- 6.7 Schneider P, Borner W, Mazess RB, Barden H. The relationship of peripheral to axial bone density. Bone Miner 1988;4:279-87.
- 6.8 Hosie CJ, Smith DA, Deacon AD, Langton CM. Comparison of broadband ultrasound attenuation of the os calcis and quantitative computed tomography of the distal radius. Clin Phys Physiol Meas 1987;8:303-8.

262

<u>ne na seconda da la construcción estas da construcción da construcción da construcción da construcción da cons</u>
6.9 Ross P, Genant HK, Davis J, Miller P, Wasnich R. Predicting vertebral fracture incidence from prevalent fractures and bone density among non-black, osteoporotic women. Osteoporosis Int 1993;3:120-6.

المحاجب والمراجع فالمعاصفة والمحاج والمحاج المحاج والمحاج المتعا

1.1.4

- 6.10 Ross PD, Davis JW, Epstein RS, Wasnich RD. Pre-existing factors and bone mass predictvertebral fracture incidence in women. Ann Intern Med 1991;114:919-23.
- 7.1 Hassager C, Christiansen C. Measurement of bone mineral density. Calcif Tissue Int. 1995;57:1-5.
- 7.2 Fujii Y, Chikawa T, Nakamura T, Goto B, Fujita T. Comparison of trabecular bone density at vertebral and radial sites using quantitative computed tomography. Osteoporosis Int 1996;6:486-90.
- 7.3 Garton M, Martin J, New S et al. Bone mass and metabolism in women aged 45-55. Clin Endocrinol 1996;44:563-70.
- 7.4 Bjarnason K, Hassager C, Ravan P, Christiansen C. Early posmenopausal diminution of forearm and spinal bone mineral density: a cross-sectional study. Osteoporosis Int 1995;5:35-8.
- 7.5 Takagi Y, Fugii Y, Miyauchi A, Goto B, Takakashi K, Fujita T. Transmenopausal changes of trabecular bone density and structural pattern assessed by peripheral quantitative computed tomography in Japanese women. J Bone Miner Res 1995;10:1830-4.
- 8.1 Kanis JA, McCloskey EV. Epidemiology of vertebral osteoporosis. Bone 1992;13:S1-S10.
- 8.2 Cooper C, Melton LJ. Vertebral fracture: How large is the silent epidemic? Br Med J 1992;304:793-4.
- 8.3 Gardsell P, Johnell O, Nilsson BE, Gullberg B. Predicting various fragility fractures in women by forearm bone densitometry: a follow-up study. Calcif Tissue Int 1993;52:348-53.
- 8.4 Cummings SR. Are patients with hip fractures more osteoporotic? Am J Med 1985;78:487-94.

263

Charles and the second second second second second of the second second second second second second second second

- 8.5 Sugimoto T, Kanbara Y, Shiraishi H et al. Femoral and spinal bone mineral density in Japanese osteoporotics with hip fracture. Osteoporosis Int 1994;4:144-8.
- 8.6 Firooznia H, Golimbu C, Rafii M, Schwartz MS, Alterman ER. Quantitative computed tomography assessment of spinal trabecular bone II. In osteoporotic women with and without vertebral fractures. J Comput Asst Tomogr 1984;8:99-103.
- 8.7 Bjarnason K, Nilas L, Hassager C, Christiansen C. Dual energy X-ray absorptiometry of the spine - decubitus lateral versus anteroposterior projection in osteoporotic women: comparison to single x-ray absorptiometry of the forearm. Bone 1995;16:255-60.
- 8.8 Stewart A, Felsenberg D, Kalidis L, Reid DM. Vertebral fractures in men and women: how discriminative are bone mass measurements? Br J Radiol 1995;68:614-20.
- 8.9 Duboeuf F, Jergas M, Schott A, Wu CY, Gluer C, Genant HK. A comparison of bone densitometry measurements of the central skeleton in postmenopausal women with and without vertebral fracture. Br J Radiol 1995;68:747-53.
- 8.10 Heuck A, Block J, Gluer CC, Steiger P, Genant HK. Mild versus definate osteoporosis: Comparison of bone densitometry techniques using different statistical models. J Bone Miner Res 1989;4:891-900.
- 8.11 Melton LJ, Atkinson EJ, O'Fallon M, Wahner HW, Riggs BL. Long-term fracture prediction by bone mineral assessed at differenct skeletal sites. J Bone Miner Res 1993;8:1227-33.
- 8.12 Schott AM, Weill-Engerer S, Hans D, du Boeuf F, Delmas PD, Meunier PJ. Ultrasound discriminates patients with hip fracture equally well as dual energy X-ray absorptiometry and independently of bone mineral density, J Bone Miner Res 1995;10:243-9.

sector(a)

aran ista in el ma

8.13 Stewart A, Reid DM, Porter RW. Broadband ultrasound attenuation (BUA) and dual energy x-ray absorptiometry (DXA) in women with hip fracture: which technique discriminates fracture risk? Calcif Tissue Int 1994;54: 466-9.

- 8.14 Turner CH, Peacock M, Timmerman L, Neal JM, Johnson CC. Calcaneal ultrasonic measurements discriminate hip fracture independently of bone mass. Osteoporosis Int 1995;5:130-5.
- 8.15 Mautalen C, Vega E, Gonzalez D, Carrilero P, Otano A, Silberman F. Ultrasound and dual x-ray absorptiometry densitometry in women with hip fracture. Calcif Tissue Int 1995;57:165-8.
- 8.16 Porter RW, Miller CG, Grainger D, Palmer SB. Prediction of hip fracture in elderly women: a prospective study. Br Med J 1990;301:638-41.
- 8.17 Bauer DC, Gluer CC, Pressman AR et al. Braodband ultrasonic attenution (BUA) and the risk of fracture: a prospective study. J Bone Miner Res 1995;10:S175.
- 8.18 Hans D, Dargent P, Schott AM et al. Ultrasound parameters predict hip fracture independently of hip bone density: the EPIDOS prospective study. J Bone Miner Res 1995;10:S169.
- 8.19 Hans D, Dargent-Molina P, Schott AM et al. Ultrasonographic heel measurements to predict hip fracture in elderly women: the EPIDOS prospective study. Lancet 1996;348:511-4.
- 8.20 Gluer CC, Cummings SR, Bauer DC et al. Osteoporosis: Association of recent fractures with quantitative US findings. Radiology 1996;199:725-32.
- 8.21 Bauer DC, Gluer CC, Genant HK, Stone K. Quantitative ultrasound and vertebral fracture in postmenopausal women. J Bone Miner Res 1995;10:353-8.

265

en en en en en en en de la ser de la ser

8.22 Ross P, Huang C, Davis J et al. Predicting vertebral deformity using bone densitometry at various skeletal sites and calcaneus ultrasound. Bone 1995;16:325-32.

「「「「」」」を見ていていた。 そう

- 8.23 Genant HK. Universal standardization for dual x-ray absorptiometry: patient and phantom cross-calibration results. J Bone Miner Res 1995;10:997-8.
- 8.24 Peacock M, Turner CH, Liu G, Manatunga AK, Timmerman L, Johnston CC. Better discrimination of hip fracture using bone density, geometry and architecture. Osteoporosis Int 1995;5:167-73.
- 8.25 Cummings SR, Cauley JA, Palermo L et al. Racial differences in hip axis lengths might explain racial differences in hip fracture. Study of Osteoporosis Fractures Research. Osteoporosis Int 1994:4;226-9.
- 8.26 Faulkner KG, Cummings SR, Black D, Palermo L, Gluer CC, Genant HK. Simple measurement of femoral geometry predicts hip fracture: the study of osteoporotic fractures. J Bone Miner Res 1993:8;1211-7.
- 8.27 Mikail MB, Vaswani AN, Aloia JF. Racial differences in femoral dimentions and their relation to hip fracture. Osteoporosis Int 1996;6:22-4.
- 8.28 Karlsson KM, Sernbo T, Obrant KJ, Redlund-Johnell I, Johnell O. Femoral neck geometry and radiographic signs of osteoporosis as predictors of hip fracture. Bone 1996;18:327-30.
- 8.29 Slemenda CW, Turner CH, Peacock M et al. The genetics of proximal femur geometry, distribution of bone mass and bone mineral density. Osteoporosis Int 1996;6:178-82.
- 8.30 Berry E, Truscott JG, Stewart SP, Smith MA. Spatial distribution of femoral bone mineral in dual energy x-ray absorptiometry images: a possible technique to improve discrimination between normal and osteoporotic patients. Br J Radiol 1996;69:743-50.

266

- 8.31 Kanis JA, McCloskey EV. Evaluation of the risk of hip fracture. Bone 1996;18(Suppl 3):1278-1328.
- 8.32 Cummings SR, Nevitt MC. Non-skeletal determinants of fractures: the potential importance of the mechanics of falls. Study of Osteoporosis Fractures Research. Osteoporosis Int 1994;4(Suppl 1):67-70.
- 8.33 Pinilla TP, Boardman KC, Bouxsein ML, Myers ER, Hayes WC. Impact direction from a fall infuences the failure load of the proximal femur as much as age-related bone loss. Calcif Tissue Int 1996;58:231-5.
- 8.34 Gilsanz V, Loro ML, Roe TF, Sayre J, Gilsanz R, Schulz EE. Vertebral size in elderly women with osteoporosis. Mechanical implications and relationship to fractures. J Clin Invest 1995;95:2332-7.
- 8.35 Mundinger A, Wiesmeier B, Dinkel E, Helwig A, Beck A, Schulte Moenting J. Quantitative image analysis of vertebral body architiecture - improved diagnosis in osteoporosis based on high resolution computed tomography. Br J Radiol 1993,66:209-13.
- 8.36 Schneider P, Borner W. Ability of peripheral QCT at the radius in comparison to DEXA of the spine to diagnose vertebral fractures in postmenopausal women. In: Christiansen CC, Overgaard K, editors. Osteoporosis: Etiology, Diagnosis, and Management. New York: Raven Press, 1990:921-4.
- 8.37 Grampp S, Jergas M, Lang P et al. Quantitative CT assessment of the lumbar spine and radius in patients with osteoporosis. AJR 1996;167:133-40.
- 8.38 Ruegsegger P, Anliker M, Dambacher M. Quantification of trabecular bone with low dose computed tomography. J Comput Assist Tomogr 1981;5:384-90.
- 8.39 Hui SL, Slemenda CW, Johnston CC. Baseline measurement of bone mass predicts fracture in white women. Ann Intern Med 1989,111:355-61.

and the second

المحمورية (1996). ومعاورة المحمور المح

- 8.40 Hui SL, Slemenda CW, Johnston CC. Age and bone mass as predictors of fracture in a prospective study. J Clin Invest 1988;81:1804-9.
- 8.41 Marshall D, Johnell O, Wedel H. Meta-analysis of how well measures of bone mineral density predict occurrence of osteoporotic fractures. Br J Med 1996;312:1254-59.
- 8.42 Bouxsein ML, Courtney AC, Hayes WC. Ultrasound and densitometry of the calcanues correlate with the failure loads of cadaveric femurs. Calcif Tissue Int 1995;56:99-103.
- 9.1 Miller PD, Bonnick SL, Rosen CJ. Consensus of an international panel on the clinical utility of bone mass measurements in the detection of low bone mass in the adult population. Calcif Tissue Int 1996;58:207-14.
- 9.2 Bergstrom R, Bergkvist L. Hormone replacement therapy and the risk for first hip fracture. A prospective, population-based cohort study. Ann Intern Med 1990;113:95-103.
- 9.3 Cauley JA, Seely DG, Ensrud K, Ettinger B, Black D, Cummings SR. Estrogen replacement therapy and fractures in older women. Ann Intern Med 1995;122:9-16.
- 9.4 Ettinger B, Genant HK, Cann CE. Long-term estrogen replacement therapy prevents bone loss and fractures. Ann Intern Med 1985;102:319-24.
- 9.5 Spector TD, Brennan P, Harris PA, Studd JWW, Silman AJ. Do current regimes of hormone replacement therapy protect against subsequent fractures? Osteoporosis Int 1992;2:219-24.
- 9.6 Grey AB, Cundy TF, Reid IR. Continuous combined oestrogen/progestogen therapy is well tolerated and increases bone mineral density at the hip and the spine in postmenopausal women. Clin Endocrinol 1994;40:671-7.

e trade a la contra de la contra

9.7 Hasling C, Charles P, Jensen FT, Mosekilde L. A comparison of the feffects of oestrogen/progestogen, high-dose oral calcium, intermittent cyclical etidronate and an ADFR regime on calcium kinetics and bone mass in postmenopausal women with spinal osteoporosis. Osteoporosis Int 1994;4:191-203.

A STATE AND A STATE OF A

and the second second second

1.15.1

- 9.8 Kohrt WM, Snead DB, Slatopolsky E, Birge SJ. Addiitive effect of weight-bearing exercise and estrogen on bone mineral density in older women. J Bone Miner Res 1995;10:1303-11.
- 9.9 Grey A, Cundy T, Evans M, Reid I. Medroxyprogesterone acetate enhances bone mineral density response to oestrogen in late post-menopausal women. Clin Endocrinol 1996;44:293-6.
- 9.10 Evans SF, Davie MWJ. Low and conventional dose transdermal oestradiol are equally effective at preventing bone loss in spine and femur at all postmenopausal ages. Clin Endocrinol 1996;44:79-84.
- 9.11 Kohrt WM, Birge SJ. Differential effects of estrogen treatment on bone mineral density of the spine, hip, wrist and total body in late postmenopausal women. Osteoporosis Int 1995;5:150-5.
- 9.12 Armamento-Villareal R, Civitelli R. Estrogen action on the bone mass of postmenopuasal women is dependent on body mass and initial bone density. J Clin Endocrinol Metab 1995;80:776-82.
- 9.13 Wimalawansa SJ. Combined therapy with estrogen and etidronate has an additive effect on bone mineral density in the hip and vertebrae: four-year randomised study. Am J Med 1995;99:35-42.

269

<u>ana na sana na sana k</u>ara ka malaka ka ka

he dan kem

- 9.14 Marslew U, Overgaard K, Riis BJ, Christiansen C. Two new combinations of estrogen and progestogen for prevention of postmenopausal bone loss: long term effects on bone, calcium, and lipid metabolism, climacteric symptoms and bleeding. Obstet and Gynecol 1992;79:202-10.
- 9.15 Storm T, Thamsberg G, Steiniche, Genant HK, Sorensen OH. Effects of intermittent cyclical etidronate therapy on bone mass and fracture rate in women with postmenopausal osteoporosis. N Eng J Med 1990;322:1265-71.
- 9.16 Watts NB, Harris ST, Genant HK et al. Intermittent cyclical etidronate treatment of postmenopausal osteoporosis. N Engl J Med 1990;323:73-9.
- 9.17 Storm T, Kollerup G, Thamsborg G, Genant HK, Sorensen OH. Five years of clinical experience with intermittent cyclical etidronate for postmenopausal osteoporosis. J Rheumatol 1996;23:1560-4.
- 9.18 Hillard TC, Whitcroft SJ, Marsh MS et al. Long-term effects of transdermal and oral hormone replacement therapy on postmenopausal bone loss. Osteoporosis Int 1994;4:341-8.
- 9.19 Marcelli C, Favier F, Kotzki PO, Ferrazzi V, Picot M-C, Simon L. The relationship between osteoarthritis of the hands, bone mineral density, and osteoporotic fractures in elderly women. Osteoporosis Int 1995;5:382-8.
- 9.20 Silberstein EB, Schnur W. Cyclical oral phosphate and etidronate increase femoral and lumbar bone mineral density and reduce lumbar spine fracture rate over three years. J Nuc Med 1992;33:1-5.
- 9.21 Harris ST, Watts NB, Jackson RD et al. Four-year study of intermittent cyclical etidronate treatment of postmenopausal osteoporosis: three years of blinded therapy followed by one year of open therapy. Am J Med 1993;95:557-67.

na falasi ne ule

ىيىن بارىمى مەكتە ئالىيى ئىلىچە تۈ<u>رۈن رالىيە بەترىكى بىلى بەترىكى بىلى يەترىكى بىلى مە</u>تتىك

9.22 Hassager C, Jensen SB, Christiansen C. Non-responders to hormone replacement therapy for the prevention of postmenopausal bone loss: do they exist? Osteoporosis Int 1994;4:36-41.

The Contract of the second second second

in the second second

- 9.23 Evans RA, Somers NM, Dunstan CR, Royle H, Kos S. The effect of low dose cyclical etidronate and calcium on bone mass in early postmenopausal women. Osteoporosis Int 1993;3:71-5.
- 9.24 Riggs BL, Melton LJ, O'Fallon WH. Drug therapy for vertebral fractures in osteoporosis: evidence that decreases in bone turnover and increases in bone mass both determine anti-fracture efficacy. Bone 1996;18(Suppl 3):197S-201S.
- 10.1 Avioli LV. Heparin induced osteopenia: an appraisal. Adv Exp Biol Med 1975;52:375-87.
- 10.2 Hall JG, Paul RM, Wilson KM. Maternal and fetal sequelae of anticoagulation during pregnancy. Am J Med 1980;79:122-40.
- 10.3 Shaul W, Emery H, Hall JC. Chondrodysplasia punctata and maternal warfarin use during pregnancy. Am J Dis Child 1975;129:360-2.
- 10.4 Fiore CE, Tamburino C, Foti R, Grimaldi D. Reduced bone mineral content in patients taking oral anticoagulants. South Med J 1990,83;538-42.
- 10.5 Resch H, Pietschmann P, Krexner E, Willvonseder R. Decreased bone mineral content in patients under anticoagulant therapy with phenprocoumon. Eur Heart J 1991;12:439-41.
- 10.6 Haarlem LJM, Knapen MJH, Hamulyak K, Vermeer C. Circulating osteocalcin during anticoagulant therapy. Thrombos Haemost 1988;60:79-82.
- 10.7 Price PA, Williamson MK. Effects of warfarin on bone. J Biol Chem 1981;256:12754-9

271

المواد سياد الأردق المحقق وتحج ومعيو ووراجي والمحكون والمواوي والمتعاوي والمراجع والمراجع والمراجع والمراجع

- 10.8 Gennari C, Civitelli R. Glucocorticoid-induced osteoporosis. Clin Rheum Dis 1986;12:637-54.
- 10.9 Price PA, Kaneda Y. Vitamin K counteracts the effects of warfarin in liver but not in bone. Thromb Res 1987;46:121-31.
- 10.10 Mazess RB, Bardon H, Ettinger M, Schultz E. Bone density of the radius, spine, and proximal femur in osteoporosis. J Bone Miner Res 1988;3:13-8.
- 10.11 Riggs BL, Melton LJ. Involutional osteoporosis. N Engl J Med 1986;314:1676-86.
- 10.12 Black DM, Cumming SR, Genant HK, Nevitt MC, Palermo L, Browner W. Axial and appendicular bone mineral density predict fractures in older women. J Bone Miner Res 1992;7:633-8.
- 10.13 Cumming SR, Black DM, Nevitt MC et al. Appendicular bone density and age predict hip fractures in women. JAMA 1990;263;655-8.
- 11.1 Deodhar AA, Woolf AD. Bone mass measurement and bone metabolism in rheumatoid arthritis: A review. Br J Rheumatol 1996;35:309-22.
- 11.2 Butler RC, Davie MWJ, Worsfold M, Sharp CA. Bone mineral content in patients with rheumatoid arthritis: relationship to low-dose steroid therapy. Br J Rheum 1991;30:86-90.
- 11.3 Verstraeten A, Dequeker J. Vertebral and peripheral bone mineral content and fracture incidence in postmenopausal patients with rheumatoid arthritis: effect of low dose corticosteroids. Ann Rheum Dis 1986;45:852-76.
- 11.4 Gough AKS, Lilley J, Eyre S, Holder RL, Emery P. Generalised bone loss in patients with early rheumatoid arthritis. Lancet 1994;344:23-7.

All the Automatical

والمتحجج والمراجع والمراجع والمحاج والمحاج والمحاج والمحاج والمحاج والمحاج والمحاج والمحاج والمحاج والمحاج

Martin Martin and

11.5 Reid DM, Kennedy NS, Smith MA, Tothill P, Nuki G. Total body calcium in rheumatoid arthritis: effects of disease activity and corticosteroid treatment. Br Med J 1982;285:330-2.

Charles and the second second second

- 11.6 Sambrook PN, Cohen ML, Eisman JA, Pocock NA, Champion GD, Yeates MG. Effect of low dose corticosteroid on bone mass in rheumatoid arthritis: a longitudinal study. Ann Rheum Dis 1989;48:535-8.
- 11.7 Sambrook PN, Eisman JA, Champion GD, Yeates MG, Pocock NA, Eberl S. Determinants of axial bone loss in rheumatoid arthritis. Arthritis Rheum 1987;30:721-8.
- 11.8 Hall GM, Spector TD, Griffin AJ, Jawad ASM, Hall ML, Doyle DV. The effect of rheumatoid arthritis and steroid therapy on bone density in postmenopausal women. Arthritis Rheum 1993;36:1510-6.
- 11.9 Lo Cascio V, Bonucci E, Imbimbo B et al. Bone loss in response to long-term glucocorticoid therapy. Bone Miner 1990;8:39-52.
- 11.10 Arnett FC, Edworthy SM, Bloch DA et al. The American Rheumatism Association 1987 revised criteria for the classification of rheumatoid arthritis. Arthritis Rheum 1988; 31:315-24.
- 11.11 Sambrook PN, Ansell BM, Foster S, Gumpel JM, Hesp R, Reeve J. Bone turnover in early rheumatoid arthritis. 2. Longitudinal bone density studies. Ann Rheum Dis 1985;44:580-4.
- 11.12 Sambrook PN, Shawe D, Hesp R et al. Rapid periarticular bone loss in rheumatoid arthritis. Arthritis Rheum 1990;33:615-22.
- 11.13 Toyoda T, Inokuchi S, Saito S, Horis Y, Tomita S. Bone loss of the radius in rheumatoid arthritis. Comparison between 34 patients and 40 controls. Acta Orthop Scand 1996;67:269-73.

273

والمستحية المستحد والمتحد والمتحد وتوكي وتحاكم والمحاج والمحاج والمحاج والمحاج والمحاج والمحاج والمحاج والمحاج

- 11.14 Joffe I, Epstein S. Osteoporosis associated with rheumatoid arthritis: pathogenesis and management. Seminars in Arthritis Rheum 1991;20:256-72.
- 11.15 Lane NE, Pressman AR, Star VL, Cummings SR, Nevitt MC. Rheumatoid arthritis and bone mineral density in elderly women. J Bone Miner Res 1995;10:257-63.
- 11.16 Sambrook PN, Eisman JA, Yeates, Pocock NA, Eberl S, Champion GD. Osteoporosis in rheumatoid arthritis: safety of low dose corticosteroids. Ann Rheum Dis 1986;45:950-3
- 11.17 Cooper C, Coupland C, Mitchell M. Rheumatoid arthritis, corticosteroid therapy and the risk of hip fracture. Ann Rheum Dis 1995;54:49-52.
- 11.18 Laan RFJM, van Riel PLCM, Erning LJTO, Lemmens JAM, Riujs SHJ, van de Putte LBA. Vertebral osteoporosis in rheumatoid arthritis patients: effect of low dose prednisolone therapy. Br J Rheumatol 1992;31:91-6.
- 11.19 Martin JC, Reid DM. Appendicular measurements in screening women for low axial bone mineral density. Br J Radiol 1996;69:234-40.
- 11.20 Peel NFA, Moore DJ, Barrington NA, Bax DE, Eastell R. Risk of vertebral fracture and relationship to bone mineral density in steroid treated rheumatoid arthritis. Ann Rheum Dis 1995;54:801-6.
- 11.21 Reid DM, England AJ. Peripheral bone mass measurements is there any clinical value in rheumatoid arthritis? Br J Rheumatol 1996;35:109-10.
- 12.1 Parfitt AM. Problems in the application of in vitro systems to the study of human bone remodelling. Calcif Tissue Int 1995;56(Suppl 1):S5-S7.
- 12.2 Feldkamp LA, GoldsteinSA, Parfitt AM, Jesion G, Kleerekoper M. The direct examination of three-dimentional bone architecture in vitro by computed tomography. J Bone Miner Res 1989;4:3-11.

<u>, na sente en la companya de la comp</u>

- 12.3 Kuhn JL, Goldstein SA, Feldkamp LA, Goulet RW, Jesion G Evaluation of a microcomputed tomography system to study trabecular bone structure. J Orthop Res 1990;8:833-42.
- 12.4 Durand EP, Ruegsegger P. Cancellous bone structure: Analysis of high resolution CT images with run-length methods. J Comput Assist Tomogr 1991;15:133-39.
- 12.5 Kaissl W, Reugsegger P. 3D-QCT at peripheral measuring sites. Osteoporosis Int 1991:1:193.
- 12.6 Chevalier F, Laval-Jeantet AM, Laval-Jeantet M, Bergot C. CT images of the vertebral trabecular network in vitro. Calcif Tissue Int 1992;51:8-13.
- 12.7 Gordon CL, Webber CE, Adachi JD, Christoforous N. In vivo assessment of trabecular bone structure at the distal radius from high resolution computed tomography images. Phys Med Biol 1996;41:495-508.
- 12.8 Muller R, Hahn M, Vogel M, Delling G, Ruegsegger P. Morphometric analysis of non-invasively assessed bone biopsies: comparison of high resolution computed tomography and histologic sections. Bone 1996;18:215-20.

and the second second second to the second second

APPENDIX 1	Glossary of Abbreviations
ACTH	Adrenocorticotrophic Hormone
AF	Atrial Fibrillation
ANOVA	Analysis of Variance
AP	Antero-posterior
AUC	Area Under the Curve
BMD	Bone Mineral Density
BMU	Basic Multicellular Unit
BUA	Broadband Ultrasound Attenuation (calcaneal site)
X ²	Chi square
CAMP	Cyclic Adenosine-3',5'-monophosphate
СМ	centi-meter
СТ	Computed Tomography
CV	Co-efficient of Variation
dB/MHz	Decibel / Mega Hertz
DHEA	Dehydroepiandrosterone
DPA	Dual Photon Absorptiometry
DXA	Dual Energy X-Ray Absorptiometry
ETD	Etidronate
FN	DXA Femoral Neck BMD
FSH	Follicle Stimulating Hormone
FТ	DXA Femoral Trochanter BMD
FW	DXA Femoral Ward's Area BMD
g	grammes
HRT	Hormone Replacement Therapy
HVR	Heart Valve Replacement
kg	Kilogram
LARS	Larsen Score
L2-L4	2nd to 4th lumbar vertebrea
ĽН	Luteinizing Hormone
LHRH	Luteinizing Hormone Releasing Hormone
LS	DXA Lumbar Spine BMD

1.114

.

ter y thin et

the state of the state

1.13

"你们,你们们还有一个都是你们的你,我们们就是你的你,你都是你的你,你们还是你们还是你们的你,你们们都是你们,你不是你们,你们们还是我们都是你们的你们,我不能能我

m	Meter
m/s	Meter / Second
MCP	Metacarpo-Phalangeal Joint
MXA	Morphometric X-ray Absorptiometry
NS	Statistically Non-Significant
1-CTP	Cross-linked Carboxyterminal Telopeptide of
	Type 1 Collagen
1-NTP	Cross-linked Aminoterminal Telopeptide of Type
	1 Collagen
PBM	Peak Bone Mass
PICP	Carboxyterminal Propeptide of Type 1 Collagen
PINP	Aminoterminal Propeptide of Type 1 Collagen
PIP	Proximal Inter-Phalangeal Joint
POST	Postmenopausal Women
PRE	Premenopausal Women
pQCT	Peripheral Quantitative Computed Tomography
	(ultra-distal radial site)
PVD	Peripheral Vascular Disease
Qcort	pQCT Cortical BMD
QCT	Quantitative Computed Tomography
Qscort	pQCT Subcortical BMD
Qtot	pQCT Total BMD
Qtrab	pQCT Trabecular BMD
RA	Rheumatoid Arthritis
RA+P	Steroid Treated RA group
RhF	Rheumatoid Factor
ROC	Receiver Operator Curve
ROI	Region of Interest
SD	Standard Deviation
SPA	Single Photon Absorptiometry
T4-L5	4th thoracic to 5th lumbar vertebrae
T-Score	The number of standard deviations above or
	below the mean value of young normals

an bara a substantia da antina da substantia da substanta da substanta da substantia da substantia da substant Presente da substantia da su

the second se

VN	Voxel Number
VN-diff	Difference in voxel numbers
vos	Ultrasonic Velocity (calcaneal site)
VTE	Recurrent Venous Thrombo-embolism
WB	DXA Whole Body
WB-ARM	DXA Arm BMD (derived from the whole body DXA
	scan image)
WB-BMD	DXA Whole Body BMD
YPM	Years Postmenopause
γr	Year
Z-Score	The number of standard deviations above or
	below the sex and age related normal mean

APPENDIX 2 Papers Published from the Thesis Work

and the second second

 Philip WJU, <u>Martin JC</u>, Richardson JM, Webster J, Reid DM, Douglas AS. Decreased axial and peripheral bone mineral density in patients taking long term warfarin.
 Quarterly Journal of Medicine 1995;88:635-40

the second metric of the At-

a production of the second state of the

1947 B.

- 2. <u>Martin JC</u>, Reid DM. Appendicular measurements in screening women for low axial bone mineral density. British Journal of Radiology 1996;69:234-40
- 3. <u>Martin JC</u>, Munro R, Campbell MK, Reid DM. Effects of disease and corticosteroids on appendicular bone mass in postmenopausal women with rheumatoid arthritis: comparison with axial measurements. British Journal of Rheumatology 1997;36:43-50

APPENDIX 3 Abstracts Published from the Thesis Work

1. <u>Martin JC</u>, Munro R, Reid DM.

and the second second second

Comparison of peripheral and axial bone mineral density (BMD) with bone ultrasound in postmenopausal women with rheumatoid arthritis on longterm corticosteroids. British Journal of Rheumatology 1992;32(Abst Suppl):40

2. Martin JC, Reid DM.

Comparison of dual energy x-ray absorptiometry (DXA), peripheral quantitative computed tomography (pQCT) and ultrasound of the os calcis (CUBA) in screening perimenopausal women for osteopenia. Bone & Tooth Society, London, Dec 1993.

Bone 1994

- 3. <u>Martin JC</u>, Philip W, Richardson J, Webster J, Reid DM, Douglas AS. Effects of longterm warfarin therapy on bone mineral density (BMD) at peripheral and axial sites. Scottish Medical Journal 1994;39:153
- 4. <u>Martin JC</u>, Munro R, Reid DM.
 Bone mass at multiple bone sites in Rheumatoid Arthritis (RA): Effects of disease and corticosteroid therapy.
 British Journal of Rheumatology 1994;33 (Abst Suppl 1):83
- Martin JC, Reid DM. Comparison of axial and peripheral bone loss in perimenopausal women.
 Bone and Mineral 1994;25(Abst Suppl 2):S27

280

6. Martin JC, Reid DM.

weeks to appreciate the test of the second state to the strategy spin test and the

The use of scan voxel numbers to improve the precision of peripheral quantitative computed tomography (pQCT). Current Research in Osteoporosis and bone Mineral Measurement III 1994: p53

· · · · ·

したい きんしたんかんか 見た

1997 (B. 1997 - 1997 - 1997)

7. <u>Martin JC</u>, A England, DM Reid.

Can changes in appendicular bone predict axial bone mineral density (BMD) during treatment with hormone replacement therapy (HRT) and cyclical etidronate?. British Journal of Rheumatology 1995;34(Abst Suppl 2):\$72

- 8. Reid DM, Stewart A, <u>Martin JC</u>. Does quantitative ultrasound have utility in predicting fractures and monitoring therapy in osteoporosis: comparison with dual enery X-ray absorptiometry (DXA). Current Research in Osteoporosis and bone Mineral Measurement IV 1996: p57
- 9. <u>Martin JC</u>, A Stewart, DM Reid

Discrimination of hip fractures by peripheral quantitative computed tomography (pQCT): comparison with dual energy xray absorptiometry (DXA) of the hip and calcaneal ultrasound measurements.

British Journal of Rheumatology 1996;35(Abst Suppl 2):p6.

APPENDIX 4 Presentations from the Thesis to Learned Societies

- Effects of longterm warfarin therapy on bone mineral density (BMD) at peripheral and axial sites.
 Scottish Society for Experimental Medicine, Dundee, December 1993.
- Bone mass at multiple bone sites in Rheumatoid Arthritis (RA): Effects of disease and corticosteroid therapy.
 British Society for Rheumatology General Meeting, Brighton April 1994.
- 3. Comparison of axial and peripheral bone loss in perimenopausal women. The 10th International Bone Densitometry Workshop, Venice, April 1994.
- 4. The use of scan voxel numbers to improve the precision of peripheral quantitative computed tomography (pQCT). The 4th Bath Conference on Osteoporosis and Bone Mineral Measurement, June 1994.

A CONTRACTOR OF A STATE

n an an the second s